# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# FORM 8-K

CURRENT REPORT Pursuant to Section 13 OR 15(d) of The Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): June 24, 2022

# EARGO, INC.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation)

> 2665 North First Street, Suite 300 San Jose, California

(Address of principal executive offices)

001-39616 (Commission File Number) 27-3879804 (IRS Employer Identification No.)

95134 (Zip Code)

Registrant's telephone number, including area code (650) 351-7700

Not Applicable

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered:
Common Stock, \$0.0001 par value per share	EAR	The Nasdag Stock Market LLC
Common Stock, \$0.0001 par value per share		The Masuay Stock Market EDC

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company  $\Box$ 

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.  $\Box$ 

# Item 1.01. Entry into a Material Definitive Agreement.

# Note Purchase Agreement

On June 24, 2022, Eargo, Inc. (the "Company") and its subsidiaries entered into a Note Purchase Agreement (the "Note Purchase Agreement") with noteholders affiliated with Patient Square Capital (the "Noteholders") and Drivetrain Agency Services, LLC, as administrative agent and collateral agent (the "Administrative Agent"). Pursuant to the Note Purchase Agreement, the Company has agreed to issue and sell up to \$125 million in senior secured convertible notes (the "Notes") of the Company (the "Financing"). The Notes will be convertible into shares ("Conversion Shares") of common stock, \$0.0001 par value per share, of the Company ("Common Stock"), as further described below.

Pursuant to the Note Purchase Agreement, the Noteholders have agreed to purchase \$100 million in Notes (the "First Tranche") on or before July 11, 2022 (the "First Tranche Closing"). The Noteholders have agreed to purchase up to an additional \$25 million of Notes (the "Second Tranche") if the Company has not completed the Rights Offering (as defined below) within 150 days after the First Tranche Closing or, alternatively, if the Company completes such Rights Offering within such 150-day period but the Company's existing stockholders subscribe to purchase less than 75 million shares of newly issued Common Stock in such Rights Offering.

The closings of the First Tranche and the Second Tranche are conditioned upon certain closing conditions. The Note Purchase Agreement also contains certain representations, warranties and covenants of the Company, including with regard to the incurrence of indebtedness and liens, making of investments, dispositions of property and transactions with affiliates.

#### Use of Proceeds

The Company is permitted to use the proceeds from the sale of the Notes as working capital, to fund its general business requirements and to pay off existing indebtedness.

#### Interest, Redemption, Maturity and Events of Default

The Notes will be senior, secured obligations of the Company, bearing interest at a rate of 12.00% per annum, payable quarterly in arrears on the first calendar day of each calendar quarter commencing on July 1, 2022. Immediately upon the occurrence and during the continuance of an event of default, the interest rate will be increased by an additional 12.00%. Other than on the Maturity Date (as defined below) or any optional redemption date, accrued interest shall be paid in-kind.

The Company may, at its option, repay all (but not a portion) of the Notes outstanding upon three business days' prior written notice at a price equal to the Repayment Value of the Notes outstanding. The Repayment Value of any Note on any applicable date means an amount payable such that the annualized return on the initial principal amount of the Notes is no less than 12.00%, or, if greater, an amount equal to 150% of the initial principal amount of such Note.

The Notes will mature and be due in cash at the Repayment Value on the one-year anniversary of the First Tranche Closing Date, subject to earlier conversion, redemption or repurchase in accordance with their terms (the "Maturity Date").

The Note Purchase Agreement contains certain "events of default" as further described in the Note Purchase Agreement, which, if triggered, would result in an increase in the amount of interest due on the Notes and may result in the acceleration of the maturity of the Notes (including the Repayment Value), among other things.

# Rights Offering

In connection with the Financing, the Company has agreed to consummate a rights offering (the "Rights Offering") for an aggregate of 375 million shares of Common Stock to the Company's stockholders as of a record date to be determined at an offering price of \$0.50 per share of Common Stock. The Rights Offering is expected to be conducted as soon as practicable following the First Tranche Closing and in any event within 180 days thereafter.

#### Conversion

If a Rights Offering is consummated within 150 days following the First Tranche Closing, then the Notes will automatically convert into (i) a number of shares of Common Stock equal to 375 million less the number of shares actually subscribed for and purchased with the Rights Offering, and (ii) cash in an amount equal to (x) the Repayment Value of Notes outstanding less (y) the Rights Offering Shortfall Amount (as defined in the Note Purchase Agreement).

If a Rights Offering is consummated after the 150th day following the First Tranche Closing but on or prior to the 180th day following the First Tranche Closing, then, subject to certain conditions, the Second Tranche Notes will be issued and (a) the total gross proceeds of the Rights Offering will be immediately used to repay in cash a portion of the Notes at the Repayment Value, with the principal amount of the Notes redeemed being equal to the total gross proceeds of the Rights Offering divided by 1.50, and (b) the remaining amount of the Notes that are not redeemed will immediately convert into a number of shares equal to 375 million less the number of shares actually subscribed for and purchased in the Rights Offering. The Second Tranche Notes may not be issued if the Rights Offering is not consummated by the 180th day following the First Tranche Closing.

# Restrictions on Certain Activities

The Note Purchase Agreement contains various covenants that limit the Company's ability to engage in specified types of transactions without the Noteholders' prior consent. These covenants limit the Company's ability to, among other things:

- encumber or license the Company's intellectual property subject to certain exceptions;
- sell, transfer, lease or dispose of the Company's assets subject to certain exclusions;
- create, incur or assume additional indebtedness;
- encumber or permit liens on any of the Company's assets other than certain permitted liens;
- make restricted payments, including paying dividends on, repurchasing or making distributions with respect to any of the Company's capital stock;
- make specified investments (including loans and advances);
- consolidate, merge with, or acquire any other entity, or sell or otherwise dispose of all or substantially all of the Company's assets; and
- enter into certain transactions with the Company's affiliates.

Nothing in the Note Purchase Agreement prohibits the Company from refinancing the Notes (at their Repayment Value) with a new equity or debt financing, or from selling the Company (and repaying the Notes at the Repayment Value in connection with such sale). The Company expects that in the event it enters into a sale transaction, then it would not proceed with the Rights Offering.

#### Collateral

Pursuant to the Note Purchase Agreement, the Company and each of its subsidiaries has granted a first-priority security interest on substantially all of its assets, including intellectual property. Any subsidiary of the Company formed or acquired is also required to become party to the Note Purchase Agreement, subject to the same requirements. Upon the occurrence of an event of default, the Administrative Agent, on behalf of the Noteholders, is entitled to, among other things, foreclose on the collateral.

# Stockholder Approval

In connection with the Financing, the Company has agreed to hold a meeting of its stockholders within 120 days of the First Tranche Closing (the "Stockholders Meeting") to obtain stockholder approval to (i) increase the number of authorized shares of Common Stock under the Company's certificate of incorporation to an amount necessary to consummate the Rights Offering and (ii) issue the full potential amount of the Conversion Shares pursuant to the rules of The Nasdaq Stock Market LLC.

The foregoing description of the Notes and the Note Purchase Agreement is qualified in its entirety by the full text of the Note Purchase Agreement, which is attached hereto as Exhibit 10.1 and incorporated herein by reference.

#### Investors' Rights Agreement

On June 24, 2022, the Company and the Noteholders entered into an Investors' Rights Agreement (the "Investors' Rights Agreement") pursuant to which, among other things, the Company granted the Noteholders certain governance, information and other rights.

#### Restrictions on Transfer

Prior to the Maturity Date (unless an event of default has occurred and is continuing), the Investor Parties (as defined below) may not transfer any of the Notes to any person without the consent of the Company, except each Noteholder may transfer the Notes (i) to its direct or indirect limited partners or other investors, in each case, to which Notes are transferred by way of distribution, or the Noteholders' affiliated funds or controlled affiliates (such controlled affiliates, limited partners and other investors that hold the Notes or Conversion Shares, collectively, the "Permitted Transferees"), (ii) in connection with customary pledges in connection with financing arrangements by the Noteholders or to the extent that all of the net proceeds of such sale are solely used to satisfy a margin call (i.e. posted as collateral) or re-pay bona fide loan or lending transaction entered into by the Noteholders (a "Financing Arrangement") to the extent necessary to satisfy a bona fide margin call on such Financing Arrangement or to avoid a bona fide margin call on such Financing Arrangement, the lender has executed an issuer agreement, or (iii) following the date the Company commences a voluntary case under Title 11 of the United States Bankruptcy Code or any other similar insolvency laws. The Noteholders and any transferee, whether during or after such three-year period, pursuant to clause (i) is an "Investor Party" and any transferee pursuant to clauses (ii) and (iii) above is an "Other Holder".

In connection with any transfer of the Notes that is effected pursuant to a transaction in which the Investor Parties (or any of their representatives) negotiate the terms of such transfer directly with the third party purchaser (other than any underwriter, dealer (including a dealer acting as a block positioner), market maker, placement agent or initial purchaser thereof), at no time may the Investor Parties and Other Holders, as applicable, knowingly (after reasonable inquiry) transfer any Notes to any Company Competitor, or any Activist Investor (each as defined in the Investors' Rights Agreement) or person who the Investor Party or Other Holder, as applicable, reasonably should know has been an Activist Investor in the last three years. In no event shall the foregoing limitations apply to, or limit in any way, sales by the Investor Parties in transactions effected on any stock exchange.

# Noteholder Observer and Director Nominees

Prior to consummation of a Rights Offering, for as long as the Investor Parties hold any Notes, the Investor Parties shall have the right to appoint one observer (the "Observer") to the Company's board of directors (the "Board").

From and after the Rights Offering, the Investor Parties shall have the right to nominate the number of directors (the "Investor Directors") to the Board that is proportionate to Investor Parties' ownership of the Company, if any, following the Rights Offering, which will be rounded up to the nearest whole number (and shall in no event be less than one).

The Company expects to enter into an indemnification agreement with each Investor Director (the "Indemnification Agreement") and entered into a Board observer agreement with the Investor Party on June 24, 2022 (the "Board Observer Agreement").

The Indemnification Agreement generally requires us, among other things, to indemnify each Investor Director against certain liabilities that may arise by reason of their status or service as a director. The Indemnification Agreement also generally requires us to advance any expenses incurred by each Investor Director as a result of any proceeding against them as to which they could be indemnified. These indemnification provisions and the Indemnification Agreements may be sufficiently broad to permit indemnification of each Investor Director for liabilities, including reimbursement of expenses incurred, arising under the Securities Act of 1933, as amended (the "Securities Act").

The Board Observer Agreement generally provides for, among other things, the Observer's rights to be invited to attend, in a non-voting observer capacity, all meetings of the Board and Board committees. The Board Observer Agreement also generally requires the Company to indemnify the Observer to the same extent provided in the Indemnification Agreement. The Observer's rights under the Board Observer Agreement shall terminate at the time the Investor no longer has the right to appoint an Observer pursuant to the Investors' Rights Agreement and the other provisions therein.

The foregoing descriptions of the Indemnification Agreement and the Board Observer Agreement are qualified in their entirety by the full text of such agreements. The form of the Indemnification Agreement is attached as Exhibit 10.2 hereto and the Board Observer Agreement is attached as Exhibit 10.3 hereto, and are incorporated herein by reference.

# Standstill

So long as either (i) the Investor Parties or Other Holders have record and beneficial ownership representing more than five percent of the outstanding Common Stock of the Company or (ii) the Investor Parties have the right to nominate a director to the Board (or any person so nominated continues to serve on the Board), the Investor Parties and Other Holders must comply with customary standstill provisions. The standstill provisions will terminate if, following any conversion in connection with the Rights Offering described below, the Investor acquires control of a majority of the voting power of the Company.

# Voting Agreement

For so long as the Investor Parties have the right to designate an Observer or designate or nominate a director to the Board (or any person so designated or nominated continues to serve on the Board), the Investor Parties or Other Holders, as applicable, have agreed at each meeting of the stockholders to vote all shares of Common Stock owned (a) in favor of each director nominated or recommended by the Board and against the removal of any director who has been elected following nomination or recommendation by the Board, (b) against any stockholder nomination for a director that is not approved and recommended by the Board, (c) in favor of the Company's "say-on-pay" proposal and any proposal by the Company relating to equity compensation that has been approved by the Board or the Board's Compensation Committee, and (d) in favor of the Company's proposal for ratification of the appointment of the Company's independent registered public accounting firm. The foregoing voting agreement will terminate if, following any conversion in connection with the Rights Offering described below, the Investor acquires control of a majority of the voting power of the Company.

# Financial Statements and Other Information Rights

For so long as the Notes are outstanding or the Investor holds at least 5% of the outstanding common stock of the Company, the Company has agreed to provide the Investor Parties with certain information and financial statements, including monthly management reports and annual operating budgets and financial projections, as further described in the Investors' Rights Agreement.

The foregoing description of the Investors' Rights Agreement is qualified in its entirety by the full text of the Investors' Rights Agreement, which is attached as Exhibit 10.4 hereto and incorporated herein by reference.

# Registration Rights Agreement

At the First Tranche Closing, the Company and the Noteholders will enter into a Registration Rights Agreement (the "Registration Rights Agreement") pursuant to which, among other things, the Company will grant the Noteholders certain customary demand and piggyback registration rights. Under the Registration Rights Agreement, the Company will be required to use its reasonable best efforts to file and maintain the effectiveness of a registration statement providing for the resale of the Conversion Shares.

The foregoing description of the Registration Rights Agreement is qualified in its entirety by the full text of the Registration Rights Agreement, which is attached as Exhibit 10.5 hereto and incorporated herein by reference.

# Item 2.03. Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

The information in Item 1.01 of this Current Report on Form 8-K (this "Report") is incorporated by reference herein.

# Item 3.02. Unregistered Sales of Equity Securities.

The information in Item 1.01 of this Report is incorporated by reference herein.

On June 24, 2022, the Company and its subsidiaries entered into the Note Purchase Agreement with the Noteholders and the Administrative Agent. Pursuant to the Note Purchase Agreement, the Company has agreed to issue and sell up to \$125 million in Notes. The Notes will be convertible into shares of Common Stock, as further described above.

Pursuant to the Note Purchase Agreement, the Noteholders have agreed to purchase \$100 million in Notes on or before July 11, 2022. The Noteholders have agreed to purchase up to an additional \$25 million of Notes if the Company has not completed the Rights Offering within 150 days after the First Tranche Closing or, alternatively, if the Company completes such Rights Offering within such 150-day period but the Company's existing stockholders subscribe to purchase less than 75 million shares of newly issued Common Stock in such Rights Offering.

The offer and sale of the shares of Notes pursuant to the Note Purchase Agreement are being made in reliance on an exemption from registration under the Securities Act pursuant to Section 4(a)(2) thereof. The Conversion Shares will be issued, if at all, in reliance upon the exemption from registration in Section 3(a)(9) of the Securities Act.

# Item 7.01. Regulation FD Disclosure.

A copy of the press release announcing the Financing and related matters is attached hereto as Exhibit 99.1 and is incorporated herein by reference.

The information contained in this Item 7.01, including Exhibit 99.1, is furnished herewith and shall not be deemed "filed" for purposes of Section 18 of the Securities and Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to the liabilities of that section, and such information shall not be deemed to be incorporated by reference into any of the Company's filings under the Securities Act or the Exchange Act, except as shall be expressly set forth by specific reference in such filing.

## Item 9.01. Financial Statements and Exhibits.

(d) Exhibits.

Exhibit No.	Description
10.1†*	Note Purchase Agreement, dated June 24, 2022, by and among Eargo, Inc., Eargo Hearing, Inc., Eargo Screening, LLC, noteholders affiliated with Patient Square Capital and Drivetrain Agency Services, LLC, as administrative agent and collateral agent.
10.2*	Form of Indemnification Agreement.
10.3*	Board Observer Agreement, dated June 24, 2022, by and between Eargo, Inc. and PSC Echo LP.
10.4†*	Investors' Rights Agreement, dated June 24, 2022, by and between Eargo, Inc. and those certain investors set forth therein.
10.5*	Registration Rights Agreement, dated June 24, 2022, by and between Eargo, Inc. and those certain investors set forth therein.
99.1	Press Release, dated June 27, 2022.
104	Cover Page Interactive Data File (embedded within the inline XBRL document).

<sup>†</sup> Certain of the exhibits and schedules to this Exhibit have been omitted in accordance with Regulation S-K Item 601(a)(5). The Registrant agrees to furnish a copy of all omitted exhibits and schedules to the SEC upon its request.

\* Certain confidential information contained in this document, marked by [\*\*\*], has been omitted pursuant to Item 601(b)(10)(iv) of Regulation S-K because it is both (i) not material and (ii) the type of information that the registrant treats as private or confidential.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: June 27, 2022

# EARGO, INC.

By: /s/ Adam Laponis

Adam Laponis Chief Financial Officer

# Certain confidential information contained in this document, marked by [\*\*\*], has been omitted pursuant to Item 601(b)(10)(iv) of Regulation S-K because it is both (i) not material and (ii) the type of information that the registrant treats as private or confidential.

# NOTE PURCHASE AGREEMENT

This NOTE PURCHASE AGREEMENT, dated as of June 24, 2022 (as amended, restated, amended and restated, supplemented or otherwise modified from time to time in accordance with <u>Section 12.6</u>, this "Agreement"), is by and among EARGO, INC., a Delaware corporation ("Eargo"), EARGO HEARING, INC., a California corporation ("Eargo Hearing"), EARGO SCREENING, LLC, a Delaware limited liability company ("Eargo Screening", together with Eargo and Eargo Hearing, each individually an "Issuer" and, collectively, the "Issuers"), the Noteholders (as defined below) from time to time party hereto, and DRIVETRAIN AGENCY SERVICES, LLC ("DriveTrain"), as administrative agent and collateral agent for the Secured Parties (in such capacities, the "Administrative Agent").

WHEREAS, the Issuer desires to issue and sell to the Noteholders, and the Noteholders have agreed to purchase, and the Administrative Agent has agreed to administer, Notes in an aggregate principal amount of up to \$125,000,000, in each case subject to the terms and conditions set forth herein; and

WHEREAS, the Issuer has agreed to secure all of its Obligations by granting to Administrative Agent, for the benefit of Secured Parties, a first priority, perfected Lien on all of its assets, subject to Permitted Liens.

NOW, THEREFORE, in consideration of the premises and the agreements, provisions and covenants herein contained, the parties hereto agree as follows:

# 1. ACCOUNTING AND OTHER TERMS

**1.1** Accounting terms not defined in this Agreement shall be construed following GAAP. Calculations and determinations must be made following GAAP. Capitalized terms not otherwise defined in this Agreement shall have the meanings set forth in <u>Section 15</u>. All other terms contained in this Agreement, unless otherwise indicated, shall have the meaning provided by the Code to the extent such terms are defined therein.

**1.2 Other Interpretive Provisions.** With reference to this Agreement and each other Note Document, unless otherwise specified herein or in such other Note Document:

(a) The meanings of defined terms are equally applicable to the singular and plural forms of the defined terms.

(b) The words "herein," "hereof" and "hereunder" and words of similar import when used in any Note Document shall refer to such Note Document as a whole and not to any particular provision thereof.

(c) The word "including" and words of similar import shall mean "including, without limitation," unless otherwise specified.

(d) References in this Agreement to an Exhibit, Schedule, Article, Section, clause or subclause refer (A) to the appropriate Exhibit or Schedule to, or Article, Section, clause or subclause in this Agreement or (B) to the extent such references are not present in this Agreement, to the Note Document in which such reference appears.

(e) The term "documents" includes any and all instruments, documents, agreements, certificates, notices, reports, financial statements and other writings, however evidenced, whether in physical or electronic form.

(f) In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including"; the words "to" and "until" each mean "to but excluding"; and the word "through" means "to and including."

(g) Section headings herein and in the other Note Documents are included for convenience of reference only and shall not affect the interpretation of this Agreement or any other Note Document.

(h) The words "asset" and "property" shall be construed as having the same meaning and effect and to refer to any and all tangible and intangible assets and properties, including cash, securities, accounts and contract rights.

(i) All references to any Person shall be constructed to include such Person's successors and assigns (subject to any restriction on assignment set forth herein) and, in the case of any Governmental Authority, any other Governmental Authority that shall have succeeded to any or all of the functions thereof.

(j) For purposes of the Note Documents, whenever a representation or warranty is made to Issuer's knowledge or awareness, to the "best of" Issuer's knowledge, or with a similar qualification, knowledge or awareness means the actual knowledge, after reasonable investigation, of any Responsible Officer.

(k) References in this Agreement to an "Issuer" shall mean all Issuers.

**1.3 References to Agreements, Laws, Etc.** Unless otherwise expressly provided herein, (a) references to Operating Documents, agreements (including the Note Documents) and other contractual instruments shall be deemed to include all subsequent amendments, restatements, extensions, supplements and other modifications thereto, but only to the extent that such amendments, restatements, extensions, supplements and other modifications are not prohibited by any Note Document; and (b) references to any Law shall include all statutory and regulatory provisions consolidating, amending, replacing, supplementing or interpreting such Law.

1.4 Time of Day. Unless otherwise specified, all references herein to times of day shall be references to New York, New York time (daylight or standard, as applicable).

**1.5 Timing of Payment or Performance.** When the payment of any obligation or the performance of any covenant, duty or obligation is stated to be due or performance required on a day which is not a Business Day, the date of such payment or performance shall extend to the immediately succeeding Business Day.

# 2. <u>NOTES.</u>

## 2.1.1 Notes Issuance.

# (a) Purchase and Sale of the First Tranche Notes.

(i) Subject to the satisfaction or waiver of the conditions set forth in <u>Section 3.1</u>, each Noteholder with a First Tranche Commitment listed on <u>Schedule I</u> hereto severally agrees to purchase, and Eargo, in its capacity as an Issuer, agrees to issue, sell and deliver, on the First Tranche Closing Date, First Tranche Notes in an amount equal to such Noteholder's First Tranche Commitment. Any First Tranche Note issued under this <u>Section 2.1.1(a)</u> and subsequently repaid or prepaid may not be

reissued. Each Noteholder's First Tranche Commitment shall terminate immediately and without further action on the First Tranche Closing Date after giving effect to the purchase of the First Tranche Notes in an amount equal to such Noteholder's First Tranche Commitment on the First Tranche Closing Date. The First Tranche Notes issued pursuant hereto shall evidence the principal amounts of all First Tranche Notes sold hereunder and the date and principal amount of each purchase and sale of First Tranche Notes to the Noteholder's by Eargo. Each payment or prepayment made on account of the principal thereof, and in each case the resulting aggregate unpaid principal balance thereof (including capitalized interest), shall be recorded by Administrative Agent in accordance with Section 2.1.3.

(ii) <u>The First Tranche Notes Closing</u>. The purchase and sale of the First Tranche Notes will occur at a closing (the "**First Tranche Closing**", and the date of the First Tranche Closing, the "**First Tranche Closing Date**") to be held at the offices of Ropes & Gray LLP, 1211 Avenue of the Americas, New York, New York 10036, or at such other location as may be agreed upon by the parties hereto, subject to the terms and conditions hereof; *provided*, that the First Tranche Closing Date shall occur on a date after the Closing Date specified by the Noteholders holding First Tranche Commitments, which, in any event, shall not be more than ten (10) Business Days after the Closing Date, subject to the terms and conditions hereof.

(iii) <u>Payment of Purchase Price</u>. Upon satisfaction or waiver of the conditions precedent specified in <u>Section 3.1</u>, at the First Tranche Closing, against payment to Eargo by wire transfer of immediately available funds, in accordance with the wire instructions for Eargo set forth on <u>Schedule I</u> or as otherwise directed by Eargo, Eargo will deliver the First Tranche Notes to each Noteholder, issued in the name of each such Noteholder and in the principal amount of each such Noteholder's First Tranche Commitment.

#### (b) Purchase and Sale of the Second Tranche Notes.

(i) Subject to the satisfaction or waiver of the conditions set forth in <u>Section 3.2</u>, each Noteholder with a Second Tranche Commitment listed on <u>Schedule I</u> hereto severally agrees to purchase, and Eargo, in its capacity as an Issuer, agrees to issue, sell and deliver, on the Second Tranche Closing Date, Second Tranche Notes in an amount equal to such Noteholder's Second Tranche Commitment. Any Second Tranche Note issued under this <u>Section 2.1.1(b)</u> and subsequently repaid or prepaid may not be reissued. Each Noteholder's Second Tranche Commitment shall terminate immediately and without further action upon the earlier to occur of (i) the Second Tranche Closing Date after giving effect to the purchase of the Second Tranche Notes in an amount equal to such Noteholder's Second Tranche Closing Date and (ii) the date that occurs one hundred eighty (180) days after the First Tranche Closing Date (such date pursuant to this <u>clause (ii)</u>, "**Second Tranche Commitment Termination Date**"). The Second Tranche Notes issued pursuant hereto shall evidence the principal amounts of all Second Tranche Notes sold hereunder and the date and principal amount of each purchase and sale of Second Tranche Notes to the Noteholders by Eargo. Each payment or prepayment made on account of the principal thereof, and in each case the resulting aggregate unpaid principal balance thereof (including capitalized interest), shall be recorded by Administrative Agent in accordance with <u>Section 2.1.3</u>.

(ii) <u>The Second Tranche Notes Closing</u>. The purchase and sale of the Second Tranche Notes will occur at a closing (the "**Second Closing**", and the date of the Second Closing the "**Second Tranche Closing Date**") to be held at the offices of Ropes & Gray LLP, 1211 Avenue of the Americas, New York, New York 10036, or at such other location as may be agreed upon by the parties hereto, subject to the terms and conditions hereof.

(iii) <u>Payment of Purchase Price</u>. Upon satisfaction or waiver of the conditions precedent specified in <u>Section 3.2</u>, at the Second Closing, against payment to Eargo by wire transfer of immediately available funds, in accordance with the wire instructions for Eargo set forth on <u>Schedule I</u> or as otherwise directed by Eargo, Eargo will deliver the Second Tranche Notes to each Noteholder, issued in the name of each such Noteholder and in the principal amount of each such Noteholder's Second Tranche Commitment.

**2.1.2 Procedures for Notes Issuance.** Eargo shall deliver to Administrative Agent a fully executed Issuance Request no later than 11:00 a.m. three (3) Business Days (or such shorter period permitted by the Administrative Agent and the Noteholders holding the applicable Commitments) prior to the First Tranche Closing Date or the Second Tranche Closing Date, as applicable, with respect to the applicable Notes. Promptly upon receipt by Administrative Agent and each Noteholder (A) may act without liability upon the basis of written or facsimile notice believed by the Administrative Agent or such Noteholder in good faith to be from Eargo (or from any Responsible Officer thereof designated in writing purportedly from Eargo to the Administrative Agent or such Noteholder), (B) shall be entitled to rely conclusively on any Responsible Officer's authority to request the purchase of Notes on behalf of Eargo until Administrative Agent or such Noteholder receives written notice to the contrary, and (C) shall have no duty to verify the authenticity of the signature appearing on any written Issuance Request.

### 2.1.3 Evidence of Debt; Register; Treatment of Interest; Notes.

(a) Evidence of Debt. The Issuer shall maintain on its internal records in accordance with its usual practice an account or accounts evidencing the Obligations of Issuer to the Noteholders. Any such recordation shall be conclusive and binding on the Noteholders, absent manifest error; *provided*, that the failure to make any such recordation, or any error in such recordation, shall not in any manner affect Issuer's Obligations in respect of any Notes; *provided*, *further*, that in the event of any inconsistency between the Register and any Issuer's records, the recordations in the Register shall govern in the absence of manifest error.

(b) <u>Register</u>. The Administrative Agent shall maintain a register for the recordation of the names and addresses of the Noteholders (including any successor or assign of any Noteholder), the Commitments, and the principal amounts of the Notes (and stated interest thereon) purchased by each Noteholder pursuant to the terms hereof from time to time (the "**Register**"). The Register shall be available for inspection by the Issuer and any Noteholder at any reasonable time and from time to time upon reasonable prior written notice. The Administrative Agent shall record in the Register the Notes, each repayment in respect of the principal amount of the Notes, and any such recordation shall be conclusive and binding on Issuer and each Noteholder, absent manifest error; *provided*, that failure to make any such recordation, or any error in such recordation, shall not affect any Noteholder's Obligations in respect of any Note. The Issuer hereby designates the Administrative Agent to serve as Issuer's non-fiduciary agent solely for purposes of maintaining the Register as provided in this <u>Section 2.1.3(b)</u>.

(c) <u>Treatment of Interest</u>. The parties intend that any interest in or with respect to the Notes under this Agreement be treated as being issued and maintained in "registered form" within the meaning of Sections 163(f), 871(h)(2), and 881(c)(2) of the Internal Revenue Code and any regulations thereunder (and any successor provisions), including under United States Treasury Regulations Section 5f.103-1(c) of Proposed Regulations Section 1.163-5 (and any successor provisions), and the provisions of this Agreement shall be construed in a manner that gives effect to such intent.

(d) <u>Exchange</u>. Any Note may be exchanged at the option of any Noteholder holding such Note for Notes of a like aggregate principal amount, but in different denominations, which in no event shall be less than \$1,000. Whenever any Notes are so surrendered for exchange, Eargo, at its expense, will execute and deliver the Notes that the Noteholder making the exchange is entitled to receive.

(e) <u>Replacement Notes</u>. If any mutilated Note is surrendered to Eargo, or if any Note is lost or stolen and Eargo receives evidence to its satisfaction of the ownership and the destruction, loss or theft of such Note, Eargo shall issue a replacement Note. If required by Eargo, an unsecured indemnity shall be supplied by the applicable Noteholder that is sufficient in the reasonable judgment of Eargo to protect Eargo from any material loss that it may suffer if a Note is replaced.

(f) <u>Restrictive Securities Legend</u>. Upon original issuance thereof, and until such time as the same is no longer required under the applicable requirements of the Securities Act, the Notes (and all Notes issued in exchange therefor or substitution thereof) shall bear the following legend:

"THIS NOTE HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "**SECURITIES ACT**"). THE HOLDER MAY NOT OFFER, SELL, TRANSFER, ASSIGN, PLEDGE, HYPOTHECATE, OR OTHERWISE DISPOSE OF OR ENCUMBER THIS NOTE EXCEPT (I) PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT OR PROSPECTUS UNDER THE SECURITIES ACT AND THE RULES AND REGULATIONS PROMULGATED THEREUNDER AND IN COMPLIANCE WITH ANY APPLICABLE STATE SECURITIES LAW, RESPECTIVELY, OR WITH AN EXEMPTION FROM SUCH REGISTRATION OR PROSPECTUS REQUIREMENT AND (II) IN COMPLIANCE WITH THAT CERTAIN NOTE PURCHASE AGREEMENT, DATED JUNE 24, 2022, AMONG THE ISSUER, THE ADMINISTRATIVE AGENT, THE NOTEHOLDERS PARTY THERETO AND THE OTHER PARTIES PARTY THERETO."

#### 2.2 Interest.

(a) <u>Interest Rate</u>. Except as increased during an Event of Default pursuant to <u>clause (d)</u> below, interest shall be payable on the principal amount of the Notes (including any PIK Interest and, to the extent not paid when due, the Repayment Value Excess) from the First Tranche Closing Date (with respect to the First Tranche Notes) or the Second Tranche Closing Date (with respect to the Second Tranche Notes) to and including the date of repayment thereof at a rate of 12.00% per annum.

(b) <u>Interest Payment Dates</u>. Interest is payable quarterly in arrears on the first (1st) calendar day of each calendar quarter, on the Maturity Date and on any Optional Redemption Date (if applicable) (each, an "**Interest Payment Date**"), and shall be computed on the basis of a three hundred sixty (360)-day year for the actual number of days elapsed in the period during which interest accrues. In computing interest, all payments received after 1:00 p.m. New York time on any day shall be deemed received at the opening of business on the next Business Day.

(c) <u>PIK Interest</u>. Other than on the Maturity Date or any Optional Redemption Date (if applicable), accrued interest shall be paid on each Interest Payment Date by capitalizing on the applicable Interest Payment Date any accrued interest (all such accrued interest capitalized from time to time, "**PIK Interest**") and adding such PIK Interest to the outstanding principal amount of the Notes. PIK Interest which is added to the outstanding principal amount of the Notes shall be ar interest in accordance with this <u>Section 2.2</u> and shall otherwise be treated as principal of such Notes for purposes of this Agreement, including being payable in cash in full on the Maturity Date and with respect to the accrual of interest on the PIK Interest amounts, in each case, through payment of the applicable Repayment Value. Accrued interest payable pursuant hereto shall be paid in cash in full on the Maturity Date and on any Optional Redemption Date (if applicable), in each case, through payment of the applicable Repayment Value.

<sup>5</sup> 

(d) <u>Default Rate</u>. Immediately upon the occurrence and during the continuance of an Event of Default, the Obligations shall bear interest at a rate per annum which is twelve percent (12.0%) above the rate that is otherwise applicable thereto (the "**Default Rate**", and any such interest "**Default Interest**"). Fees and expenses which are required to be paid by Issuer pursuant to the Note Documents but are not paid when due shall bear interest until paid at a rate equal to the highest rate applicable to the Notes. Payment or acceptance of the increased interest rate provided in this <u>Section 2.2(d)</u> is not a permitted alternative to timely payment and shall not constitute a waiver of any Event of Default or otherwise prejudice or limit any rights or remedies of the Secured Parties.

**2.3 Fees**. Unless otherwise provided in this Agreement or in a separate writing by the applicable Secured Party, Issuer shall not be entitled to any credit, rebate, or repayment of any fees earned by such Secured Party pursuant to this Agreement notwithstanding any termination of this Agreement.

#### 2.4 Payments; Application of Payments; Debit of Accounts.

**2.4.1** Payments on the Maturity Date. The Notes, together with all other amounts owed to the Noteholders hereunder with respect thereto (including accrued interest pursuant to Section 2.2, but excluding, for the avoidance of doubt, Notes which have been converted pursuant to Section 2.5 or redeemed pursuant to Section 2.6), shall be paid in full in cash on the Maturity Date in the amount of the Repayment Value. The Repayment Value shall also be payable (without duplication) in connection with the acceleration of the Obligations in accordance with Section 9 following the occurrence of any Event of Default (whether or not such payment occurs before or after (or as a result of) an Event of Default, acceleration or an insolvency proceeding). To the extent that the Issuer fails to pay the Repayment Value when due pursuant to this Section 2.4.1, without waiving any rights or remedies of the Secured Parties, the Repayment Value Excess shall be capitalized and added to the outstanding principal amount of the Notes, shall bear interest in accordance with Section 2.2, and shall otherwise be treated as principal of such Notes for purposes of this Agreement.

2.4.2 No prepayments of the Notes shall be permitted without the prior written consent of the Noteholder whose Notes are being prepaid, except upon conversion pursuant to <u>Section 2.5</u> in accordance with the terms thereof or upon redemption pursuant to <u>Section 2.6</u> in accordance with the terms thereof.

**2.4.3** All payments related to the Notes to be made by Issuer under any Note Document shall be made to the Administrative Agent (or as otherwise directed by the applicable Noteholder, so long as the Issuers provides written notice to the Administrative Agent that any such payment has been made and the amount thereof (it being understood and agreed that the Administrative Agent shall be entitled to conclusively rely on such notice of the Issuers in all respects, including, without limitation, with respect to updating the Register or calculating interest or other payments unless the applicable Noteholder notifies the Administrative Agent within two (2) Business Days of the date that such payment was due that such payment was not received by it) for the account of the applicable Noteholder, in immediately available funds in Dollars, without setoff or counterclaim except as otherwise contemplated by <u>Section 2.7</u>, before 1:00 p.m. New York time on the date when due. Payments of principal and/or interest received after 1:00 p.m. New York time shall be deemed received at the opening of business on the next Business Day. When a payment is due on a day that is not a Business Day, the payment shall be due the next Business Day, and additional fees or interest, as applicable, shall continue to accrue until paid.

## 2.5 Conversion and Redemption of Notes Upon Rights Offering.

**2.5.1** <u>Rights Offering</u>. As soon as practicable following the Stockholders Meeting and in any event within one hundred eighty (180) days of the First Tranche Closing, but only if the Stockholder Approvals are obtained, Eargo shall take all necessary action to initiate, execute and consummate a Rights Offering. For the avoidance of doubt, the terms of the Rights Offering shall include Rights to purchase 375,000,000 shares of Common Stock at a Rights Offering Price of \$0.50 per share, adjusted as needed pursuant to <u>Section 2.5.4</u> or <u>2.5.5</u> below. Eargo shall provide the Noteholders and their counsel with reasonable opportunity (in the case of draft filings with the SEC, at least ten (10) Business Days prior to such filing) to review and comment on drafts of the disclosure, transaction and other documentation related to the Rights Offering and shall consider in good faith and incorporate reasonable comments from Noteholders and their counsel.

**2.5.2** <u>Redemption and Conversion Between Five and Six Months</u>. If a Rights Offering is consummated after the 150<sup>th</sup> day following the First Tranche Closing but on or prior to the 180<sup>th</sup> day following the First Tranche Closing, then, subject to the conditions set forth in this Agreement, the Second Tranche Notes will be issued and (a) the total gross proceeds of the Rights Offering will be immediately used to repay in cash a portion of the Notes at the Repayment Value of the Notes to be redeemed, with the principal amount of the Notes redeemed being equal to the total gross proceeds of the Rights Offering *divided by* 1.50, and (b) the remaining amount of the Notes (including any accrued interest or principal comprised of PIK Interest) that are not redeemed in cash pursuant to clause (a) will immediately convert into a number of shares equal to 375,000,000 *less* the number of shares actually subscribed for and purchased in the Rights Offering. Attached hereto as <u>Annex I</u> under the heading "\$187.5 Million Rights Offering After Month Five" is an illustrative calculation of the foregoing.

**2.5.3** <u>Redemption and Conversion Within Five Months.</u> If a Rights Offering is consummated within one hundred fifty (150) days following the First Tranche Closing, then the Notes will automatically convert into (i) a number of shares of Common Stock equal to 375,000,000 *less* the number of shares actually subscribed for and purchased with the Rights Offering, and (ii) cash in an amount equal to (x) the Repayment Value of Notes outstanding *less* (y) the Rights Offering Shortfall Amount, which cash shall immediately be paid by the Issuer to the applicable Noteholders. The "**Rights Offering Shortfall Amount**" shall be an amount equal to the product of (a) 375,000,000 *less* the number of shares of Common Stock subscribed for and actually purchased in the Rights Offering, *multiplied by* (b) \$0.50. Attached hereto as <u>Annex I</u> under the heading "\$187.5 Million Rights Offering Before Month Five" is an illustrative calculation of the foregoing.

**2.5.4** <u>Subdivisions, Combinations and Other Issuances</u>. If Eargo shall at any time or from time to time prior to the Rights Offering, subdivide the shares of Common Stock as to which the Notes may be converted into pursuant to <u>Section 2.5.2</u> or <u>Section 2.5.3</u> (such shares, the "**Conversion Shares**"), by stock dividend, stock split or otherwise, or combine the Conversion Shares, the Rights Offering Price and number of Rights subject to the Rights Offering shall be adjusted such that the total potential gross proceeds from the Rights Offering is the same and the number of Conversion Shares issuable upon any such conversion shall forthwith be proportionately increased in the case of such subdivision, or proportionately decreased in the case of a combination. Any adjustment under this <u>Section 2.5.4</u> shall become effective at the close of business on the date the subdivision or combination becomes effective. Eargo shall provide Administrative Agent and the Noteholders written notice within ten (10) Business Days after any such subdivision or combination together with copies of duly executed documents evidencing the same from Eargo or its successor.

**2.5.5** <u>Reclassification, Reorganization and Consolidation</u>. In case of any reclassification, capital reorganization or change in the capital stock of Issuer (other than as a result of a subdivision or combination provided for in <u>Section 2.5.4</u> above) at any time or from time to time prior to the date of the consummation of the Rights Offering, then, as a condition of such reclassification, reorganization or change, appropriate provisions shall be made with respect to the rights and interest of the Noteholders so that the provisions hereof shall thereafter be applicable with respect to any shares of stock or other securities or property deliverable upon conversion pursuant to <u>Section 2.5.2</u> or <u>Section 2.5.3</u>, and appropriate adjustments shall be made to the Rights Offering Price and number of Rights to be issued in the Rights Offering such that the total potential gross proceeds from the Rights Offering are the same. Eargo shall provide the Noteholders written notice within ten (10) Business Days after any such reclassification, reorganization or change, together with copies of duly executed documents evidencing the same from Eargo or its successor.</u>

**2.5.6** <u>No Fractional Shares or Scrip</u>. No fractional shares or scrip representing fractional shares shall be issued pursuant to <u>Section 2.5.2</u> or <u>Section 2.5.3</u>, but in lieu of such fractional shares Eargo shall make a cash payment therefor to the Noteholders equal to the Rights Offering Price *multiplied by* such fraction.

**2.5.7** Stockholder Approval; Authorized Shares; Listing Matters. Eargo shall take all action necessary to hold a meeting of its stockholders within one hundred twenty (120) days of the First Tranche Closing (the "**Stockholders Meeting**") and shall use its commercially reasonable efforts to cause to be obtained the Stockholder Approvals at the Stockholders Meeting. Thereafter, so long as the Notes are outstanding, Eargo shall take all action necessary, including amending Eargo's governing documents to authorize and reserve and keep reserved the requisite number of shares of capital stock such that the requisite number of shares of Common Stock shall be duly and validly authorized, reserved (to the extent applicable) and available for issuance at the time of the conversion of the Notes, as applicable, and upon issuance in accordance with the terms hereof, as applicable, the Conversion Shares will be duly and validly issued, fully paid and nonassessable and free of restrictions on transfer, other than restrictions on transfer set forth in (w) the Investor Rights Agreement, as applicable, (x) Eargo's governing documents, (y) applicable federal and state securities Laws or (z) liens or encumbrances created by or imposed by the Investors. Issuer shall use its commercially reasonable efforts to (a) obtain any approvals of Nasdaq necessary for the issuance of the Conversion Shares and the transactions contemplated by this Agreement and the Investor Rights Agreement, including to cause the Conversion Shares to be approved for listing on Nasdaq, subject to official notice of issuance, and (b) maintain the listing of all of its shares of Common Stock, including the Conversion Shares. Eargo shall pay all fees and expenses in connection with satisfying the obligations under this <u>Section 2.5.8</u>.

**2.5.8** <u>Antitrust Approvals</u>. If any notifications, filings or approvals are required to be obtained under any Antitrust Laws that are applicable to the conversion of the Notes pursuant to and in accordance with <u>Section 2.5.2</u> or <u>Section 2.5.3</u>, then each of Eargo and the applicable Noteholders shall use commercially reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary or desirable under any applicable Laws to make such notifications or filings and to obtain such approvals as promptly as reasonably practicable. In such event, each of Eargo and the applicable Noteholders shall cooperate with the other party's legal advisors in the preparation and filing of any documentation, notifications, filings, registrations, submissions and other materials required or necessary under any applicable Antitrust Law and providing, within a reasonable time, all documents and information necessary to prepare and make any such filing. Eargo and the applicable Noteholders shall timely provide all information, documents and statements required by the applicable Governmental Authorities for the analysis of any such filing. All filings made pursuant to any applicable Antitrust Laws shall be made in substantial compliance with the requirements of such Antitrust Laws and any other applicable Laws. To the extent available, each of Eargo and the applicable Noteholders shall use its reasonable best efforts to cause any required filings under the Hart-Scott Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder, and any other applicable Antitrust Law. The parties acknowledge and agree that no conversion of Notes into Common Stock will be consummated until any waiting period prescribed under the Hart-Scott Rodino Antitrust Improvements Act of 1976, as amended, or any other applicable Antitrust Law has elapsed.

**2.5.9** Upon the consummation of the Rights Offering, each Noteholder shall present its Note to the Issuer for cancellation and the Issuer shall cancel such Note as of the date of the consummation of the Rights Offering.

**2.5.10** Notwithstanding anything to the contrary, other than as set forth in the second sentence of this <u>Section 2.5.10</u>, the Administrative Agent shall have no duties under this <u>Section 2.5</u> and shall have no responsibility or obligation, or be deemed to have any knowledge, with respect to the Rights Offering (including, without limitation, <u>Annex I</u> hereto) or the calculation of any amounts related thereto or any other matter described in this <u>Section 2.5</u> or in the definition of Second Tranche Commitment. The Issuer and the Noteholders shall, jointly upon the consummation of the Rights Offering, direct the Administrative Agent to update the Register to reflect the cancellation of the Notes of such Noteholders specified in such direction, and the Administrative Agent shall so update the Register in conclusive reliance upon such direction.

**2.6 Optional Redemption**. The Issuer may, at its option, repay all (but, for the avoidance of doubt, not a portion) of the Notes outstanding upon three (3) Business Days' (or such shorter period as the Administrative Agent (with the consent of the Required Noteholders) may agree) prior written notice to the Administrative Agent at a price equal to the Repayment Value of the Notes outstanding (the date of such redemption, the "**Optional Redemption Date**").

### 2.7 Withholding.

**2.7.1** Payments received (or deemed for applicable tax purposes to be received or accrued) by any Secured Party from the Issuer under this Agreement will be made free and clear of and without deduction for any and all present or future taxes, levies, imposts, duties, deductions, withholdings, assessments, fees or other charges imposed by any Governmental Authority (including any interest, additions to tax or penalties applicable thereto) (collectively, "**Taxes**"), except as required by applicable Law. Specifically, however, if at any time any Governmental Authority, applicable Law, regulation or international agreement requires Issuer or the Administrative Agent to make any withholding or deduction from any such payment or other sum payable hereunder to any Secured Party (or requires a Secured Party to make any withholding or deduction from a direct or indirect owner of a Secured Party in respect of any such payment or accrual), the Issuer hereby covenants and agrees that (a) if such withheld or deducted Tax is an Indemnified Tax, the amount due from Issuer with respect to such payment or other sum payable hereunder will be increased to the extent necessary to ensure that, after the making of such required withholding or deduction, such Secured Party (and each such direct or indirect owner of a Secured Party) receives a net sum equal to the sum which it would have received had no such withholding or deduction been required, and (b) Issuer or any other applicable withholding agent shall pay the full amount withheld or deducted by Issuer or such other applicable withholding agent, as applicable, to the relevant Governmental Authority.

**2.7.2** The Issuer will, upon request, furnish any Secured Party with proof reasonably satisfactory to such Secured Party indicating that the Issuer has made such withholding payment.

**2.7.3** The Issuer shall indemnify each Secured Party (or direct or indirect owner of a Secured Party) within ten (10) days after demand therefor, for the full amount of any Indemnified Taxes described in this section (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this Section) payable or paid by such Secured Party (or by a direct or indirect owner of a Secured Party) or required to be withheld or deducted from a payment or accrual to such Secured Party (or in respect of any payment or accrual to a direct or indirect owner of a Secured Party) and any reasonable expenses arising therefrom or with respect thereto, whether or not such taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to the Issuer by a Secured Party (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Noteholder, shall be conclusive absent manifest error.

#### 2.7.4 IRS Form W-9/W-8.

(a) Any Noteholder that is a "United States person" within the meaning of Section 7701(a)(30) of the IRC shall deliver to the Issuer and the Administrative Agent on or about the date on which such Noteholder becomes a Noteholder under this Agreement executed copies of IRS Form W-9 certifying that such Noteholder is exempt from U.S. federal backup withholding tax.

(b) Any Noteholder that is not a "United States person" within the meaning of Section 7701(a)(30) of the IRC shall, to the extent it is legally entitled to do so, deliver to the Issuer and the Administrative Agent on or about the date on which such Noteholder becomes a Noteholder under this Agreement an applicable IRS Form W-8, together with any appropriate attachments or certifications.

(c) If any Noteholder is entitled to an exemption from or reduction of withholding Tax with respect to payments made under this Agreement, such Noteholder shall deliver or cause to be delivered to the Issuer and the Administrative Agent, at the time or times reasonably requested by the Issuer or the Administrative Agent, such properly completed and executed documentation reasonably requested by the Issuer or the Administrative Agent as will permit such payments or accrual, as applicable, to be made without withholding or at a reduced rate of withholding (to the extent not provided pursuant to Section 2.7.4(a) or Section 2.7.4(b). If any direct or indirect owner of a Noteholder is entitled to an exemption from or reduction of withholding Tax with respect to any payment or accrual relating to amounts received from the Issuer under this Agreement, the Noteholder shall, to the extent it is reasonable to do so, request that such direct or indirect owner deliver or cause to be delivered to the Noteholder, such properly completed and executed documentation as will permit such payment or accrual, as applicable, to be made without withholding or at a reduced rate of withholding; and the Noteholder shall, upon the reasonable request of the Issuer or the Administrative Agent, inform the Issuer and the Administrative Agent of the portion of such payment or accrual as to which the Noteholder has received such documentation permitting any such exemption or reduction. Any Noteholder, if reasonably requested by the Issuer or the Administrative Agent, shall deliver such other documentation prescribed by applicable Law or reasonably requested by the Issuer or the Administrative Agent as will enable the Issuer or the Administrative Agent to determine whether or not such Noteholder is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in this Section 2.7.4(c), the completion, execution and submission of any documentation described in this Section 2.7.4(c) (other than the documentation set forth in Section 2.7.4(a) or (b)) shall not be required if in the applicable Noteholder's reasonable judgment such completion, execution or submission would subject such Noteholder to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender or would require the Noteholder to disclose the identity of its direct or indirect investors.

**2.7.5 Other Taxes**. The Issuer shall timely pay to the relevant Governmental Authority in accordance with applicable Law, or, at the option of the Administrative Agent or a Noteholder, timely reimburse it for the payment of, any present or future stamp, court or documentary, intangible, recording, filing or similar taxes that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, a Note.

2.7.6 The agreements and obligations of Issuer contained in this Section 2.7 shall survive the termination of this Agreement.

# 3. CONDITIONS.

**3.1 Conditions Precedent to Effectiveness**. This Agreement's effectiveness is subject to the satisfaction (or waiver) of the following conditions precedent, each to be completed in form and substance satisfactory to the Noteholders:

(a) <u>Note Documents</u>. The Administrative Agent and the Noteholders shall have received sufficient copies of each Note Document (other than the Notes and the IP Agreement), each duly executed and delivered by a Responsible Officer of each Issuer.

(b) <u>Operating Documents; Incumbency</u>. The Administrative Agent and the Noteholders shall have received a certificate for each Issuer, by such Issuer's secretary or assistant secretary, dated the Closing Date, attaching (i) copies of each Operating Document of such Issuer; (ii) signature and incumbency certificates of the officers of such Person executing the Note Documents; (iii) Borrowing Resolutions of such Issuer; and (iv) a good standing certificate from the applicable Governmental Authority of such Issuer's jurisdiction of incorporation, organization or formation, each dated a recent date prior to the Closing Date.

(c) <u>Governmental Authorizations and Consents</u>. Each Issuer shall have obtained all governmental authorizations and all consents of other Persons, in each case that are necessary or advisable in connection with the transactions contemplated by the Note Documents and each of the foregoing shall be in full force and effect and in form and substance reasonably satisfactory to the Noteholders. All applicable waiting periods shall have expired without any action being taken or threatened by any competent authority which would restrain, prevent or otherwise impose adverse conditions on the transactions contemplated by the Note Documents and no action, request for stay, petition for review or rehearing, reconsideration, or appeal with respect to any of the foregoing shall be pending, and the time for any applicable agency to take action to set aside its consent on its own motion shall have expired.

(d) Collateral. The Administrative Agent and the Noteholders shall have received:

(i) a completed Perfection Certificate dated the Closing Date and duly executed and delivered by a Responsible Officer of each Issuer, together with all attachments contemplated thereby; and

(ii) the results of a recent search of all effective UCC financing statements (or equivalent filings) made with respect to any assets or property of any Issuer in the jurisdictions specified in the Perfection Certificate, together with copies of all such filings disclosed by such search, accompanied by written evidence (including any UCC termination statements) that the Liens indicated in any such financing statements either constitute Permitted Liens or have been or, in connection with the issuance of the First Tranche Notes on the First Tranche Closing Date, will be terminated or released.

(e) <u>Opinion of Counsel to Issuer</u>. The Administrative Agent and the Noteholders shall have received an executed copy of a favorable written opinion of Latham & Watkins LLP, counsel for each Issuer, as to such matters as Administrative Agent or the Noteholders may reasonably request, dated as of the Closing Date, in form and substance satisfactory to Administrative Agent and the Noteholders (and each Issuer hereby instructs such counsel to deliver such opinion to the Administrative Agent and the Noteholders).

(f) <u>Solvency Certificate</u>. The Administrative Agent and the Noteholders shall have received a Solvency Certificate duly executed and delivered by the Chief Financial Officer of Eargo substantially in the form of <u>Exhibit B</u>, dated as of the Closing Date, certifying that, after giving effect to the consummation of the transactions contemplated herein, Eargo and its Subsidiaries, on a consolidated basis, are and will be Solvent as of the Closing Date.

(g) <u>Closing Certificate</u>. Eargo shall have delivered to the Administrative Agent a duly executed certificate of a Responsible Officer of Eargo, certifying that the conditions set forth in <u>clauses (c)</u>, (<u>h</u>), (<u>l</u>), (<u>m</u>) and (<u>n</u>) have been satisfied on the Closing Date, in form and substance satisfactory to the Administrative Agent.

(h) <u>No Litigation</u>. There shall not exist any action, suit, investigation, litigation or proceeding or other legal or regulatory developments pending or, to Issuer's knowledge, threatened in writing in any court or before any arbitrator or Governmental Authority that singly or in the aggregate materially impairs any of the transactions contemplated by the Note Documents or that could be reasonably expected to have a Material Adverse Effect.

(i) <u>Bank Regulations</u>. The Administrative Agent and the Noteholders shall have received all documentation and other information reasonably requested by such Person at least three (3) Business Days prior to the Closing Date that is required by such Person to satisfy any administrative or regulatory requirements, including as required by bank regulatory authorities under applicable "know-your-customer" and anti-money laundering rules and regulations, including the PATRIOT Act, and all such documentation and other information shall be in form and substance reasonably satisfactory to the Administrative Agent or such Noteholder.

(j) Investor Rights Agreement; Registration Rights Agreement. The Administrative Agent shall have received a duly executed copy of the Investor Rights Agreement and the Registration Rights Agreement.

(k) <u>Insurance</u>. The Administrative Agent shall have received evidence satisfactory to the Noteholders that the insurance policies and endorsements required by <u>Section 6.5</u> hereof are in full force and effect, together with appropriate evidence showing lender loss payable and/or additional insured clauses or endorsements in favor of the Administrative Agent.

(1) <u>Representations and Warranties</u>. The representations and warranties in this Agreement and in each other Note Document shall be true, accurate, and complete in all material respects on the Closing Date; *provided*, *however*, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects as of such date.

(m) No Material Adverse Effect. There shall not have occurred any Material Adverse Effect on or prior to the Closing Date.

(n) <u>No Default or Event of Default</u>. No Default or Event of Default shall exist or would result from such proposed Note issuance or from the application of the proceeds therefrom.

(o) <u>Fees and Expenses</u>. The Issuers shall have paid to the Administrative Agent or the Noteholders, as applicable, all fees and expenses (including the fees and disbursements of counsel) then due and payable on the Closing Date pursuant to <u>Section 12.9</u> (which amounts, in the case of amounts due to the Noteholders, may be offset against the proceeds of the Notes at the election of the Noteholders).

**3.2 Conditions Precedent to Purchase of First Tranche Notes**. Each applicable Noteholder's obligation to purchase the First Tranche Notes is subject to the satisfaction (or waiver) of the following conditions precedent, each to be completed in form and substance satisfactory to such Noteholders:

(a) <u>Notes</u>. The Administrative Agent and the applicable Noteholders shall have received copies of each First Tranche Note for such Noteholders in accordance with <u>Section 2.1.1(a)(iii)</u>, each duly executed and delivered by a Responsible Officer of each Issuer.

(b) <u>Operating Documents; Incumbency</u>. The Administrative Agent and each Noteholder with a First Tranche Commitment shall have received a certificate for each Issuer, by such Issuer's secretary or assistant secretary, dated the First Tranche Closing Date, attaching (i) copies of each Operating Document of such Issuer (or, with respect to the bylaws (or equivalent governing document) of each Issuer, confirming no change to such document since the last delivery of such document to the Administrative Agent); (ii) signature and incumbency certificates of the officers of such Person executing the Note Documents; (iii) Borrowing Resolutions of such Issuer; and (iv) a good standing certificate from the applicable Governmental Authority of such Issuer's jurisdiction of incorporation, organization or formation, each dated a recent date prior to the First Tranche Closing Date.

(c) <u>Governmental Authorizations and Consents</u>. Each Issuer shall have obtained all governmental authorizations and all consents of other Persons, in each case that are necessary or advisable in connection with the transactions contemplated by the Note Documents and each of the foregoing shall be in full force and effect and in form and substance reasonably satisfactory to the Noteholders. All applicable waiting periods shall have expired without any action being taken or threatened by any competent authority which would restrain, prevent or otherwise impose adverse conditions on the transactions contemplated by the Note Documents and no action, request for stay, petition for review or rehearing, reconsideration, or appeal with respect to any of the foregoing shall be pending, and the time for any applicable agency to take action to set aside its consent on its own motion shall have expired.

(d) <u>Opinion of Counsel to Issuer</u>. The Administrative Agent and the Noteholders with a First Tranche Commitment shall have received an executed copy of a favorable written opinion of Latham & Watkins LLP, counsel for each Issuer, as to such matters as such Noteholders may reasonably request, dated as of the First Tranche Closing Date, in form and substance satisfactory to Administrative Agent and such Noteholders (and each Issuer hereby instructs such counsel to deliver such opinion to the Administrative Agent and the Noteholders).

(e) <u>Solvency Certificate</u>. The Administrative Agent and the Noteholders with a First Tranche Commitment shall have received a Solvency Certificate duly executed and delivered by the Chief Financial Officer of Eargo substantially in the form of <u>Exhibit B</u>, dated as of the First Tranche Closing Date, certifying that, after giving effect to the consummation of the transactions contemplated herein, including the issuance of the First Tranche Notes on the First Tranche Closing Date and the application of proceeds therefrom, Eargo and its Subsidiaries, on a consolidated basis, are and will be Solvent.

(f) <u>Closing Certificate</u>. Eargo shall have delivered to the Administrative Agent and the Noteholders with a First Tranche Commitment a duly executed certificate of a Responsible Officer of Eargo, certifying that the conditions set forth in <u>clauses (c), (g), (k), (l)</u> and (<u>m</u>) have been satisfied on the First Tranche Closing Date, in form and substance satisfactory to the Noteholders.

(g) <u>No Litigation</u>. There shall not exist any action, suit, investigation, litigation or proceeding or other legal or regulatory developments pending or, to Issuer's knowledge, threatened in writing in any court or before any arbitrator or Governmental Authority that singly or in the aggregate materially impairs any of the transactions contemplated by the Note Documents or that could be reasonably expected to have a Material Adverse Effect.

(h) <u>Bank Regulations</u>. The Administrative Agent and the Noteholders shall have received all documentation and other information reasonably requested by such Person at least three (3) Business Days prior to the First Tranche Closing Date that is required by such Person to satisfy any administrative or regulatory requirements, including as required by bank regulatory authorities under applicable "know-your-customer" and antimoney laundering rules and regulations, including the PATRIOT Act, and all such documentation and other information shall be in form and substance reasonably satisfactory to the Administrative Agent or such Noteholder.

(i) <u>Issuance Request</u>. The Administrative Agent shall have received a fully executed and delivered Issuance Request in respect of the First Tranche Notes.

(j) <u>Silicon Valley Bank Payoff</u>. The Noteholders shall have received evidence satisfactory to them of the repayment of all Indebtedness and other obligations under that certain Loan and Security Agreement, dated as of June 6, 2018 (as amended, restated, amended and restated, supplemented or otherwise modified from time to time prior to the date hereof, the "**Existing Loan Agreement**"), between Issuer and Silicon Valley Bank, and the release and termination of all Liens and other security interest related thereto. As of the First Tranche Closing Date, after giving effect to the transactions contemplated hereby, the Issuer shall have no Indebtedness for borrowed money, other than pursuant to this Agreement.

(k) <u>Representations and Warranties</u>. The representations and warranties in this Agreement and in each other Note Document shall be true, accurate, and complete in all material respects on the First Tranche Closing Date; *provided, however*, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects as of such date.

(1) No Material Adverse Effect. There shall not have occurred any Material Adverse Effect on or prior to the First Tranche Closing Date.

(m) <u>No Default or Event of Default</u>. No Default or Event of Default shall exist or would result from such proposed Note issuance or from the application of the proceeds therefrom.

(n) <u>Fees and Expenses</u>. The Issuers shall have paid to the Administrative Agent or the Noteholders, as applicable, all fees and expenses (including the fees and disbursements of counsel) then due and payable on the First Tranche Closing Date pursuant to <u>Section 12.9</u> (which amounts, in the case of amounts due to the Noteholders, may be offset against the proceeds of the Notes at the election of the Noteholders).

(o) <u>Collateral</u>. In order to create in favor of Administrative Agent, for the benefit of Secured Parties, a valid, perfected first priority security interest (subject to Permitted Liens) in the Collateral, the Administrative Agent and the Noteholders shall have received:

(i) evidence reasonably satisfactory to the Noteholders of the compliance by each Issuer with their obligations hereunder (including their obligations to authorize and deliver UCC financing statements, originals of securities (including stock certificates, if any, representing pledged equity interests along with appropriate endorsements), instruments and chattel paper); provided that, notwithstanding any other provision of this Agreement or any other Note Document, no Issuer will be required to enter into any Control Agreement prior to the date that is thirty (30) days after the Closing Date; and

(ii) the IP Agreement, duly executed and delivered by a Responsible Officer of each Issuer.

**3.3 Conditions Precedent to Purchase of Second Tranche Notes**. Each applicable Noteholder's obligation to purchase the Second Tranche Notes is subject to the satisfaction (or waiver) of the following conditions precedent, each to be completed in form and substance satisfactory to the Noteholders:

(a) <u>No Rights Offering Within Five Months</u>; <u>Under-Subscribed Rights Offering Within Five Months</u>. Either (i) the Rights Offering shall not have occurred on or prior to the date that is one hundred fifty (150) days after the First Tranche Closing Date or (ii) if the Rights Offering has been completed on or prior to the date that is one hundred fifty (150) days after the First Tranche Closing Date, the holders of Common Stock that participate in the Rights Offering shall have subscribed for and purchased less than 75,000,000 shares of Common Stock in the Rights Offering. If <u>clause (ii)</u> has occurred, the Issuer shall have provided a certification to the Administrative Agent of the Second Tranche Commitment and an updated schedule of the Second Tranche Commitments that shall, to the extent approved by the Noteholders with a Second Tranche Commitment, amend and replace <u>Schedule I</u> hereto with respect to the Second Tranche Commitments.

(b) <u>Notes</u>. The Administrative Agent and the applicable Noteholders shall have received copies of each Second Tranche Note for such Noteholder in accordance with <u>Section 2.1.1(b)(iii)</u>, each duly executed and delivered by a Responsible Officer of each Issuer.

(c) <u>Operating Documents; Incumbency</u>. The Administrative Agent and each Noteholder with a Second Tranche Commitment shall have received a certificate for each Issuer, by such Issuer's secretary or assistant secretary, dated the Second Tranche Closing Date, attaching (i) copies of each Operating Document of such Issuer (or, with respect to the bylaws (or equivalent governing document) of each Issuer, confirming no change to such document since the last delivery of such document to the Administrative Agent); (ii) signature and incumbency certificates of the officers of such Person executing the Note Documents; (iii) Borrowing Resolutions of such Issuer; and (iv) a good standing certificate from the applicable Governmental Authority of such Issuer's jurisdiction of incorporation, organization or formation, each dated a recent date prior to the Second Tranche Closing Date.

(d) <u>Governmental Authorizations and Consents</u>. Each Issuer shall have obtained all governmental authorizations and all consents of other Persons, in each case that are necessary or advisable in connection with the transactions contemplated by the Note Documents and each of the foregoing shall be in full force and effect and in form and substance reasonably satisfactory to the Noteholders. All applicable waiting periods shall have expired without any action being taken or threatened by any competent authority which would restrain, prevent or otherwise impose adverse conditions on the transactions contemplated by the Note Documents and no action, request for stay, petition for review or rehearing, reconsideration, or appeal with respect to any of the foregoing shall be pending, and the time for any applicable agency to take action to set aside its consent on its own motion shall have expired.

(e) <u>Issuance Request</u>. The Administrative Agent shall have received a fully executed and delivered Issuance Request in respect of the Second Tranche Notes.

(f) <u>Closing Certificate</u>. Eargo shall have delivered to the Administrative Agent and the Noteholders with a Second Tranche Commitment a duly executed certificate of a Responsible Officer of Eargo, certifying that the conditions set forth in <u>clauses (d), (i), (j), (k)</u> and (l) have been satisfied on the Second Tranche Closing Date, in form and substance satisfactory to the Noteholders.

(g) <u>Solvency Certificate</u>. The Administrative Agent and each Noteholder with a Second Tranche Commitment shall have received a Solvency Certificate duly executed and delivered by the Chief Financial Officer of Eargo substantially in the form of <u>Exhibit B</u>, dated as of the Second Tranche Closing Date, certifying that, after giving effect to the issuance of the Second Tranche Notes on the Second Tranche Closing Date and the application of proceeds therefrom, Eargo and its Subsidiaries, on a consolidated basis, are and will be Solvent;

(h) <u>Opinion of Counsel to Issuer</u>. The Administrative Agent and the Noteholders with a Second Tranche Commitment shall have received an executed copy of a favorable written opinion of Latham & Watkins LLP, counsel for each Issuer, as to such matters as Administrative Agent or such Noteholders may reasonably request, dated as of the Second Tranche Closing Date, in form and substance satisfactory to Administrative Agent and such Noteholders (and each Issuer hereby instructs such counsel to deliver such opinion to the Administrative Agent and the Noteholders).

(i) <u>No Litigation</u>. There shall not exist any action, suit, investigation, litigation or proceeding or other legal or regulatory developments pending or, to Issuer's knowledge, threatened in writing in any court or before any arbitrator or Governmental Authority that singly or in the aggregate materially impairs any of the transactions contemplated by the Note Documents or that could be reasonably expected to have a Material Adverse Effect.

(j) <u>Representations and Warranties</u>. The representations and warranties in this Agreement and in each other Note Document shall be true, accurate, and complete in all material respects on the Second Tranche Closing Date; *provided*, *however*, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects as of such date.

(k) <u>No Material Adverse Effect</u>. There shall not have occurred any Material Adverse Effect on or prior to the Second Tranche Closing Date.

(1) <u>No Default or Event of Default</u>. No Default or Event of Default shall exist or would result from such proposed Note issuance or from the application of the proceeds therefrom.

(m) <u>Fees and Expenses</u>. The Issuers shall have paid to the Administrative Agent or the Noteholders, as applicable, all fees and expenses (including the fees and disbursements of counsel) then due and payable on the Second Tranche Closing Date pursuant to <u>Section 12.9</u> (which amounts, in the case of amounts due to the Noteholders, may be offset against the proceeds of the Notes at the election of the Noteholders).

(n) Second Tranche Commitment Termination Date. The Second Tranche Commitment Termination Date shall not have occurred.

# 4. CREATION OF SECURITY INTEREST.

**4.1 Grant of Security Interest**. As of the First Tranche Closing Date, to secure the prompt payment and performance in full of all of the Obligations, Issuer hereby grants to the Administrative Agent, for the benefit of the Secured Parties, a continuing security interest in and Lien upon the Collateral, wherever located, whether now owned or hereafter acquired or arising, and all proceeds and products thereof. If this Agreement and the Commitments of the Noteholders are terminated, the Administrative Agent's Lien in the Collateral shall continue until the Obligations) are repaid in full in cash. Upon payment in full in cash of the Obligations (other than inchoate indemnity obligations) and the termination of the Commitments, the Administrative Agent shall, at the request of and at the sole cost and expense of Issuer, promptly release its Liens in the Collateral pursuant to documentation prepared by the Issuer in form and substance reasonably satisfactory to the Administrative Agent and all rights therein shall revert to the Issuer.

**4.2 Priority of Security Interest**. Issuer represents, warrants, and covenants that the security interest granted herein is and shall at all times continue to be a first priority perfected security interest in the Collateral (subject only to Permitted Liens that are expressly permitted pursuant to the terms of this Agreement to have superior priority to the Administrative Agent's Lien under this Agreement). If Issuer shall acquire a commercial tort claim greater than \$100,000, Issuer shall promptly notify the Administrative Agent in a writing signed by Issuer of the general details thereof and grant to the Administrative Agent in such writing a security interest therein and in the proceeds thereof, all upon the terms of this Agreement, with such writing to be in form and substance reasonably satisfactory to the Required Noteholders.

**4.3** Authorization to File Financing Statements. Issuer hereby authorizes (without obligation) the Administrative Agent to file, at the Issuer's expense, financing statements (and any amendments, modifications or continuations with respect thereto) relating to the Collateral, without notice to the Issuer, with all appropriate jurisdictions to perfect or protect the Secured Parties' interest or rights hereunder, which may (a) indicate the Collateral (i) as "all assets" of the Issuer or words of similar effect, if appropriate, regardless of whether any particular asset comprised in the Collateral falls within the scope of Article 9 of the Code or (ii) by specific Collateral category and (b) provide any other information required by part 5 of Article 9 of the UCC for the sufficiency or filing office acceptance of any financing statement or amendment. Issuer authorizes (without obligation) the Administrative Agent to execute any such financing statements and/or control agreements in the Issuer's name and to perform all other acts, at the Issuer's expense, which the Required Noteholders deem appropriate to perfect and to continue perfection of the security interest of the Administrative Agent.

# 5. <u>REPRESENTATIONS AND WARRANTIES</u>

Each Issuer represents and warrants as follows:

### 5.1 Due Organization, Authorization; Power and Authority; Authorized and Outstanding Stock.

(a) Issuer is duly existing and in good standing as a Registered Organization in its jurisdiction of formation, and is qualified and licensed to do business and is in good standing in any jurisdiction in which the conduct of its business or its ownership of property requires that it be qualified except where the failure to do so could not reasonably be expected to have a Material Adverse Effect. In connection with this Agreement, Issuer has delivered to the Administrative Agent a completed certificate signed by Issuer, entitled "Perfection Certificate" (the "**Perfection Certificate**"). Issuer represents and warrants to the Administrative Agent that (a) Issuer's exact legal name is that indicated on the Perfection Certificate and on the signature page hereof; (b) Issuer is an organization of the type and is organized in the jurisdiction set forth in the Perfection Certificate; (c) the Perfection Certificate accurately sets forth Issuer's organizational identification number or accurately states that Issuer has none; (d) the Perfection Certificate accurately sets forth Issuer's place of business, or, if more than one, its chief executive office as well as Issuer's mailing address (if different than its chief executive office); (e) Issuer (and each of its predecessors) has not, in the past five (5) years, changed its jurisdiction of formation, organizational structure or type, or any organizational number assigned by its jurisdiction; and (f) all other information set forth on the Perfection Certificate pertaining to Issuer is accurate after the Closing Date to the extent permitted by one or more specific provisions in this Agreement).

(b) The execution, delivery and performance by Issuer of the Note Documents to which it is a party have been duly authorized, and do not (i) conflict with any of Issuer's organizational documents, (ii) contravene, conflict with, constitute a default under or violate any material Requirement of Law, (iii) contravene, conflict or violate any applicable order, writ, judgment, injunction, decree, determination or award of any Governmental Authority by which Issuer or any of its property or assets may be bound or affected, (iv) require any action by, filing, registration, or qualification with, or Governmental Approval from, any Governmental Authority (except (a) such Governmental Approvals which have already been obtained and are in full force and effect (or are being obtained pursuant to Section 6.1(b)), (b) the filing of a UCC-1 financing statement with the Delaware and/or California Secretaries of State covering the Collateral in connection with the security interests granted herein, (c) the filing of a Form D and current report on Form 8-K with the SEC, (d) filings with applicable state securities commissions and (e) the listing of the Conversion Shares with Nasdaq) or (v) conflict with, contravene, constitute a default or breach under, or result in or permit the termination or acceleration of, any material agreement by which Issuer is bound. Issuer is not in default under any agreement to which it is a party or by which it is bound in which the default could reasonably be expected to have a Material Adverse Effect.

(c) As of the date of this Agreement, the authorized capital stock of Eargo consists of 300,000,000 shares of common stock, \$0.0001 par value per share ("**Common Stock**") and 5,000,000 shares of preferred stock, par value \$0.0001 per share ("**Preferred Stock**").

(d) As of May 31, 2022, (i) no shares of Preferred Stock were issued and outstanding, (ii) 39,358,558 shares of Common Stock were issued and outstanding, (iii) 2,098,050 shares of Common Stock were subject to outstanding restricted stock units granted pursuant to the Stock Plans, (iv) options to purchase 5,209,086 shares of Common Stock granted pursuant to the Stock Plans were outstanding, (v) 7,307,136 shares of Common Stock were reserved for issuance upon the exercise of outstanding stock options or the vesting of restricted stock units granted pursuant to the Stock Plans, and (vi) the amount of unrestricted cash presented on the balance sheet of Eargo is \$36,738,797. Except as set forth in this Section, there are no outstanding options, warrants, convertible securities, subscription rights, conversion rights, exchange rights or other agreements that require the Company to issue or sell any of its equity ownership interests (or securities convertible into or exchangeable for shares of its equity ownership interests).

(e) All of the issued and outstanding shares of Common Stock are and, when issued in accordance with the terms of <u>Section 2.5.2</u> or <u>Section 2.5.3</u>, as applicable, will be duly authorized and validly issued and fully paid and non-assessable. Subject only to the Stockholder Charter Approval, the Conversion Shares will be reserved for issuance and, when issued in accordance with the terms of this Agreement, will be validly issued and fully paid and non-assessable and will not be subject to any preemptive right or any restrictions on transfer under applicable Law or any contract to which Eargo is a party, other than those under applicable state and federal securities and antitakeover Laws, this Agreement, the Investor Rights Agreement. When issued in accordance with the terms hereof, the Conversion Shares will be free and clear of all Liens (other than Liens incurred by the Noteholders, restrictions arising under applicable securities Laws, or restrictions imposed by this Agreement, the Investor Rights Agreement or the Registration Rights Agreement).

(f) Except as set forth in the Section 5.1(f) of the Disclosure Schedule or as otherwise expressly described in this Agreement: (i) no subscription, warrant, option, convertible security or other right issued by Eargo to purchase or acquire any shares of capital stock of Eargo is authorized or outstanding; (ii) there is not any commitment of Eargo to issue any subscription, warrant, option, convertible security or other such right or to issue or distribute to holders of any shares of its capital stock; (iii) Eargo has no obligation to purchase, redeem or otherwise acquire any shares of its capital stock or to pay any dividend or make any other distribution in respect thereof; and (iv) there are no agreements between Eargo and any holder of its capital stock relating to the acquisition, disposition or voting of the capital stock of Eargo. No person or entity is entitled to any preemptive right granted by Eargo with respect to the issuance of any capital stock of Eargo.

(g) This Agreement has been, and the Investor Rights Agreement and Registration Rights Agreement, at the First Tranche Closing will be, duly executed and delivered by the Issuer. Assuming due execution and delivery thereof by each of the other parties hereto or thereto, this Agreement, the Investor Rights Agreement and the Registration Rights Agreement will each be a valid and binding obligation of the Issuer enforceable against the Issuer in accordance with its terms, except as such enforceability may be limited by applicable laws relating to bankruptcy, insolvency, reorganization, moratorium or other similar legal requirement relating to or affecting creditors' rights generally and except as such enforceability is subject to general principles of equity (regardless of whether enforceability is considered in a proceeding in equity or at law). The Issuer has taken all appropriate actions so that the restrictions on business combinations contained in Section 203 of the Delaware General Corporation Law will not apply with respect to or as a result of the issuance of the Notes (or the Conversion Shares) to the Noteholders or any permitted transfer thereof in accordance with this Agreement, without any further action on the part of the stockholders or the board of directors of the Issuer.

# 5.2 Collateral.

(a) The Issuer has good and valid title to, rights in, and the power to transfer each item of the Collateral upon which it purports to grant a Lien hereunder, free and clear of any and all Liens except Permitted Liens. The Issuer has no Collateral Accounts at or with any bank or financial institution other than as permitted in accordance with <u>Section 6.6</u> hereof. To the Issuer's knowledge, the Accounts are bona fide, existing obligations of the Account Debtors.

(b) The Collateral is not in the possession of any third party bailee (such as a warehouse) except as otherwise provided in the Perfection Certificate or as otherwise permitted under <u>Section 7.2</u>. None of the components of the Collateral shall be maintained at locations other than as provided in the Perfection Certificate or as permitted pursuant to <u>Section 7.2</u>; *provided*, *however*, the foregoing shall not apply to any location that maintains Collateral with an aggregate value less than \$10,000.

(c) Issuer is the sole owner of the Intellectual Property which it owns or purports to own and otherwise has a valid and enforceable right to use all Intellectual Property used in its business, except where the failure to own or have the right to use such Intellectual Property would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. To Issuer's knowledge, each registration for Intellectual Property which it owns or purports to own and which is material to Issuer's business is subsisting, valid and enforceable, and no part of the Intellectual Property which Issuer owns or purports to own and which is material to Issuer's business has been judged invalid or unenforceable, in whole or in part. To the best of Issuer's knowledge, no claim has been made or threatened that any part of the Intellectual Property infringes, misappropriates or otherwise violates the rights of any third party except to the extent such claim would not reasonably be expected to have a Material Adverse Effect.

(d) Except as noted on the Perfection Certificate or as otherwise disclosed to the Administrative Agent pursuant to <u>Section 6.8(b)</u>, Issuer is not a party to, nor is it bound by, any Restricted License.

(e) Upon (i) the filing of appropriate financing statements and the recordation of the IP Agreement with the U.S. Patent and Trademark Office and the U.S. Copyright Office, as applicable, all actions necessary to perfect the security interest, so far as perfection is possible under relevant Laws, in the Collateral of the Issuer created under this Agreement with respect to which a Lien may be perfected by filing pursuant to the UCC or 35 U.S.C. §261, 15 U.S.C. §1060 or 17 U.S.C. §205 shall have been duly made or taken and are in full force and effect, and this Agreement creates in favor of the Administrative Agent for the benefit of the Secured Parties a valid and, together with such filings, perfected, so far as perfection is possible under relevant Laws through making such filings, first priority security interest in such Collateral of the Collateral with respect to which a security interest may be perfected only by possession or control by the Administrative Agent of the Collateral of the Issuer created under this Agreement with respect to which a Lien may be perfected by taking of possession or control, all actions necessary to perfect the security interest, so far as perfection is possible under relevant Laws, in such Collateral of the Issuer created under this Agreement with respect to which a Lien may be perfected by taking of possession or control shall have been duly made or taken, and this Agreement creates in favor of the Administrative Agent for the benefit of the Secured Parties a valid and, together with possession or control so far as perfection is possible under relevant Laws, in such Collateral of the Issuer (subject to Permitted Liens), security interest in such Collateral of the Issuer (subject to Permitted Liens), security interest in such Collateral of the Issuer (subject to Permitted Liens), securing the payment of the Obligations.

(f) The Issuer represents and warrants that all certificates or instruments representing or evidencing the equity interests constituting Collateral in existence on the date hereof have been delivered to the Administrative Agent in suitable form for transfer by delivery or accompanied by duly executed undated instruments of transfer or assignment in blank, and that (assuming continuing possession by the Administrative Agent) the Administrative Agent has a perfected first priority security interest therein.

**5.3 Litigation**. Except as set forth in Section 5.3 of the Disclosure Schedule or other than those of which Issuer has notified the Administrative Agent pursuant to Section 6.2(i), there are no actions or proceedings pending or, to the knowledge of any Responsible Officer, threatened in writing by or against the Issuer that could reasonably be expected to result in liability of the Issuer involving more than, individually or in the aggregate, \$250,000. Neither the Issuer nor any of its Subsidiaries is in default with respect to any order, writ, injunction, decree, ruling or decision of any court, commission, board or other government agency that is expressly applicable to the Issuer or any of its Subsidiaries, which would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

**5.4 Financial Statements; Financial Condition**. All consolidated financial statements for the Issuer delivered to the Noteholders fairly present in all material respects Issuer's consolidated financial condition and Issuer's consolidated results of operations as of the dates and for the periods set forth therein. There has not been any material deterioration in Issuer's consolidated financial condition since the date of the most recent financial statements delivered to the Administrative Agent or otherwise filed with the SEC.

**5.5 Solvency**. The fair salable value of Issuer's consolidated assets (including goodwill minus disposition costs) exceeds the fair value of Issuer's liabilities. Issuer is not left with unreasonably small capital after the transactions in this Agreement; and Issuer is able to pay its debts (including trade debts) as they mature ("**Solvent**"). The Noteholders have provided the Issuer with reasonably equivalent value with respect to the Collateral and the Liens and security interests granted hereunder. The Repayment Value Excess shall be disregarded for purposes of any determination made pursuant the first sentence of this <u>Section 5.5</u>.

**5.6 Regulatory Compliance**. Issuer is not, and immediately after giving effect to the sale of the Notes in accordance with this Agreement and the application of the proceeds thereof will not be required to be registered as, an "investment company" or a company "controlled" by an "investment company," within the meaning of the Investment Company Act of 1940, as amended. Issuer is not engaged as one of its important activities in extending credit for margin stock (under Regulations X, T and U of the Federal Reserve Board of Governors). Except as set forth in <u>Section 5.6</u> of the Disclosure Schedule, Issuer (a) has complied in all material respects with all Requirements of Law, and (b) has not violated any Requirements of Law the violation of which could reasonably be expected to have a Material Adverse Effect. None of Issuer's properties or assets has been used by Issuer or, to the best of Issuer's knowledge, by previous Persons, in disposing, producing, storing, treating, or transporting any hazardous substance other than in compliance in all material respects with Requirements of Law. The Issuer has obtained all consents, approvals and authorizations of, made all declarations or filings with, and given all notices to, all Governmental Authorities that are necessary to continue their respective businesses as currently conducted, except to the extent that failure to do so could not reasonably be expected to have a Material Adverse Effect.

**5.7 Subsidiaries; Investments**. The Issuer does not own any stock, partnership, or other ownership interest or other equity securities except for Permitted Investments. The Issuer, directly or indirectly, owns of record and beneficially free and clear of all Liens, all of the issued and outstanding capital stock or equity interests of each of its Subsidiaries. All of the issued and outstanding capital stock or equity interest of the Issuer's Subsidiaries has been duly authorized and validly issued, and in the case of corporations, is fully paid and non-assessable. There are no outstanding rights, options, warrants, preemptive rights, conversion rights, rights of first refusal or similar rights for the purchase or acquisition from any of the Issuer's Subsidiaries of any securities of such Subsidiaries nor are there any commitments to issue or execute any such rights, options, warrants, preemptive rights, conversion rights.

**5.8 Tax Returns and Payments; Pension Contributions**. The Issuer is not a U.S. real property holding corporation within the meaning of Section 897(c)(2) of the IRC. Except as set forth in <u>Section 5.8</u> of the Disclosure Schedule, the Issuer has timely filed (taking into account valid extensions to file) all required tax returns and reports, and Issuer has timely paid all foreign, federal, state and local taxes, assessments, deposits and contributions owed by Issuer except (a) to the extent such taxes are being contested in good faith by appropriate proceedings promptly instituted and diligently conducted, so long as such reserve or other appropriate provision, if any, as shall be required in conformity with GAAP shall have been made therefor, or (b) if such taxes, assessments, deposits and contributions do not, individually or in the aggregate, exceed \$150,000.

To the extent Issuer defers payment of any contested taxes, Issuer shall (i) notify the Administrative Agent in writing of the commencement of, and any material development in, the proceedings, and (ii) post bonds or take any other steps reasonably required to prevent the governmental authority levying such contested taxes from obtaining a Lien upon any of the Collateral that is more than \$150,000 (other than a Permitted Lien). Issuer is unaware of any claims or adjustments proposed for any of Issuer's prior tax years which could result in additional taxes becoming due and payable by Issuer in excess of \$150,000. Issuer has paid all amounts necessary to fund all present pension, profit sharing and deferred compensation plans in accordance with their terms, and Issuer has not withdrawn from participation in, and has not permitted partial or complete termination of, or permitted the occurrence of any other event with respect to, any such plan which could reasonably be expected to result in any material liability of Issuer, including any liability to the Pension Benefit Guaranty Corporation or its successors or any other governmental agency.

**5.9** Use of Proceeds. Issuer shall use the proceeds of the Notes solely as working capital, to fund its general business requirements and to repay Indebtedness of the Issuer outstanding pursuant to the Existing Loan Agreement, and not for personal, family, household or agricultural purposes.

#### 5.10 Full Disclosure.

(a) No written representation, warranty or other statement of Issuer in any report, certificate, or written statement submitted to the Administrative Agent or the Noteholders in connection with this Agreement or the other Note Documents, as of the date such representation, warranty, or other statement was made, taken together with all such written reports, written certificates and written statements submitted to the Administrative Agent or the Noteholders in connection with this Agreement or the other Note Documents, contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements contained in the reports, certificates, or written statements not misleading (it being recognized by the Administrative Agent and the Noteholders that the projections and forecasts provided by Issuer in good faith and based upon reasonable assumptions are not viewed as facts and that actual results during the period or periods covered by such projections and forecasts may differ from the projected or forecasted results).

(b) Except as set forth in Section 5.10 of the Disclosure Schedule, since January 1, 2021, Eargo has timely filed (a) all annual and quarterly reports and proxy statements (including all amendments, exhibits and schedules thereto) and (b) all other reports and other documents (including all amendments, exhibits and schedules thereto), in each case required to be filed by Eargo with the SEC pursuant to the Exchange Act and the Securities Act except, in the case of clause (b), where the failure to file has not had, and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. As of their respective filing dates, such SEC Documents complied in all material respects with the requirements of the Securities Act, the Exchange Act and the rules and regulations of the SEC thereunder applicable to such SEC Documents, and as of their respective dates, none of the SEC Documents contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The financial statements of Eargo included in the SEC Documents comply as of their respective dates in all material respects with applicable accounting requirements and the rules and regulations of the SEC with respect thereto (except as may be indicated in the notes thereto or, in the case of the unaudited statements, as permitted by Form 10-Q promulgated by the SEC), and present fairly in all material respects as of their consolidated cash flows for each of the respective periods, all in conformity with GAAP (except as may be indicated in such financial statements or the notes thereto).

# 5.11 Securities Act.

(a) Subject in part to the truth and accuracy of each Noteholder's representations set forth in <u>Section 14</u> of this Agreement, the offer, sale and issuance of the Notes as contemplated by this Agreement are exempt from the registration requirements of any applicable state and federal securities Laws, including the Securities Act, and none of the Issuer or any authorized agent acting on its behalf will take any action hereafter that would cause the loss of such exemption.

(b) Neither the Issuer nor any Person acting on its behalf has offered or sold the Notes by any form of general solicitation or general advertising (within the meaning of Regulation D under the Securities Act) or in any manner involving a public offering, in violation of the Securities Act.

(c) None of the Issuer or any Person acting on its behalf has sold or issued any securities that would be integrated with the offering of the Notes contemplated hereby pursuant to the Securities Act, the rules and regulation thereunder or the interpretations thereof by Securities and Exchange Commission.

(d) When the Notes are issued pursuant to this Agreement, such Notes will not be of the same class (within the meaning of Rule 144A under the Securities Act) as securities of the Issuer that are listed on a national securities exchange registered under Section 6 of the Securities Exchange Act of 1934, as amended, or that are quoted in a United States automated inter-dealer quotation system.

(e) Any shares of common stock of the Issuer issuable upon conversion of the Notes, when issued upon conversion of the Notes will be duly authorized, fully paid and nonassessable and not subject to any Liens or other restrictions (other than those arising under applicable securities Laws) or any preemptive, option to purchase, or similar rights.

(f) Prior to the date hereof, the Issuer has exercised reasonable care, in accordance with Securities and Exchange Commission rules and guidance, to determine whether any Covered Person (as defined below) is subject to any of the "bad actor" disqualifications described in Rule 506(d)(1) (i) through (viii) under the Securities Act ("**Disqualification Events**"). To the Issuer's knowledge, no Covered Person is subject to a Disqualification Event event, except for a Disqualification Event by Rule 506(d)(2) or (d)(3) under the Securities Act. The Issuer has complied, to the extent applicable, with any disclosure obligations under Rule 506(e) under the Securities Act. "**Covered Persons**" are those persons specified in Rule 506(d) (1) under the Securities Act, including the Issuer, any predecessor or Affiliate of the Issuer, any director, executive officer, or other officer of the Issuer participating in the offering, general partner or managing member of the Issuer, any beneficial owner of 20% or more of the Issuer's outstanding voting equity securities, calculated on the basis of voting power, any promoter (as defined in Rule 405 under the Securities Act) connected with the Issuer in any capacity at the time of the sale of the Notes (a "Solicitor"), any general partner or managing member of any Solicitor, and any director, executive officer or other officer participating in the offering of any Solicitor or general partner or managing member of any Solicitor.

**5.12** No Brokers or Finders. The Issuer has not retained, utilized or been represented by, or otherwise become obligated to, any broker, placement agent, financial advisor or finder in connection with the transactions contemplated hereby whose fees the Noteholders would be required to pay.

5.13 Cybersecurity; Data Protection. Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, the Issuer's information technology assets and equipment, computers, systems, networks, hardware, software, websites, applications, and databases (collectively, "IT Systems") are adequate for, and operate and perform as required in connection with the operation of the business of the Issuer as currently conducted, free and clear of all material bugs, errors, defects, Trojan horses, time bombs, malware and other corruptants. The Issuer has implemented and maintained commercially reasonable controls, policies, procedures, and safeguards designed to maintain and protect their material confidential information and the integrity, continuous operation, redundancy and security of all IT Systems and data (including all personal, personally identifiable, sensitive, confidential or regulated data (including protected health information) ("Personal Data")) collected, used, stored or processed in connection with their businesses, and except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, there have been no breaches, violations, outages or unauthorized uses of or accesses to same, nor any incidents under internal review or investigations relating to the same. Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, the Issuer has been for the past three (3) years and are presently in compliance with all applicable Laws or statutes (including the Health Insurance Portability and Accountability Act of 1996, as amended by the Health Information Technology for Economic and Clinical Health Act ("HIPAA")) and all judgments, orders, rules and regulations of any court or arbitrator or governmental or regulatory authority, internal and external privacy policies, applicable industry standards and contractual obligations relating to the privacy and security of IT Systems and Personal Data and to the protection of such IT Systems and Personal Data from unauthorized use, access, misappropriation or modification and regarding the collection, use, transfer, import, export, storage, disposal and disclosure by the Issuer of

Personal Data ("**Data Protection Obligations**"). The Issuer has not received any written notification of or written complaint regarding, or is aware of any other facts that, individually or in the aggregate, would reasonably indicate non-compliance in any material respect with any Data Protection Obligation. There is no pending, or to the knowledge of the Issuer, threatened, action, suit or proceeding by or before any court or governmental agency, authority or body pending or threatened alleging non-compliance in any material respect with any Data Protection Obligation. The Issuer has taken commercially reasonable steps in accordance with industry standard practices (including implementing and monitoring compliance with adequate measures with respect to technical and physical security) designed to protect Personal Data collected, used or otherwise processed by the Issuer. Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, the Issuer has made all disclosures to users or customers required by applicable Laws and regulatory rules or requirements and no such disclosures have been inaccurate or in violation of any applicable Laws or regulatory rules and requirements.

#### 5.14 Compliance with Health Care Laws.

(a) The Issuer has been for the past three (3) years and is in compliance with all applicable Health Care Laws, except where such noncompliance has not had or would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. Except as set forth in <u>Section 5.14</u> of the Disclosure Schedule, since April 1, 2019: (i) the Issuer has not received any Form 483, notice of adverse finding, warning letter, untitled letter or other correspondence or notice from the FDA, or other written notice of any claim, action, suit, proceeding, hearing, enforcement, investigation, arbitration or other action from any court or arbitrator or governmental authority alleging a material violation of any Health Care Laws, and, to the Issuer's knowledge, no such claim, action, suit, proceeding, hearing, enforcement, investigation, arbitration or other action is threatened, (ii) the Issuer is not a party to or has ongoing reporting obligations pursuant to any corporate integrity agreements, deferred prosecution agreements, monitoring agreements, consent decrees, settlement orders, plans of correction or similar agreements with or imposed by any governmental authority and (iii) none of the Issuer or its respective employees, officers or directors, or, to the knowledge of the Issuer, agents, has been excluded, suspended or debarred from participation in any U.S. federal health care program or, to the knowledge of the Issuer, is subject to a governmental inquiry, investigation, proceeding or other similar action that could reasonably be expected to result in debarment, suspension, or exclusion.

(b) (i) There have been no recalls, field notifications, field corrections, market withdrawals or replacements, warnings, "dear doctor" letters, investigator notices, safety alerts or other notice of action relating to an alleged lack of safety, efficacy, or regulatory compliance of the Issuer's products ("**Safety Notice**"), except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, and (ii) to the Issuer's knowledge, there are no facts that would be reasonably likely to result in a Safety Notice with respect to the Issuer's products or services, or a termination or suspension of marketing or testing of any of the Issuer's products, except, in each case, such as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

(c) The research studies and clinical trials conducted by or on behalf of, or sponsored by, the Issuer, or in which the Issuer has participated, were and, if still pending, are being, conducted in all material respects in accordance with all applicable Health Care Laws. The Issuer has not received any written notices or other written correspondence from any governmental authority or institutional review board requiring the termination, suspension or material adverse modification of any research studies or clinical trials conducted by or on behalf of, or sponsored by, the Issuer or in which the Issuer has participated and, to the Issuer's knowledge, there are no reasonable grounds for the same.

**5.15 Employee Matters.** Except where the failure to comply has not had, and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, the Issuer has been for the past three (3) years and is in compliance with all applicable Laws relating to labor, employment, fair employment practices, terms and conditions of employment, and wages and hours, and with the terms of the ERISA Documents, and each such ERISA Document is in compliance with all applicable requirements of ERISA.

**5.16 Environmental Matters.** The Issuer has been for the past three (3) years and are in compliance with all applicable Requirements of Environmental Law and required Environmental Permits, except, in each case, where the failure to comply has not had, and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. The Issuer has not received within the past three (3) years any written notice from any governmental authority of any violation or alleged violation of any Requirements of Environmental Law or Environmental Permit in connection with their respective properties, except as has not had, and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

**5.17 Registration Rights.** Except as set forth in <u>Section 5.17</u> of the Disclosure Schedule or as provided in this Agreement or the Registration Rights Agreement or disclosed in the SEC Documents, the Issuer has not granted any rights to register under the Securities Act any of its presently outstanding securities or any of its securities that may be issued subsequently.

**5.18 Intellectual Property.** The Issuer owns or has the valid and enforceable right to use all Intellectual Property used in the conduct of their respective businesses as currently conducted except where any failure to own or have the valid and enforceable right would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. To the knowledge of the Issuer, (a) the conduct of the businesses of the Issuer does not infringe the Intellectual Property right of any third party and (b) the Intellectual Property owned by the Issuer is not being infringed by any third party, except in each case for any infringement that would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

**5.19 Nasdaq.** As of the date hereof, Eargo's Common Stock is listed on Nasdaq, and no event has occurred, and Eargo is not aware of any event that is reasonably likely to occur, that would result in the Common Stock being delisted from Nasdaq.

**5.20 Illegal Payments; FCPA Violations.** Except as has not had, and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, none of the Issuer or, to the knowledge of the Issuer, any officer, director, employee, agent, representative or consultant acting on behalf of the Issuer (and only in their capacities as such) has, in connection with the business of the Issuer: (a) unlawfully offered, paid, promised to pay, or authorized the payment of, directly or indirectly, anything of value, including money, loans, gifts, travel, or entertainment, to any Government Official with the purpose of (i) influencing any act or decision of such Government Official in his or her official capacity; (ii) inducing such Government Official to perform or omit to perform any activity in violation of his or her legal duties; (iii) securing any improper advantage; or (iv) inducing such Government Official to influence or affect any act or decision of such Governmental Authority, except, with respect to the foregoing <u>clauses (i)</u> through (<u>iv)</u>, as permitted under the U.S. Foreign Corrupt Practices Act of 1977, as amended, or other applicable Law; (b) made any illegal contribution to any political party or candidate; (c) made, offered or promised to pay any unlawful bribe, payoff, influence payment, kickback, unlawful rebate, or other similar unlawful payment of any nature, directly or indirectly, in connection with the business of the Issuer, to any Person, including any supplier or customer; (d) knowingly established or maintained any unrecorded fund or asset or made any false entry on any book or record of the Issuer for any purpose; or (e) otherwise violated the U.S. Foreign Corrupt Practices Act of 1977, as amended, the UK Bribery Act 2010, as amended, or any other applicable anti-corruption or anti-bribery Law.

**5.21 Economic Sanctions.** Except as has not had, and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, the Issuer is not in contravention of any sanction, and has not engaged in any conduct sanctionable, under U.S. economic sanctions Laws, including applicable Laws administered and enforced by the U.S. Department of the Treasury's Office of Foreign Assets Control, 31 C.F.R. Part V, the Iran Sanctions Act, as amended, the Comprehensive Iran Sanctions, Accountability and Divestment Act, as amended, the Iran Threat Reduction and Syria Human Rights Act, as amended, the Iran Freedom and Counter-Proliferation Act of 2012, as amended, and any executive order issued pursuant to any of the foregoing.

**5.22** Absence of Changes. Since January 1, 2021, there has not been any change, development, occurrence or event that has had or would reasonably be expected to have a Material Adverse Effect.

# 6. AFFIRMATIVE COVENANTS

The Issuer shall do all of the following:

# 6.1 Government Compliance.

(a) Except as otherwise permitted by <u>Section 7.3</u>, maintain its legal existence and good standing in its jurisdiction of formation, and maintain qualification in each jurisdiction in which the failure to so qualify would reasonably be expected to have a Material Adverse Effect the Issuer's business or operations. The Issuer shall comply in all material respects, with all Laws, ordinances and regulations to which it is subject.

(b) Obtain all of the Governmental Approvals necessary for the performance by the Issuer of its obligations under the Note Documents to which it is a party and the grant of a security interest to the Administrative Agent in the Collateral. The Issuer shall promptly provide copies of any such obtained Governmental Approvals to the Administrative Agent.

(c) The Issuer will comply, and shall cause each of its Subsidiaries and all other Persons, if any, on or occupying any real property, to comply, with the requirements of all applicable Laws, rules, regulations and orders of any Governmental Authority, non-compliance with which could reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

6.2 Financial Statements; Reports. Provide the Administrative Agent with the following:

(a) <u>Monthly Management Report</u>. As soon as available, but no later than thirty (30) days after the last day of each month, commencing with the month ending May 31, 2022, a monthly management report of Eargo and its Subsidiaries covering operations for such month, in such form prepared by Eargo's management team consistent with its past practice and reasonably acceptable to the Required Noteholders (it being understood that the sample management report provided by Eargo to the Noteholders on June 10, 2022 is reasonably acceptable to the Required Noteholders as of the date hereof).

(b) <u>Quarterly Financial Statements</u>. As soon as available, but no later than forty-five (45) days after the last day of each fiscal quarter, commencing with the fiscal quarter ending June 30, 2022, an unaudited consolidated balance sheet, income statement, and statement of cash flows, in each case, of Eargo and its Subsidiaries, covering operations for such fiscal quarter in a form consistent with those filed with the SEC.



(c) <u>Annual Audited Financial Statements</u>. As soon as available, but no later than ninety (90) days after the last day of each fiscal year of Eargo, commencing with the fiscal year ending December 31, 2022, an audited consolidated balance sheet, income statement, and statement of cash flows, in each case, of Eargo and its Subsidiaries, prepared in accordance with GAAP, consistently applied, together with an unqualified opinion from an independent certified public accounting firm of nationally recognized standing reasonably acceptable to the Required Noteholders with respect to such audited financial statements for such fiscal year.

(d) <u>Compliance Statement</u>. Together with delivery of financial statements pursuant to <u>clause (a)</u> (other than with respect to the third month of any fiscal quarter), (b) and (c) above, a completed Compliance Statement, confirming that, as of the end of such month, fiscal quarter or fiscal year, as applicable, the Issuer was in full compliance with the terms and conditions of this Agreement, and including such other information as the Required Noteholders may reasonably request.

(e) <u>Annual Operating Budget and Financial Projections</u>. Within sixty (60) days after the last day of each fiscal year of Eargo, (A) annual operating budgets (including income statements, balance sheets and cash flow statements, by month) for the upcoming fiscal year of the Issuer and (B) annual financial projections for the following fiscal year (on a quarterly basis) as approved by the board of directors of Eargo, together with any related business forecasts used in the preparation of such annual financial projections, in each case, in form and substance reasonably acceptable to the Required Noteholders and approved in accordance with the Investor Rights Agreement.

(f) <u>Other Statements</u>. Within five (5) days of delivery, copies of all statements, reports and notices made available to holders of Eargo's capital stock generally in their capacity as such;

(g) <u>SEC Filings</u>. Within five (5) days of filing, copies of all periodic and other reports, proxy statements and other materials filed by the Issuer with the SEC, any Governmental Authority succeeding to any or all of the functions of the SEC or with any national securities exchange.

(h) Legal Action Notice. A prompt report of any legal actions pending or threatened in writing against the Issuer that could reasonably be expected to result in damages or costs to the Issuer of, individually or in the aggregate, \$250,000 or more.

(i) <u>Notice of Change in Board of Directors</u>. Within four (4) Business Days, written notice of any change in the board of directors (or similar governing body) of Eargo.

(j) <u>Information Regarding Collateral</u>. Promptly, written notice if any material portion of the Collateral is damaged or destroyed. The Issuer agrees not to effect or permit any change in its legal name, organizational structure, jurisdiction of formation, or tax identification number, unless previously approved in writing by the Required Noteholders and all filings have been made under the UCC or otherwise that are required (as determined by the Required Noteholders) in order for Administrative Agent to continue at all times following such change to have a valid, legal and perfected security interest in all of the Collateral and for the Collateral at all times following such change to have a valid, legal and perfected security interest as contemplated in the Collateral Documents.

(k) <u>Notice of Default</u>. Promptly (but in any event within three (3) Business Days) upon any officer of the Issuer obtaining knowledge (i) of any condition or event that constitutes a Default or an Event of Default or that notice has been given to the Issuer with respect thereto or (ii) of the occurrence of any event or change that has caused or would reasonably be expected to result in, in any case or in the aggregate, a Material Adverse Effect, in each case, a certificate of a Responsible Officer of the Issuer specifying the nature and period of existence of such condition, event or change, or specifying the notice given and action taken by any such Person and the nature of such claimed Event of Default, Default, default, event or condition, and what action the Issuer has taken, is taking and proposes to take with respect thereto.

(1) Other Information. Other information reasonably requested by the Administrative Agent or any Noteholder.

Any submission by the Issuer of a Compliance Statement or any financial statement to the Administrative Agent shall be deemed to be a representation by the Issuer that (i) as of the date of such Compliance Statement or financial statement, the financial information and calculations set forth therein are true, accurate, and correct, (ii) as of the end of the compliance period forth in such submission, the Issuer is in complete compliance with all required covenants set forth herein and in the other Note Documents, except as noted in such Compliance Statement or financial statement; (iii) as of the date of such submission, no Events of Default have occurred or are continuing; (iv) all representations and warranties, other than any representations or warranties that are made as of a specific date in <u>Section 5</u>, remain true and correct in all material respects as of the date of such submission (other than any representation or warranty that is qualified by materiality, Material Adverse Effect or words of similar import, in which case such representations and warranties remain true and correct in all respects), except as noted in such Compliance Statement or financial statement; (v) as of the date of such submission, the Issuer has timely filed all required tax returns and reports, and the Issuer has timely paid all foreign, federal, state and local taxes, assessments, deposits and contributions owed by the Issuer, except as otherwise permitted pursuant to the terms of <u>Section 5.9</u>; and (vi) as of the date of such submission, no Liens have been levied or claims made against the Issuer relating to unpaid employee payroll or benefits of which the Issuer has not previously provided written notification to the Administrative Agent.

Documents required to be delivered pursuant to the terms hereof (x) may be delivered electronically and, if so delivered, shall be deemed to have been delivered on the date on which Eargo posts such documents or (y) shall be deemed to have been delivered to the Administrative Agent on the date on which the Issuer posts such documents to EDGAR (or such other free, publicly-accessible internet database that may be established and maintained by the SEC as a substitute for or successor to EDGAR), in each case of <u>clauses (x) and (y)</u>, so long as the Issuer immediately notifies the Administrative Agent and the Noteholders in writing (which may be by electronic mail) of the posting of any such documents and provides a link thereto.

**6.3 Inventory; Returns**. Keep all Inventory in good and marketable condition, free from material defects. Returns and allowances between the Issuer and its Account Debtors shall follow the Issuer's customary practices as they exist on the Closing Date. The Issuer must promptly notify the Administrative Agent of all returns, recoveries, disputes and claims in respect of the Issuer's Inventory that involve more than \$300,000.

**6.4 Taxes; Pensions.** Timely file (taking into account valid extensions to file) all required tax returns and reports and timely pay all foreign, federal, state and local taxes, assessments, deposits and contributions owed by the Issuer, except for (i) deferred payment of any taxes contested pursuant to the terms of <u>Section 5.8</u> hereof and (ii) up to \$150,000 in other taxes so long as no Lien has been filed in connection with the same, and shall deliver to the Administrative Agent appropriate certificates attesting to such payments, and pay all amounts necessary to fund all present pension, profit sharing and deferred compensation plans in accordance with their terms.

#### 6.5 Insurance.

(a) Keep its business and the Collateral insured for risks in amounts standard for companies in the Issuer's industry and location and as the Required Noteholders may reasonably request. Insurance policies shall be in a form, with financially sound and reputable insurance companies that are not Affiliates of Issuer, and in amounts that are reasonably satisfactory to the Required Noteholders. All property policies shall have a lender's loss payable endorsement showing the Administrative Agent as the sole loss payee. All liability policies shall show, or have endorsements showing, the Administrative Agent as an additional insured. The Administrative Agent shall be named as loss payee and/or additional insured with respect to any such insurance providing coverage in respect of any Collateral.

(b) Ensure that proceeds payable under any property policy are, at the Noteholders' option, payable to the Administrative Agent on account of the Obligations. Notwithstanding the foregoing, (i) so long as no Event of Default has occurred and is continuing, the Issuer shall have the option of applying the proceeds of any casualty policy up to \$500,000 with respect to any loss, but not exceeding \$1,000,000 in the aggregate for all losses under all casualty policies in any one year, toward the replacement or repair of destroyed or damaged property, *provided* that any such replaced or repaired Collateral (A) shall be of equal or like value as the replaced or repaired Collateral and (B) shall be deemed Collateral in which the Administrative Agent has been granted a first priority security interest (subject to Permitted Liens) and (ii) after the occurrence and during the continuance of an Event of Default, all proceeds payable under such casualty policy shall, at the option of Noteholders, be payable to the Administrative Agent on account of the Obligations.

(c) At the Administrative Agent's request (at the direction of the Required Noteholders), the Issuer shall deliver certified copies of insurance policies and evidence of all premium payments. Each provider of any such insurance required under this <u>Section 6.5</u> shall agree, by endorsement upon the policy or policies issued by it or by independent instruments furnished to the Administrative Agent, that it will give the Administrative Agent thirty (30) days prior written notice (or ten (10) days for cancellation as a result of nonpayment of premium) before any such policy or policies shall be canceled. If the Issuer fails to obtain insurance as required under this <u>Section 6.5</u> or to pay any amount or furnish any required proof of payment to third persons and the Administrative Agent, the Administrative Agent or the Required Noteholders may make all or part of such payment or obtain such insurance policies required in this <u>Section 6.5</u> and take any action, or direct the Administrative Agent to take any action, under the policies that the Required Noteholders deem prudent.

**6.6 Operating Accounts**. For each Collateral Account that the Issuer at any time maintains, the Issuer shall cause the applicable bank or financial institution at or with which any Collateral Account is maintained to execute and deliver a Control Agreement with respect to such Collateral Account to perfect the Administrative Agent's Lien in such Collateral Account in accordance with the terms hereunder, which Control Agreement may not be terminated without the prior written consent of the Administrative Agent (acting at the direction of the Required Noteholders); provided, that, with respect to any Collateral Account existing on the Closing Date, the Issuer shall not be considered to have breached this <u>Section 6.6</u> until the date that is thirty (30) days after the Closing Date (or such later date as is approved in writing by the Required Noteholders). The provisions of the previous sentence shall not apply to (a) deposit accounts exclusively used for payroll, payroll taxes and other employee wage and benefit payments to or for the benefit of the Issuer's employees and identified to the Administrative Agent by the Issuer as such and (b) accounts that are maintained at all times on a "zero balance" basis. The Issuer shall also be permitted to maintain merchant accounts (including Stripe, Ally Financial, Bread, PayPal, CareCredit, LLC, CyberSource / Visa and Covered Care (the foregoing being as described in the Perfection Certificate) (collectively, the "**Merchant Accounts**")), *provided* that Merchant Accounts shall automatically sweep on a daily basis to a Collateral Account subject to a Control Agreement.

**6.7 Cash Management**. Maintain cash and Cash Equivalents on deposit in Collateral Accounts (which, for the avoidance of doubt, are subject to Control Agreements in accordance with Section 6.6) in an aggregate amount of no less than \$40,000,000 at all times. The Issuer agrees, at any time and from time to time, upon the request of the Required Noteholders, to provide evidence reasonably satisfactory to such Person or Persons of the Issuer's compliance with this Section 6.7.

## 6.8 Protection of Intellectual Property Rights.

(a) Take all actions reasonably necessary to (i) maintain, and file and pursue each application to obtain, the registration of each of its material Patents, Trademarks, Domain Names and, to the extent consistent with past practices, Copyrights, including by filing applications for renewal, affidavits of use, affidavits of noncontestability and, if reasonably necessary (taking into account the projected cost of such proceedings versus the expected benefit thereof), by initiating opposition and interference and cancellation proceedings against third parties, (ii) maintain and protect the secrecy or confidentiality of its material Trade Secrets and (iii) otherwise protect, defend and preserve Issuer's rights in, and the validity or enforceability of, its material Intellectual Property.

(b) Promptly advise the Administrative Agent in writing of any infringement, misappropriation or other violation of Issuer's Intellectual Property or any other event in each case of which it becomes aware (including the institution of, or any determination or development in, any proceeding in the United States Patent and Trademark Office, the United States Copyright Office, court or other similar governmental body) that could reasonably be expected to materially and adversely affect the value of, or the Administrative Agent's ability to dispose of its rights and remedies related to, Issuer's Intellectual Property, and take such actions that, in Issuer's commercially reasonable business judgment, are reasonable and appropriate under the circumstances to protect such Intellectual Property (including collecting damages for such infringement, misappropriation or other violation).

(c) Not allow any Intellectual Property material to Issuer's business to lapse or be abandoned, forfeited or dedicated to the public without the Required Noteholder's prior written consent.

(d) Provide written notice to the Administrative Agent within thirty (30) days of entering or becoming bound by any Restricted License (other than over-the-counter software that is commercially available to the public).

(e) To the extent not already disclosed in the Perfection Certificate or the IP Agreement, if the Issuer (i) obtains any Patent, registered Trademark, registered Copyright, registered mask work, registered Domain Name or any pending application for any of the foregoing, as owner or an exclusive licensee or (ii) applies for any Patent or the registration of any Trademark, Copyright or mask work, then the Issuer shall immediately provide written notice thereof to the Administrative Agent and shall execute such intellectual property security agreements (or a supplement to the IP Agreement in form and substance satisfactory to the Required Noteholders) and other documents and take such other actions as the Required Noteholders may request to perfect and maintain a first priority perfected security interest in favor of the Administrative Agent in such property. If the Issuer decides to register any Patent or Trademark with the United States Patent and Trademark Office or Copyright or mask work with the United States Copyright Office, the Issuer shall: (x) provide the Administrative Agent with at least fifteen (15) days' prior written notice of the Issuer's intent to register such Intellectual Property, together with a copy of the application it intends to file with such office; (y) execute an intellectual property security agreement (or a supplement to the IP Agreement in form and substance satisfactory to the Required Noteholders) and such other documents and take such other actions as the Required Noteholders may request to perfect and maintain a first priority perfected security interest in favor of the Administrative Agent in such Intellectual Property intended to be registered with the applicable office; and (z) record such intellectual property security agreement (or such supplement to the IP Agreement) with the applicable office contemporaneously with the filing of the Intellectual Property with the applicable office. The Issuer shall promptly provide to the Administrative Agent copies of all applications that it files for Patents or for the registration of Trademarks or Copyrights or mask works, together with evidence of the recording of the intellectual property security agreement required for the perfection or maintenance of perfection of the Administrative Agent's first priority perfected security interest in such property.

**6.9 Litigation Cooperation**. From the date hereof and continuing through the termination of this Agreement, make available to the Administrative Agent or any Noteholder, without expense to the Administrative Agent or such Noteholder, the Issuer and its officers, employees and agents and the Issuer's books and records to the extent that the Administrative Agent or such Noteholder may deem them reasonably necessary to prosecute or defend any third-party suit or proceeding instituted by or against the Administrative Agent or such Noteholder with respect to any Collateral or relating to the Issuer.

**6.10 Access to Collateral; Books and Records.** Allow the Administrative Agent or any Noteholder, or their agents, at reasonable times during reasonable business hours, on five (5) Business Days' prior written notice (*provided* that no notice shall be required if an Event of Default has occurred and is continuing), to inspect the Collateral and audit and copy the Issuer's Books. The foregoing inspections and audits shall occur no more than once every twelve (12) months (unless an Event of Default is continuing, in which case, the foregoing may occur as often as the Administrative Agent or such Noteholder reasonably deems necessary) and shall be at Issuer's expense.

**6.11 Joinder of Subsidiaries; Pledge of Subsidiaries**. Notwithstanding and without limiting the negative covenants contained in <u>Sections 7.3</u> and <u>7.7</u> hereof, (x) with respect to any Subsidiary owned by the Issuer on the Closing Date or (y) at the time that the Issuer forms any direct or indirect Subsidiary or acquires any direct or indirect Subsidiary after the Closing Date (including pursuant to a Division), the Issuer shall (a) cause any such Subsidiary to deliver to the Administrative Agent a joinder (in form and substance satisfactory to the Required Noteholders) to this Agreement to cause such Subsidiary to become an "Issuer" hereunder, together with such appropriate financing statements, Control Agreements, and such other documents or instruments as the Required Noteholders may reasonably request, all in form and substance satisfactory to the Required Noteholders (including being sufficient to grant the Administrative Agent a first priority Lien (subject to Permitted Liens) in and to the Collateral of such Subsidiary), (b) deliver to the Administrative Agent equiry interest certificates and blank transfer powers in order to pledge all of the equity interest held by such Issuer in such Subsidiary, each in form and substance satisfactory to the Required Noteholders, each in form and substance satisfactory to the Required Noteholders, including one or more opinions of counsel to such Issuer and such Subsidiary satisfactory to the Administrative Agent, which in the opinion of the Administrative Agent is appropriate with respect to the execution and delivery of the applicable documentation referred to above. Any document, agreement, or instrument executed or issued pursuant to this <u>Section 6.11</u> shall be a Note Document.

## 6.12 Other Equity Interests; Instruments; Commercial Tort Claims.

(a) If the Issuer shall at any time hold or acquire any certificated securities (other than equity interests of a Subsidiary pledged to the Administrative Agent pursuant to Section 6.11), promissory notes, tangible chattel paper, negotiable documents or warehouse receipts relating to the Collateral, in each case, with a value in excess of \$250,000, the Issuer shall promptly endorse, assign and deliver the same to the Administrative, accompanied by such instruments of transfer or assignment duly executed in blank as the Administrative may from time to time specify.

(b) If the Issuer shall at any time hold or acquire a commercial tort claim with a value in excess of \$250,000, the Issuer shall promptly notify the Administrative Agent in a writing signed by the Issuer of the particulars thereof and grant to the Administrative Agent in such writing a security interest therein and in the proceeds thereof, all upon the terms of this Agreement, with such writing to be in form and substance satisfactory to the Required Noteholders.

**6.13 Further Assurances**. Execute any further instruments and take further actions as the Required Noteholders reasonably request to perfect or continue the Administrative Agent's Lien in the Collateral or to effect the purposes of this Agreement or the other Note Documents.

**6.14 Tax Treatment**. The Issuer and the Noteholders agree that the Notes shall be treated for U.S. federal, state, and local income and franchise tax purposes as reasonably determined by the Noteholders in consultation with KPMG or another nationally recognized accounting firm reasonably acceptable to the Issuer, which determination as to treatment shall include (but is not limited to) determinations in respect of interest accruals or payments on the Notes and a conversion, redemption, sale or exchange, or other disposition of the Note, in whole or in part. The Noteholders, in making such determination, shall in good faith consult with the Issuer and consider any comments from the Issuer regarding such determination. The parties hereto agree to treat and report the Notes consistent therewith in any and all filings with any U.S. federal, state or local taxing authority, except as required by a "determination" within the meaning of IRC section 1313(a) (or similar state or local tax provision). The inclusion of this <u>Section 6.14</u> is not an admission by any Noteholder that it is subject to United States taxation. The Issuer and the Noteholders agree that this <u>Section 6.14</u> shall survive termination of this Agreement and/or the repayment or conversion of the Notes, and, to the extent that any Noteholder no longer constitutes a "Noteholder" after the date hereof, such Noteholder shall be a third-party beneficiary of this <u>Section 6.14</u>.

## 7. <u>NEGATIVE COVENANTS</u>

Issuer shall not do any of the following without the Administrative Agent's (acting at the direction of the Required Noteholders) prior written consent:

**7.1 Dispositions**. Convey, sell, lease, transfer, assign, license, sublicense, option or otherwise dispose of (including pursuant to a Division) (collectively, "**Transfer**") all or any part of its assets, except for Transfers (a) of Inventory in the ordinary course of business; (b) of worn-out or obsolete Equipment that is, in the reasonable judgment of the Issuer, no longer economically practicable to maintain or useful in the ordinary course of business of the Issuer; (c) consisting of Permitted Liens and Permitted Investments, including licenses permitted hereunder; (d) consisting of the sale or issuance of any stock or other equity interest of the Issuer permitted under <u>Section 7.2</u>; (e) consisting of the Issuer's use or transfer of money or Cash Equivalents in a manner that is not prohibited by the terms of this Agreement or the other Note Documents; (f) consisting of transactions permitted by <u>Section 7.3</u>; (g) consisting of dividends, distributions, redemptions, and other payments permitted under <u>Section 7.7(a)</u>; and (h) of other assets of the Issuer that do not exceed \$150,000 in the aggregate in any fiscal year.

**7.2 Changes in Business, Management, Control, or Business Locations**. (a) Engage in any business other than the businesses currently engaged in by the Issuer or reasonably related thereto; (b) liquidate or dissolve; (c) other than for cause, remove, demote, retire, voluntarily cease the employment relationship of, promote or otherwise change the employment relationship of Eargo with, any Key Person, or hire or otherwise add any Key Person to the management of Eargo, in each case, without the prior written consent of the Required Noteholders; *provided, however*, that the foregoing shall not apply if any Key Person terminates their employment relationship with Eargo (other than for purposes of avoiding non-compliance with this Section 7.2); or (d) permit or suffer any Change in Control, unless the Obligations (other than inchoate indemnity obligations) have been repaid in full in cash (with respect to the Notes, pursuant to Section 2.6) and all Commitments have been terminated upon the consummation of such Change in Control. The Issuer shall not: (1) add any new offices or business locations without providing written notice to the Administrative Agent within thirty (30) days of such addition, including warehouses (unless such new offices or business locations contain less than \$500,000 of the Issuer's assets (taken as a whole)), or deliver any portion of the Collateral valued, individually or in the aggregate, in excess of

\$500,000 to a bailee, other than to a bailee at a location already disclosed in the Perfection Certificate, (2) without the prior written consent of the Administrative Agent, change its jurisdiction of organization, (3) without the prior written consent of the Administrative Agent, change its organizational structure or type, (4) without at least ten (10) days prior written notice to the Administrative Agent, change its jurisdiction of organization at least ten (10) days prior written notice to the Administrative Agent, change its jurisdiction of organization.

**7.3 Mergers or Acquisitions**. Merge or consolidate with any other Person, or acquire all or substantially all of the capital stock or property of another Person (including by the formation of any Subsidiary or pursuant to a Division) without the prior written consent of the Required Noteholders (in their sole discretion); *provided, however*; that such prior written consent shall not be required if the Obligations (other than inchoate indemnity obligations) have been repaid in full in cash (with respect to the Notes, pursuant to <u>Section 2.6</u>) and all Commitments have been terminated upon the consummation of such merger, consolidation, or acquisition.

7.4 Indebtedness. Create, incur, assume, or be liable for any Indebtedness, other than Permitted Indebtedness.

**7.5 Encumbrances**. Create, incur, allow, or suffer any Lien on any of its assets, or assign or convey any right to receive income, including the sale of any Accounts, except for Permitted Liens and Transfers permitted by Section 7.1(f) or (h), permit any Collateral not to be subject to the first priority security interest granted herein, or enter into any agreement, document, instrument or other arrangement (except with or in favor of the Administrative Agent) with any Person which directly or indirectly prohibits or has the effect of prohibiting the Issuer from assigning, mortgaging, pledging, granting a security interest in or upon, or encumbering any of the Issuer's Intellectual Property in favor of the Administrative Agent, except (a) as is otherwise permitted in Section 7.1 and the definition of "Permitted Liens", (b) customary restrictions in permitted license agreements on the licensed property where the Issuer is the licensee and not the licensor, and (c) covenants with such restrictions in contracts of sale or merger or acquisition agreements, *provided* that, with respect to clause (c), (i) such covenants do not prohibit or restrict the Issuer from granting a security interest in any Collateral, and (iii) on or prior to the consummation of such sale, acquisition or merger, the Issuer shall have either (x) repaid the Obligations (other than inchoate indemnity obligations) in full in cash (with respect to the Notes, pursuant to Section 2.6) and terminated all Commitments or (y) obtained the Required Noteholders' prior written consent to such transaction (which consent shall be at the Required Noteholders' sole discretion).

7.6 Maintenance of Collateral Accounts. Maintain any Collateral Account except pursuant to the terms of Section 6.6(b) hereof.

**7.7 Distributions; Investments.** (a) Pay any dividends or make any distribution or payment or redeem, retire or purchase any capital stock or other equity interest, *provided* that the Issuer may (i) convert any of its convertible securities (including warrants) into other securities pursuant to the terms of such convertible securities or otherwise in exchange thereof, (ii) pay dividends or distributions solely in common stock with the prior written consent of the Required Noteholders, (iii) repurchase the stock of former employees, directors, officers, or consultants pursuant to stock purchase agreements so long as an Event of Default does not exist at the time of such repurchase and would not exist after giving effect to such repurchase, *provided* that the aggregate amount of all such repurchases by the Issuer does not exceed \$100,000 per fiscal year, (iv) with the prior written consent of the Required Noteholders, make purchases of capital stock in connection with the exercise of stock options or stock appreciation rights by way of a cashless exercise, (v) purchase capital stock or options to acquire such capital stock with the proceeds (*provided* that the amount of such proceeds exceeds the amount of such purchases) received from a

substantially concurrent issuance of capital stock or convertible securities, *provided* that (x) such purchases do not in the aggregate exceed \$100,000 per fiscal year and (y) no Event of Default is continuing or would result therefrom; (vi) make cash payments in lieu of the issuance of fractional shares, and (vii) grant stock-settled equity awards to employees, officers, independent contractors or directors pursuant to incentive equity plans existing on, and pursuant to the terms thereof as of, the Closing Date (including the Stock Plans), and distribute equity securities to employees, officers, independent contractors or other awards granted pursuant to incentive equity plans existing on, and pursuant to the terms thereof as of, the Closing Date (including the Stock Plans), in each case, with the consent of the Required Noteholders; or (b) directly or indirectly make any Investment (including by the formation of any Subsidiary) other than Permitted Investments.

**7.8 Transactions with Affiliates**. Directly or indirectly enter into or permit to exist any material transaction with any Affiliate of the Issuer, except for (a) transactions that are in the ordinary course of the Issuer's business, upon fair and reasonable terms that are no less favorable to the Issuer than would be obtained in an arm's length transaction with a non-affiliated Person; (b) transactions permitted pursuant to the terms of <u>Section 7.7</u> of this Agreement; (c) transactions under <u>clauses (a), (f)</u>, or (g), of the definition of "Permitted Investments"; (d) employment or compensation arrangements and employee benefit plans approved by the Issuer's board of directors and entered into in the ordinary course of business prior to fiscal year 2021, and (e) transactions between or among the Issuers in the ordinary course of business and not otherwise prohibited hereunder.

7.9 Cash Management. From and after the Closing Date, exceed \$25,000,000 in Cash Burn.

**7.10 Compliance**. Become an "investment company" or a company controlled by an "investment company", under the Investment Company Act of 1940, as amended, or undertake as one of its important activities extending credit to purchase or carry margin stock (as defined in Regulation U of the Board of Governors of the Federal Reserve System), or use the proceeds of any Notes for that purpose; fail to (a) meet the minimum funding requirements of ERISA, (b) permit a Reportable Event or Prohibited Transaction, as defined in ERISA to occur, or (c) fail to comply with the Federal Fair Labor Standards Act or violate any other Law or regulation, in each case, if the violation could reasonably be expected to have a Material Adverse Effect; withdraw from participation in, permit partial or complete termination of, or permit the occurrence of any other event with respect to any present pension, profit sharing and deferred compensation plan which could reasonably be expected to result in any material liability of the Issuer, including any liability to the Pension Benefit Guaranty Corporation or its successors or any other governmental agency.

# 8. EVENTS OF DEFAULT

Any one of the following shall constitute an event of default (an "Event of Default") under this Agreement:

**8.1 Payment Default**. The Issuer fails to (a) make any payment of principal or interest on any Note when due or (b) pay any other Obligations within three (3) Business Days after such Obligations are due and payable.

## 8.2 Covenant Default.

(a) The Issuer fails or neglects to perform any obligation in <u>Sections 6.1(a)</u>, <u>6.2</u>, <u>6.4</u>, <u>6.5</u>, <u>6.6</u>, <u>6.7</u>, <u>6.10</u>, <u>6.11</u>, violates any covenant in <u>Section 7</u>; or

(b) the Issuer fails or neglects to perform, keep, or observe any other term, provision, condition, covenant or agreement contained in this Agreement or any other Note Document, and as to any default (other than those specified in this <u>Section 8</u>) under such other term, provision, condition, covenant or agreement that can be cured, has failed to cure such default within ten (10) days after the occurrence thereof; *provided*, *however*, that if such Default cannot by its nature be cured within such ten (10) day period and such Default is likely (in the sole judgment of the Required Noteholders) to be cured within a reasonable time, then Issuer shall have an additional period determined by the Required Noteholders in their sole discretion to cure such Default (*provided*, that any such cure period provided under this <u>clause (b)</u> shall not apply to any covenant set forth in <u>clause (a)</u> above).

8.3 Material Adverse Effect. A Material Adverse Effect occurs.

### 8.4 Attachment; Levy; Restraint on Business.

(a) (i) The service of process seeking to attach, by trustee or similar process, any funds of the Issuer or of any entity under the control of the Issuer (including a Subsidiary) in excess of \$100,000, or (ii) a notice of lien or levy is filed against any of the Issuer's assets with a value in excess of \$100,000 by any Governmental Authority, and the same under <u>subclauses (i)</u> and (<u>ii)</u> hereof are not, within ten (10) Business Days after the occurrence thereof, discharged or stayed (whether through the posting of a bond or otherwise); or

(b) (i) any material portion of the Issuer's assets is attached, seized, levied on, or comes into possession of a trustee or receiver, or (ii) any court order enjoins, restrains, or prevents the Issuer from conducting all or any material part of its business.

**8.5 Insolvency**. (a) The Issuer is unable to pay its debts (including trade debts) as they become due or otherwise becomes not Solvent; (b) the Issuer begins an Insolvency Proceeding; or (c) an Insolvency Proceeding is begun against the Issuer and is not dismissed or stayed within forty-five (45) days.

**8.6 Other Agreements**. There is, under any agreement to which the Issuer is a party with a third party or parties, any default resulting in a right by such third party or parties, whether or not exercised, to accelerate the maturity of any Indebtedness in an amount individually or in the aggregate in excess of \$300,000.

**8.7 Judgments; Penalties**. One or more fines, penalties or final judgments, orders or decrees for the payment of money in an amount, individually or in the aggregate, of at least \$300,000 (which is not covered by independent third-party insurance as to which liability has not been rejected by such insurance carrier) shall be rendered against the Issuer by any Governmental Authority, and the same are not, within ten (10) Business Days after the entry, assessment or issuance thereof, discharged, satisfied, or paid, or after execution thereof, stayed or bonded pending appeal, or such judgments are not discharged prior to the expiration of any such stay.

**8.8 Misrepresentations**. The Issuer or any Person acting for the Issuer makes any representation, warranty, or other statement in this Agreement, any other Note Document, or in any writing delivered or previously delivered to the Administrative Agent or any Noteholder that, when taken as a whole, is incorrect or misleading in any material respect when made.

**8.9 Collateral Documents and other Note Documents**. This Agreement or any Collateral Document ceases to be in full force and effect (other than by reason of a release of Collateral in accordance with the terms hereof or thereof or the satisfaction in full of the Obligations in accordance with the terms hereof) or shall be declared null and void, or Administrative Agent shall not have or shall cease to have a valid and perfected Lien in any Collateral purported to be covered by the Collateral Documents with the priority required by the relevant Collateral Document, in each case for any reason other than the failure of Administrative Agent or any Secured Party to take any action within its control, or (iii) the Issuer shall contest the validity or enforceability of any Note Document in writing or deny in writing that it has any further liability under any Note Document to which it is a party.

**8.10 Governmental Approvals.** Any Governmental Approval shall have been (a) revoked, rescinded, suspended, modified in an adverse manner or not renewed in the ordinary course for a full term or (b) subject to any decision by a Governmental Authority that designates a hearing with respect to any application for renewal of any of such Governmental Approval or that could result in the Governmental Authority taking any of the actions described in <u>clause (a)</u> above, and such decision or such revocation, rescission, suspension, modification or non-renewal (i) causes, or could reasonably be expected to cause, a Material Adverse Effect or (ii) adversely affects the legal qualifications of the Issuer to hold such Governmental Approval in any applicable jurisdiction and such revocation, rescission, suspension, modification or non-renewal could reasonably be expected to affect the status of or legal qualifications of the Issuer to hold any Governmental Approval in any other jurisdiction.

**8.11 Cessation of Business**. The Issuer is enjoined, restrained or in any way prevented by the order of any court or any Governmental Authority from conducting all or any material part of their consolidated business for more than fifteen (15) days.

**8.12 Proceedings**. The conviction of the Issuer under any criminal statute, pursuant to which statute the penalties or remedies sought or available include forfeiture to any Governmental Authority of any material portion of the property of such Person.

**8.13 Nasdaq Listing.** The Issuer receives written notice of delisting from Nasdaq or the Issuer otherwise fails to maintain the listing of its Common Stock on Nasdaq.

**8.14 Rights Offering**. The Issuer fails to consummate a Rights Offering on the terms specified herein prior to the end of the day that is six (6) months after the Closing Date.

**8.15 Change in Control**. The Issuer undergoes or is otherwise subject to a Change in Control; *provided*, *however*, that the foregoing shall not result in an Event of Default so long as the Obligations (other than inchoate indemnity obligations) have been repaid in full in cash (with respect to the Notes, pursuant to <u>Section 2.6</u>) and all Commitments have been terminated upon the consummation of such Change in Control.

## 9. <u>REMEDIES</u>

**9.1 Remedies**. Upon the occurrence and during the continuance (after the expiration of any cure period provided in <u>Section 8.1</u> or <u>8.2</u> hereof, if applicable) of an Event of Default, the Administrative Agent, at the direction of the Required Noteholders, shall, without notice or demand, do any or all of the following:

(a) declare all Obligations (including the Repayment Value) to be immediately due and payable; <u>provided</u>, that, if an Event of Default described in <u>Section 8.5</u> has occurred, all Obligations (including the Repayment Value) shall be immediately due and payable without any action by any party;

## (b) terminate the Commitments;

(c) verify the amount of, demand payment of and performance under, and collect any Accounts and General Intangibles, settle or adjust disputes and claims directly with Account Debtors for amounts on terms and in any order that the Administrative Agent (acting at the direction of the Required Noteholders) considers advisable, and notify any Person owing the Issuer money of the Administrative Agent's security interest in such funds;

(d) make any payments and do any acts that the Required Noteholders consider necessary or reasonable to protect the Collateral and/or its security interest in the Collateral. The Issuer shall assemble the Collateral if the Administrative Agent (acting at the direction of the Required Noteholders) requests and make it available as the Administrative Agent (acting at the direction of the Required Noteholders) designates. The Administrative Agent may enter premises where the Collateral is located, take and maintain possession of any part of the Collateral, and pay, purchase, contest, or compromise any Lien which appears to be prior or superior to its security interest and pay all expenses incurred. The Issuer grants the Administrative Agent a license to enter and occupy any of its premises, without charge, to exercise any of the Administrative Agent's rights or remedies;

(e) apply to the Obligations any (i) balances and deposits of the Issuer in its possession or which it controls or (ii) amount held by the Administrative Agent owing to or for the credit or the account of Issuer;

(f) ship, reclaim, recover, store, finish, maintain, repair, prepare for sale, advertise for sale, and sell the Collateral. The Administrative Agent is hereby granted a non-exclusive, royalty-free, irrevocable (until termination of this Agreement) license or other right to use, without charge, Issuer's labels, Intellectual Property, and advertising matter, or any similar property as it pertains to the Collateral, in completing production of, advertising for sale, and selling any Collateral or, otherwise in connection with the Administrative Agent's exercise of its rights under this <u>Section 9.1</u>, *provided* that the Issuer's rights under all licenses and all franchise agreements inure to the Administrative Agent's benefit;

(g) deliver a notice of exclusive control, any entitlement order, or other directions or instructions pursuant to any Control Agreement or similar agreements providing control of any Collateral;

(h) demand and receive possession of the Issuer's Books; and

(i) exercise all rights and remedies available to the Administrative Agent under the Note Documents or at Law or equity, including all remedies provided under the Code (including disposal of the Collateral pursuant to the terms thereof).

## 9.2 Repayment Value.

(a) The parties hereto further acknowledge and agree that the Repayment Value is not intended to act as a penalty or to punish the Issuer for any such redemption, prepayment, repayment or other event or occurrence.

(b) Notwithstanding anything to the contrary in this Agreement or any other Note Document, it is understood and agreed that if the Obligations are accelerated as a result of the occurrence and continuance of any Event of Default (including as a result of the commencement of any insolvency proceeding or by operation of Law or otherwise), the Repayment Value determined as of the date of such acceleration will also be due and payable (without duplication of any other Obligations) and will be treated and deemed as though the Notes were redeemed as of such date and shall constitute part of the Obligations for all purposes herein.

(c) The Repayment Value shall be presumed to be equal to the liquidated damages sustained by the Noteholders as the result of the occurrence of such acceleration, and the Issuer agrees that it is reasonable under the circumstances currently existing. The Repayment Value shall also be payable in the event the Obligations are satisfied or released by foreclosure (whether by power of judicial proceeding), deed in lieu of foreclosure or by any other means. THE ISSUER EXPRESSLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, THE PROVISIONS OF ANY PRESENT OR FUTURE STATUTE OR LAW THAT PROHIBITS OR MAY PROHIBIT THE ACCRUAL OR COLLECTION OF THE FOREGOING REPAYMENT VALUE IN CONNECTION WITH ANY SUCH ACCELERATION. The Issuer expressly agrees that (i) the Repayment Value is reasonable and is the product of an arm's length transaction between sophisticated business people, ably represented by counsel, (ii) the Repayment Value shall be payable notwithstanding the then prevailing market rates at the time payment is made, (iii) there has been a course of conduct between the Noteholders and the Issuer giving specific consideration in this transaction for such agreement to pay the Repayment Value is a material inducement to the Noteholders to purchase the Notes, and (vi) the Repayment Value represents a good faith, reasonable estimate and calculation of the lost profits or damages of the Noteholders as a result of any acceleration.

**9.3 Power of Attorney**. The Issuer hereby irrevocably appoints (without obligation) the Administrative Agent as its lawful attorney-in-fact, exercisable upon the occurrence and during the continuance of an Event of Default, to: (a) endorse the Issuer's name on any checks or other forms of payment or security; (b) sign the Issuer's name on any invoice or bill of lading for any Account or drafts against Account Debtors; (c) settle and adjust disputes and claims about the Accounts directly with Account Debtors, for amounts and on terms the Administrative Agent (acting at the direction of the Required Noteholders) determines reasonable; (d) make, settle, and adjust all claims under the Issuer's insurance policies; (e) pay, contest or settle any Lien, charge, encumbrance, security interest, and adverse claim in or to the Collateral, or any judgment based thereon, or otherwise take any action to terminate or discharge the same; and (f) transfer the Collateral into the name of the Administrative Agent or a third party as the Code permits. The Issuer hereby appoints (without obligation) the Administrative Agent's security interest in the Collateral regardless of whether an Event of Default has occurred until all Obligations (other than inchoate indemnity obligations) have been satisfied in full in cash and the Commitments have been terminated. The Administrative Agent's foregoing appointment as the Issuer's attorney in fact, and all of the Administrative Agent's rights and powers are coupled with an interest and are irrevocable until all Obligations (other than inchoate indemnity obligations) have been fully repaid in cash and all Commitments have been terminates have been terminated. The powers conferred on the Administrative Agent or any Secured Party to exercise any such powers.

**9.4 Protective Payments**. If the Issuer fails to obtain the insurance called for by Section 6.5 or fails to pay any premium thereon or fails to pay any other amount which the Issuer is obligated to pay under this Agreement or any other Note Document or which may be required to preserve the Collateral, the Administrative Agent (acting at the direction of the Required Noteholders) may obtain such insurance or make such payment, and all amounts so paid shall be immediately due and payable in cash, bearing interest at the then highest rate applicable to the Obligations and secured by the Collateral. The Administrative Agent (acting at the direction of the Required Noteholders) will make reasonable efforts to provide the Issuer with notice of the Administrative Agent obtaining such insurance. No payments by the Administrative Agent are deemed an agreement to make similar payments in the future or the Administrative Agent's or any other Secured Party's waiver of any Event of Default.

**9.5 Application of Payments and Proceeds Upon Default**. If an Event of Default has occurred and is continuing, the Administrative Agent shall have the right (upon the direction of the Required Noteholders), after payment of any fees, costs and expenses of the Administrative Agent under this Agreement and any other Note Document, to apply in any order any funds in its possession, whether from the Issuer's account balances, payments, proceeds realized as the result of any collection of Accounts or other disposition of the Collateral, or otherwise, to the Obligations. The Administrative Agent shall pay any surplus to the Issuer by credit to a deposit account designated in writing by the Issuer to the Administrative Agent; *provided*, *however*, that the Issuer shall remain liable to the Administrative Agent for any deficiency. If the Administrative Agent (acting at the direction of the Required Noteholders) shall have the option, exercisable at any time, of either reducing the Obligations by the principal amount of the purchase price or deferring the reduction of the Obligations (other than Obligations owing to the Administrative Agent) until the actual receipt by the Administrative Agent of cash therefor.

**9.6 Administrative Agent's Liability for Collateral**. So long as the Administrative Agent deals with the Collateral in its possession in the same manner as it deals with similar property held by it as a third party agent, the Administrative Agent shall not be liable or responsible for: (a) the safekeeping of the Collateral; (b) any loss or damage to the Collateral; (c) any diminution in the value of the Collateral; or (d) any act or default of any carrier, warehouseman, bailee, or other Person. The Issuer bears all risk of loss, damage or destruction of the Collateral.

**9.7 No Waiver; Remedies Cumulative**. The Administrative Agent's or any other Secured Party's failure, at any time or times, to require strict performance by the Issuer of any provision of this Agreement or any other Note Document shall not waive, affect, or diminish any right of the Administrative Agent or any other Secured Party thereafter to demand strict performance and compliance herewith or therewith. No waiver hereunder shall be effective unless signed by the Administrative Agent (at the direction of the Required Noteholders) and then is only effective for the specific instance and purpose for which it is given. The Administrative Agent and each Secured Party has all rights and remedies under this Agreement and the other Note Documents are cumulative. The Administrative Agent and each Secured Party has all rights and remedies provided under the Code, by Law, or in equity. The Administrative Agent's or any other Secured Party's exercise of one right or remedy is not an election and shall not preclude the Administrative Agent from exercising any other remedy under this Agreement or other remedy available at Law or in equity, and the Administrative Agent's or any other Secured Party's waiver of any Event of Default is not a continuing waiver. The Administrative Agent's or any other Secured Party's delay in exercising any remedy is not a waiver, election, or acquiescence.

**9.8 Demand Waiver**. Except as otherwise set forth in the Note Documents, the Issuer waives demand, notice of default or dishonor, notice of payment and nonpayment, notice of any default, nonpayment at maturity, release, compromise, settlement, extension, or renewal of accounts, documents, instruments, chattel paper, and guarantees held by the Administrative Agent on which the Issuer is liable.

**9.9 Issuer Liability**. Each Issuer hereby appoints each other Issuer as agent for such appointing Issuer for all purposes hereunder. Each Issuer hereunder shall be jointly and severally obligated to repay all Obligations, regardless of which Issuer actually receives the proceeds of any Notes, as if each Issuer directly received all such proceeds. Each Issuer waives (a) any suretyship defenses available to such Issuer under the Code or any other applicable Law, including the benefit of California Civil Code Section 2815 permitting revocation as to future transactions and the benefit of California Civil Code Sections 1432,

2809, 2810, 2819, 2839, 2845, 2847, 2848, 2849, 2850, and 2899 and 3433, and (b) any right to require the Administrative Agent to: (i) proceed against any Issuer or any other Person; (ii) proceed against or exhaust any security; or (iii) pursue any other remedy. The Administrative Agent may exercise or not exercise any right or remedy it has against any Issuer or any security it holds (including the right to foreclose by judicial or non-judicial sale) without affecting any Issuer's liability. Notwithstanding any other provision of this Agreement or other related document, each Issuer irrevocably waives all rights that it may have at Law or in equity (including any Law subrogating the Issuer to the rights of the Administrative Agent under this Agreement) to seek contribution, indemnification or any other form of reimbursement from any other Issuer or any other Person now or hereafter primarily or secondarily liable for any of the Obligations, for any payment made by the such Issuer with respect to the Obligations in connection with this Agreement or otherwise and all rights that it might have to benefit from, or to participate in, any security for the Obligations as a result of any payment made by such Issuer with respect to the Obligations in connection with this Agreement or otherwise. Any agreement providing for indemnification, reimbursement or any other arrangement prohibited under this Section shall be null and void. If any payment is made to an Issuer in contravention of this Section, such Issuer shall hold such payment in trust for the Administrative Agent and, at the election of the Noteholders, such payment shall be promptly delivered to the Administrative Agent for application to the Obligations, whether matured or unmatured.

## 9.10 Pledged Equity; Voting.

(a) The Administrative Agent (acting at the direction of the Required Noteholders) shall have the right, at any time upon the occurrence and during the continuance of any Event of Default, without notice to the Issuer, to endorse, assign or otherwise transfer to or to register in the name of the Administrative Agent or any of its nominees or endorse for negotiation any or all of the equity interests constituting Collateral, without any indication that such equity interests are subject to the security interest hereunder, and the Administrative Agent (acting at the direction of the Required Noteholders) shall have the right to exchange certificates representing or evidencing such Equity Interests for certificates of smaller or larger denominations.

(b) The Administrative Agent and the Noteholders agree that, unless an Event of Default shall have occurred and be continuing, the Issuer may, to the extent the Issuer has such right as a holder of the Collateral consisting of equity interests or other securities or Indebtedness owed by any obligor, vote and give consents, ratifications and waivers with respect thereto, except to the extent that any such vote, consent, ratification or waiver would diminish the value thereof as Collateral or which would result in any violation of any provision of this Agreement or the other Note Documents.

(c) The Administrative Agent and the Noteholders agree that the Issuer may, unless an Event of Default shall have occurred and be continuing, receive and retain all distributions with respect to the Collateral consisting of equity interests or other securities or indebtedness owed by any obligor; *provided, however*, that any and all such distributions consisting of securities that would otherwise constitute Collateral shall be received in trust for the benefit of the Administrative Agent and be promptly (but in any event within five (5) Business Days after receipt thereof) delivered to the Administrative Agent as Collateral in the same form as so received (with any necessary endorsement).

(d) Upon the occurrence and during the continuance of any Event of Default:

(i) all rights of the Issuer to exercise the voting and other consensual rights it would otherwise be entitled to exercise pursuant to <u>Section 9.10(b)</u> shall immediately cease, and all such rights shall thereupon become vested in the Administrative Agent, which shall have the sole right to exercise such voting and other consensual rights; and

(ii) all rights of the Issuer to receive distributions which it would otherwise be authorized to receive and retain pursuant to <u>Section 9.10(c)</u> shall immediately cease and all such rights shall thereupon become vested in the Administrative Agent, which shall have the sole right to receive and hold such distributions as Collateral.

## 10. NOTICES

All notices, consents, requests, approvals, demands, or other communication by any party to this Agreement or any other Note Document must be in writing and shall be deemed to have been validly served, given, or delivered: (a) upon the earlier of actual receipt and three (3) Business Days after deposit in the U.S. mail, first class, registered or certified mail return receipt requested, with proper postage prepaid; (b) upon transmission, when sent by electronic mail; (c) one (1) Business Day after deposit with a reputable overnight courier with all charges prepaid; or (d) when delivered, if hand-delivered by messenger, all of which shall be addressed to the party to be notified and sent to the address or email address indicated below. The Administrative Agent, the Noteholders and the Issuer may change its mailing or email address by giving each other party written notice thereof in accordance with the terms of this <u>Section 10</u>.

If to Issuer:

Eargo, Inc. 2665 North First Street, Suite 300 San Jose, CA 95134 Email: [\*\*\*] Attention: Chief Legal Officer

with a copy (which shall not constitute notice) to:

Latham & Watkins LLP 10250 Constellation Blvd. Suite 1100 Los Angeles, CA 90067 E-mail: [\*\*\*] Attention: David Zaheer; Charles Ruck; Kathleen Wells

If to the Administrative Agent:

Drivetrain Agency Services, LLC 410 Park Avenue, Suite 900 New York, NY 10022 Attn: Tim Daileader Email: [\*\*\*]

with a copy (which shall not constitute notice) to:

Seward & Kissel LLP One Battery Park Plaza New York, NY 10004 E-mail: [\*\*\*] Attention: Gregg Bateman

If to the Noteholders:

Patient Square Equity Advisors, LP 2884 Sand Hill Road, Suite 100 Menlo Park, CA 94025 E-mail: [\*\*\*] Attention: Adam Fliss, Justin Sabet-Peyman

with a copy (which shall not constitute notice) to:

Ropes & Gray LLP Prudential Tower 800 Boylston Street Boston, MA 02199 E-mail: [\*\*\*] Attention: Tom Fraser

Ropes & Gray LLP Three Embarcadero Center San Francisco, CA 94111 Email: [\*\*\*] Attention: Jason Freedman

Ropes & Gray LLP 1211 Avenue of the Americas New York, NY 10036-8704 Email: [\*\*\*] Attention: Alex Zeltser

## 11. CHOICE OF LAW, VENUE, JURY TRIAL WAIVER AND JUDICIAL REFERENCE

(a) This Agreement shall be governed by, and construed in accordance with, the Laws of the state of New York, without giving effect to any choice of Law or conflict of Law rules or provisions (whether of the state of New York) or any other jurisdiction) that would cause the applicable of the Laws of any other jurisdiction other than the state of New York. Each party hereto submits to the exclusive jurisdiction of the State and Federal courts in New York County, New York; *provided, however*, that nothing in this Agreement shall be deemed to operate to preclude any Secured Party from bringing suit or taking other legal action in any other jurisdiction to realize on the Collateral or any other security for the Obligations, or to enforce a judgment or other court order in favor of such Secured Party. The Issuer expressly submits and consents in advance to such jurisdiction, improper venue, or forum non conveniens and hereby consents to the granting of such legal or equitable relief as is deemed appropriate by such court. The Issuer hereby waives personal service of the summons, complaints, and other process issued in such action or suit and agrees that service of such summons, complaints, and other process issued to Issuer at the address set forth in, or subsequently provided by the Issuer in accordance with, <u>Section 10</u> of this Agreement and that service so made shall be deemed completed upon the earlier to occur of the Issuer's actual receipt thereof or three (3) days after deposit in the U.S. mails, proper postage prepaid.

## (b) TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY HERETO WAIVES ITS RIGHT TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION ARISING OUT OF OR BASED UPON THIS AGREEMENT, THE NOTE DOCUMENTS OR ANY CONTEMPLATED TRANSACTION, INCLUDING CONTRACT, TORT, BREACH OF DUTY AND ALL OTHER CLAIMS. THIS WAIVER IS A MATERIAL INDUCEMENT FOR THE PARTIES TO ENTER INTO THIS AGREEMENT. EACH PARTY HAS REVIEWED THIS WAIVER WITH ITS COUNSEL.

(c) This <u>Section 11</u> shall survive the termination of this Agreement.

#### 12. GENERAL PROVISIONS

## 12.1 Successors and Assigns.

(a) <u>Generally</u>. This Agreement binds and is for the benefit of the successors and permitted assigns of each party. The Issuer may not assign this Agreement or any other Note Document or any rights or obligations under such Note Document without the Noteholders' prior written consent (which may be granted or withheld in the Noteholders' discretion). Each Noteholder has the right, without the consent of or notice to Issuer, to sell, transfer, assign or grant participations in all or any part of, or any interest in, such Noteholder's obligations, rights, and benefits under this Agreement and the other Note Documents (including all or any portion of the Notes), pursuant to an assignment agreement in form and substance reasonably acceptable to the Required Noteholders; *provided*, that, unless an Event of Default has occurred and is continuing, no Noteholder shall consummate any such sale, transfer, assignment or grant prior to the third (3<sup>rd</sup>) anniversary of the Closing Date, except as permitted under the Investor Rights Agreement; *provided, further*, with respect to any participation by a Noteholder in all or any part of, or any interest in, such Noteholder's obligations, rights, and benefits, (A) such Noteholder's obligations under this Agreement shall remain unchanged, (B) such Noteholder shall remain solely responsible to the other parties hereto for the performance of such obligations, and (C) the Issuer, the Administrative Agent and the other Noteholders shall continue to deal solely and directly with such Noteholder in connection with such Noteholder's rights and obligations under this Agreement.

(b) Each Noteholder that sells a participation shall, acting solely for this purpose as a non-fiduciary agent of the Issuer, maintain a register complying with the requirements of Sections 163(f), 871(h) and 881(c)(2) of the IRC and the Treasury regulations issued thereunder on which is entered the name and address of each participant described in <u>Section 12.1(a)</u> above and the principal amounts (and stated interest) of each participant's interest in the Notes or other obligations under the Note Documents (the "**Participant Register**"); <u>provided</u> that no Issuer shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any participant or any information relating to a participant's interest in any commitments, loans, letters of credit or its other obligations under any Note Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan, letter of credit or other obligations, or is otherwise required thereunder; *provided*, *further*, (A) such Noteholder's obligations under this Agreement shall remain unchanged, (B) such Noteholder shall remain solely responsible to the other parties hereto for the performance of such obligations, and (C) the Issuer, the Administrative Agent and the other Noteholder shall continue to deal solely and directly with such Noteholder in connection with such Noteholder shall treat each Person whose name is recorded in the Participant Register shall be conclusive absent manifest error, and such Noteholder shall treat each Person whose name is recorded in the Participant Register shall be conclusive Agent) shall have no responsibility with respect to any participation by a Noteholder, including, without limitation, for maintaining a Participant Register.

## 12.2 Indemnification.

12.2.1 IN ADDITION TO THE PAYMENT OF EXPENSES PURSUANT TO SECTION 12.9, WHETHER OR NOT THE TRANSACTIONS CONTEMPLATED HEREBY SHALL BE CONSUMMATED, EACH ISSUER AGREES TO DEFEND, INDEMNIFY AND HOLD HARMLESS, ADMINISTRATIVE AGENT AND EACH NOTEHOLDER. THEIR AFFILIATES AND THEIR RESPECTIVE OFFICERS, PARTNERS, MEMBERS, INVESTORS, ADVISORS, REPRESENTATIVES, DIRECTORS, TRUSTEES, EMPLOYEES AND AGENTS (EACH, AN "INDEMNITEE"), FROM AND AGAINST ANY AND ALL INDEMNIFIED LIABILITIES; PROVIDED, NO ISSUER SHALL HAVE ANY OBLIGATION TO ANY INDEMNITEE HEREUNDER WITH RESPECT TO ANY INDEMNIFIED LIABILITIES TO THE EXTENT SUCH INDEMNIFIED LIABILITIES ARISE FROM THE GROSS NEGLIGENCE, BAD FAITH OR WILLFUL MISCONDUCT, AS DETERMINED BY A COURT OF COMPETENT JURISDICTION IN A FINAL, NON-APPEALABLE ORDER, OF THAT INDEMNITEE. TO THE EXTENT THAT THE UNDERTAKINGS TO DEFEND, INDEMNIFY AND HOLD HARMLESS SET FORTH IN THIS SECTION 12.2.1 MAY BE UNENFORCEABLE IN WHOLE OR IN PART BECAUSE THEY ARE VIOLATIVE OF ANY LAW OR PUBLIC POLICY, THE ISSUER SHALL CONTRIBUTE THE MAXIMUM PORTION THAT IT IS PERMITTED TO PAY AND SATISFY UNDER APPLICABLE LAW, WITHOUT ANY FURTHER LIABILITY, TO THE PAYMENT AND SATISFACTION OF ALL INDEMNIFIED LIABILITIES INCURRED BY INDEMNITEES OR ANY OF THEM. THIS SECTION 12.2.1 SHALL NOT APPLY WITH RESPECT TO ANY TAXES, OTHER THAN ANY TAXES THAT REPRESENT LOSSES, CLAIMS, DAMAGES, ETC. ARISING FROM ANY NON-TAX CLAIM, WHICH SHALL BE EXCLUSIVELY GOVERNED BY SECTION 2.7.3. THE AGREEMENTS IN THIS SECTION 12.2.1 SHALL SURVIVE REPAYMENT OF THE OBLIGATIONS AND ALL OTHER AMOUNTS PAYABLE UNDER THE NOTE DOCUMENTS, THE TERMINATION OF THIS AGREEMENT AND THE RESIGNATION OR REMOVAL OF THE ADMINISTRATIVE AGENT.

**12.2.2** To the extent permitted by applicable Law, no party hereto shall assert, and each other party hereto hereby waives, any claim against any other party and such party's respective Affiliates, directors, employees, attorneys or agents, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) (whether or not the claim therefor is based on contract, tort or duty imposed by any applicable legal requirement) arising out of, in connection with, as a result of, or in any way related to, this Agreement or any Note Document or any agreement or instrument contemplated hereby or thereby or referred to herein or therein, the transactions contemplated hereby or thereby, any Note or the use of the proceeds thereof or any act or omission or event occurring in connection therewith, and each party hereto hereby waives, releases and agrees not to sue upon any such claim (for the sake of clarification amounts in respect of any damages incurred or paid by an Indemnitee to a third party or for any out-of-pocket expenses shall be deemed to be direct, actual damages) or any such damages, whether or not accrued and whether or not known or suspected to exist in its favor.

12.3 Time of Essence. Time is of the essence for the performance of all Obligations in this Agreement.

12.4 Severability of Provisions. Each provision of this Agreement is severable from every other provision in determining the enforceability of any provision.

**12.5 Correction of Note Documents**. The Administrative Agent (acting at the direction of the Required Noteholders) may correct patent errors and fill in any blanks in the Note Documents consistent with the agreement of the parties so long as the Administrative Agent provides Issuer with written notice of such correction and allows Issuer at least five (5) days to object to such correction. In the event of such objection, such correction shall not be made except by an amendment signed by both the Administrative Agent (acting at the direction of the Required Noteholders) and Issuer. The Administrative Agent shall have no obligation to enter into any such amendment or make any such correction that affects its rights, obligations, protections, immunities or indemnities.

**12.6 Amendments in Writing; Waiver; Integration**. No purported amendment or modification of any Note Document, or waiver, discharge or termination of any obligation under any Note Document, shall be enforceable or admissible unless, and only to the extent, expressly set forth in a writing signed by the Issuer and the Required Noteholders; *provided* that no such agreement shall amend, modify or otherwise affect the rights or duties of the Administrative Agent hereunder or under any other Note Document without the prior written consent of the Administrative Agent. Without limiting the generality of the foregoing, no oral promise or statement, nor any action, inaction, delay, failure to require performance or course of conduct shall operate as, or evidence, an amendment, supplement or waiver or have any other effect on any Note Document. Any waiver granted shall be limited to the specific circumstance expressly described in it and shall not apply to any subsequent or other circumstance, whether similar or dissimilar, or give rise to, or evidence, any obligation or commitment to grant any further waiver. The Note Documents represent the entire agreement about this subject matter and supersede prior negotiations or agreements. All prior agreements, understandings, representations, warranties, and negotiations between the parties about the subject matter of the Note Documents merge into the Note Documents.

12.7 Counterparts. This Agreement may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, is an original, and all taken together, constitute one Agreement. Delivery of an executed counterpart of a signature page of this Agreement by email or other electronic means (including a ".pdf" or ".tif" file) shall be effective as delivery of a manually executed counterpart of this Agreement.

**12.8** Confidentiality. In handling any confidential information, each Secured Party shall exercise the same degree of care that it exercises for its own proprietary information, but disclosure of information may be made: (a) to such Secured Party's Subsidiaries or Affiliates, directors (or equivalent managers), officers, employees, independent auditors, or other agents, experts and advisors, including accountants, legal counsel and other advisors; (b) to prospective transferees or purchasers of any interest in the Notes (*provided, however*, that such Secured Party shall use its reasonable best efforts to obtain any prospective transferee's or purchaser's agreement to the terms of this provision); (c) as required by Law, regulation, subpoena, or other order; (d) to such Secured Party's regulators or as otherwise required in connection with such Secured Party's examinations or audits; (e) as such Secured Party considers appropriate in exercising remedies under the Note Documents or performing its obligations under the Note Documents; and (f) to third-party service providers of such Secured Party so long as such service providers have executed a confidentiality agreement with such Secured Party with terms no less restrictive than those contained herein. Confidential information does not include information that is either: (i) in the public domain or in any Secured Party is possession when disclosed to such Secured Party, or becomes part of the public domain (other than as a result of its disclosure by such Secured Party does not know that the third party is prohibited from disclosing the information. The Secured Parties and their Affiliates may use anonymous forms of confidential information for aggregate datasets, for analyses or reporting, and for any other uses not expressly prohibited in writing by Issuer. The provisions of the immediately preceding sentence shall survive termination of this Agreement.

**12.9 Expenses.** The Issuer agrees to pay the Administrative Agent such fees as set forth in the Agent Fee Letter. The Issuer agrees to pay promptly (and, in any event, within ten (10) Business Days of the Administrative Agent's or any Noteholder's request; *provided*, that any costs and expenses incurred prior to the Closing Date, the First Tranche Closing Date or the Second Tranche Closing Date shall be paid on the Closing Date, the First Tranche Closing Date to the extent invoiced on or prior to such date, as applicable) (a) all of Administrative Agent's and the

Noteholders' reasonable and documented out-of-pocket costs and expenses of preparation of the Note Documents and any consents, amendments, waivers or other modifications thereto; (b) all the reasonable and documented out-of-pocket fees, expenses and disbursements of one primary external counsel to Administrative Agent (and, if necessary, any regulatory or subject matter expert counsel and one additional counsel in each relevant material jurisdiction) and one primary external counsel to the Noteholders (and, if necessary, any regulatory or subject matter expert counsel and one additional counsel in each relevant material jurisdiction) in connection with the negotiation, preparation, execution and administration of the Note Documents and any consents, amendments, waivers or other modifications thereto; (c) all of the actual costs and documented out-of-pocket expenses of creating and perfecting Liens in favor of Administrative Agent, for the benefit of Secured Parties, including filing and recording fees, expenses and taxes, stamp or documentary taxes, search fees, title insurance premiums and the reasonable and documented out-of-pocket fees, expenses and disbursements of one counsel in each relevant material jurisdiction providing customary legal opinions that Administrative Agent or Required Noteholders may request in respect of the Collateral or the Liens created pursuant to the Collateral Documents; (d) all of Administrative Agent's and the Noteholders' reasonable and documented out-of-pocket fees and expenses for, and disbursements of any of Administrative Agent's auditors, accountants, consultants or appraisers; (e) all the actual, reasonable and documented out-of-pocket costs and expenses (including the reasonable fees, expenses and disbursements of any appraisers, consultants, advisors and agents employed or retained by Administrative Agent, the Noteholders and their respective counsel) in connection with the custody or preservation of any of the Collateral; (f) all the reasonable and documented out-of-pocket costs and expenses of Administrative Agent and Noteholders in connection with the attendance at any meetings in connection with this Agreement and the other Note Documents; (g) all other and reasonable and documented out-of-pocket costs and expenses incurred by Administrative Agent and the Noteholders in connection with the negotiation, preparation and execution of the Note Documents and any consents, amendments, waivers or other modifications thereto and the transactions contemplated thereby; and (h) all actual and documented out-of-pocket costs and expenses, including attorneys' fees and costs of settlement, incurred by Administrative Agent and Noteholders in enforcing any Obligations of or in collecting any payments due from any Issuer hereunder or under the other Note Documents by reason of such Default or Event of Default (including in connection with the sale of, collection from, or other realization upon any of the Collateral) or in connection with any refinancing or restructuring of the credit arrangements provided hereunder in the nature of a "work out" or pursuant to any insolvency or bankruptcy cases or proceedings or otherwise. The agreements in this Section 12.9 shall survive repayment of the Obligations and all other amounts payable under the Note Documents, the termination of this Agreement and the resignation or removal of the Administrative Agent.

**12.10 Electronic Execution of Documents**. The words "execution," "signed," "signature" and words of like import in any Note Document shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity and enforceability as a manually executed signature or the use of a paper-based recordkeeping systems, as the case may be, to the extent and as provided for in any applicable Law, including any state Law based on the Uniform Electronic Transactions Act.

12.11 Captions. The headings used in this Agreement are for convenience only and shall not affect the interpretation of this Agreement.

**12.12 Construction of Agreement**. The parties mutually acknowledge that they and their attorneys have participated in the preparation and negotiation of this Agreement. In cases of uncertainty this Agreement shall be construed without regard to which of the parties caused the uncertainty to exist.

**12.13 Relationship**. The relationship of the parties to this Agreement is determined solely by the provisions of this Agreement. The parties do not intend to create any agency, partnership, joint venture, trust, fiduciary or other relationship with duties or incidents different from those of parties to an arm's-length contract.

12.14 Survival. All representations and warranties made hereunder and in any other Note Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by the Administrative Agent and each Noteholder and the Issuer, regardless of any investigation made by the Administrative Agent or any Noteholder or the Issuer or on their behalf and notwithstanding that the Administrative Agent or any Noteholder or the Issuer may have had notice or knowledge of any Default or Event of Default at the time of any issuance of Notes, and shall continue in full force and effect as long as any Note or any other Obligation hereunder shall remain unpaid or unsatisfied (other than contingent indemnification obligations as to which no claim has been asserted).

**12.15 Third Parties**. Nothing in this Agreement, whether express or implied, is intended to: (a) confer any benefits, rights or remedies under or by reason of this Agreement on any persons other than the express parties to it and their respective permitted successors and assigns; (b) relieve or discharge the obligation or liability of any person not an express party to this Agreement; or (c) give any person not an express party to this Agreement any right of subrogation or action against any party to this Agreement.

**12.16 Set-Off**. In addition to any rights now or hereafter granted under applicable Law and not by way of limitation of any such rights, upon the occurrence of any Event of Default, each Noteholder, and their respective Affiliates is hereby authorized by each Issuer at any time or from time to time, without notice to any Issuer or to any other Person (other than Administrative Agent), any such notice being hereby expressly waived, to set off and to appropriate and to apply any and all deposits (general or special, including Indebtedness evidenced by certificates of deposit, whether matured or unmatured, but not including trust accounts (in whatever currency)) and any other Indebtedness at any time held or owing by such Noteholder to or for the credit or the account of any Issuer (in whatever currency) against and on account of the obligations and liabilities of any Issuer to such Noteholder hereunder, the participations under the other Note Documents, including all claims of any nature or description arising out of or connected hereto, or with any other Note Document, irrespective of whether or not (a) such Noteholder shall have made any demand hereunder, (b) the principal of or the interest in respect of the Notes or any other amounts due hereunder shall have become due and payable and although such obligations and liabilities, or any of them, may be contingent or unmatured or (c) such obligation or liability is owed to a branch or office of such Noteholder different from the branch or office holding such deposit or obligation or such Indebtedness. Each Noteholder agrees to notify the Issuer and the Administrative Agent promptly of its exercise of any rights under this <u>Section 12.16</u>, but the failure to provide such notice shall not otherwise limit its rights under this <u>Section 12.16</u> or result in any liability to such Noteholder.

12.17 Non-Recourse Parties. Without limiting any of the provisions contained in the equity commitment letter executed by Patient Square Equity Partners, LP on the date hereof, this Agreement may only be enforced against, and any claim or cause of action based upon, arising out of, or related to this Agreement, or the transactions contemplated hereby may only be brought against the entities that are expressly named as parties hereto and their respective successors and assigns. Except as set forth in the immediately preceding sentence, no past, present or future director, officer, employee, incorporator, member, partners, stockholder, Affiliate, agent, attorney, advisor or representative of any party hereto (collectively, the "Non-Recourse Parties") shall have any liability for any obligations or liabilities of any party hereto under this Agreement or for any claim based on, in respect of, or by reason of, the transactions contemplated hereby (whether in contract, in tort, in Law or in equity or provided by statute, whether by or through attempted piercing of the corporate, limited partnership or limited liability company veil or otherwise), except to the extent expressly set forth in this Agreement. For the avoidance of doubt, this Section 12.17 is intended to benefit and may be enforced by each Non-Recourse Party (and each such Person shall be a third-party beneficiary of this Section 12.17) and shall be binding on all respective successors and permitted assigns thereof.

**12.18 Termination.** This Agreement and any Note Documents shall terminate and be void and of no further force and effect, and all rights and obligations of the parties hereunder and thereunder shall terminate without any further liability on the part of any party in respect thereof (a) upon the mutual written agreement of each of the parties hereto to terminate this Agreement, and (b) if the First Tranche Closing does not occur on or prior to July 11, 2022; *provided, however*, that nothing herein will relieve any party from liability for any willful breach hereof prior to the time of termination, and each party will be entitled to any remedies at Law or in equity to recover losses, liabilities or damages arising from any such willful breach.

## 13. ADMINISTRATIVE AGENT.

## 13.1 Appointment of Administrative Agent.

(a) DriveTrain is hereby appointed as the Administrative Agent hereunder and under the other Note Documents and each Noteholder hereby authorizes DriveTrain, in such capacity, to act as its agent in accordance with the express terms hereof and the other Note Documents to perform, exercise and enforce any and all other rights and remedies of the Noteholders with respect to the Issuer, the Obligations or otherwise related to any of same to the extent reasonably incidental to the exercise by the Administrative Agent of the rights and remedies specifically authorized to be exercised by the Administrative Agent by the terms of this Agreement or any other Issuer. Concurrently herewith, each Noteholder directs the Administrative Agent, and the Administrative Agent is authorized, to enter into this Agreement and the other Note Documents and any other related agreements in the forms presented to the Administrative Agent. For the avoidance of doubt, each Noteholder agrees that it will be subject to and bound by the terms of this Agreement and the other Note Documents.

(b) The Administrative Agent hereby agrees to act upon the express conditions contained herein and the other Note Documents, as applicable. The provisions of this <u>Section 13</u> are solely for the benefit of the Administrative Agent and Noteholders and no Issuer shall have any rights as a third-party beneficiary of any of the provisions thereof. In performing its functions and duties hereunder, the Administrative Agent shall act solely as an agent of Noteholders and does not assume and shall not be deemed to have assumed any obligation towards or relationship of agency or trust with or for the Issuer or any of its Subsidiaries.

(c) The Administrative Agent agrees that, upon receipt of any notice, report, financial statement, certificate or other written communication from the Issuer (other than such communications that that has been contemporaneously provided by the Issuer or such Subsidiary to the Noteholders or that has been deemed to have been delivered by posting on or provision of a link on Eargo's or other website), the Administrative Agent shall promptly (and, in any event, within one (1) Business Day) provide such notice, report, financial statement, certificate or other written communication to the Noteholders in accordance with <u>Section 10</u>.

## 13.2 Powers and Duties; Exculpatory Provisions.

(a) <u>Powers and Duties</u>. Each Noteholder irrevocably authorizes the Administrative Agent to take such action on such Noteholder's behalf and to exercise such powers, rights and remedies hereunder and under the other Note Documents as are specifically delegated or granted to the Administrative Agent by the terms hereof and thereof, together with such powers, rights and remedies as are reasonably incidental thereto. The Administrative Agent shall have only those duties and responsibilities that are expressly specified herein and the other Note Documents. The Administrative Agent may exercise such powers, rights and remedies and perform such duties by or through its agents or employees. The Administrative Agent shall not have, by reason hereof or any of the other Note Documents, a fiduciary relationship in respect of any Noteholder. Nothing herein or in any of the other Note Documents, expressed or implied, is intended to or shall be so construed as to impose upon the Administrative Agent any obligations in respect hereof or any of the other Note Documents except as expressly set forth herein or therein. The Administrative Agent may assume that, the conditions set forth in each of Section 3.1, Section 3.2 and Section 3.3 were satisfied on the Closing Date, the First Tranche Closing Date or the Second Tranche Closing Date, as applicable, unless notified otherwise in writing by a Noteholder prior to such date.

(b) Exculpatory Provisions. Neither Administrative Agent nor any of its officers, partners, directors, employees or agents shall be liable to Noteholders or any other Person for any action taken or omitted by the Administrative Agent under or in connection with any of the Note Documents except to the extent caused by the Administrative Agent's gross negligence, bad faith or willful misconduct, as determined by a court of competent jurisdiction in a final, non-appealable order; provided that, no action taken or not taken by the Administrative Agent pursuant to and in accordance with the direction of the Required Noteholders shall be deemed to constitute gross negligence or willful misconduct of the Administrative Agent. The Administrative Agent shall be entitled to refrain from any act or the taking of any action (including the failure to take an action) in connection herewith or any of the other Note Documents or from the exercise of any power, discretion or authority vested in it hereunder or thereunder unless and until the Administrative Agent shall have received instructions in respect thereof from Required Noteholders and, upon receipt of such instructions from Required Noteholders, Administrative Agent shall be entitled to act or (where so instructed) refrain from acting, or to exercise such power, discretion or authority, in accordance with such instructions, and shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement and the other Note Documents in accordance with a request of the Required Noteholders; provided that the Administrative Agent shall not be required to take any action at the direction of the Required Noteholders unless it is furnished with an indemnification reasonably satisfactory to the Administrative Agent with respect thereto or that is contrary to this Agreement, any other Note Document or applicable Law. For purposes of clarity, but without limiting any rights, protections, immunities or indemnities afforded to the Administrative Agent hereunder or under any other Note Document, phrases such as "satisfactory to the Administrative Agent," "approved by the Administrative Agent," "acceptable to the Administrative Agent," "as determined by the Administrative Agent," "in the Administrative Agent's discretion," "selected by the Administrative Agent," "elected by the Administrative Agent," "requested by the Administrative Agent," and phrases of similar import that authorize or permit the Administrative Agent to approve, disapprove, determine, act or decline to act in its discretion shall be subject to the Administrative Agent receiving written direction from the Required Noteholders to take such action or to exercise such rights. Nothing contained in this Agreement or any other Note Documents shall require the Administrative Agent to exercise any discretionary acts.

(c) Without prejudice to the generality of the foregoing, (i) Administrative Agent shall be entitled to conclusively rely, and shall be fully protected in relying, upon any communication, instrument or document believed by it to be genuine and correct and to have been signed or sent by the proper Person or Persons, and shall be entitled to conclusively rely and shall be protected in relying on opinions and judgments of attorneys (who may be attorneys for the Issuer and its Subsidiaries), accountants, experts and other professional advisors selected by it; and (ii) no Person shall have any right of action whatsoever against Administrative Agent as a result of Administrative Agent acting or (where so instructed) refraining from acting hereunder or any of the other Note Documents in accordance with the instructions of Required Noteholders.

(d) Neither the Administrative Agent nor any of its officers, directors, employees, agents, attorneys-in-fact or Affiliates shall be responsible in any manner for maintenance of any insurance required under this Agreement or any other Note Document or for the creation, perfection or priority of any Lien or security interest created or purported to be created under this Agreement or any other Note Documents, or for any failure of any Issuer or any other Person to perform its obligations hereunder or thereunder.

(e) The Administrative Agent shall not be under any obligation to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or the satisfaction of any conditions of, this Agreement or any other Note Document, or to inspect the properties, books or records of any Issuer or any other Person.

(f) In no event shall the Administrative Agent be liable for any failure or delay in the performance of its obligations under this Agreement or any other Note Document because of circumstances beyond its control, including, but not limited to, a failure, termination, or suspension of a clearing house, securities depositary, settlement system or central payment system in any applicable part of the world or acts of God, flood, war (whether declared or undeclared), civil or military disturbances or hostilities, nuclear or natural catastrophes, political unrest, explosion, earthquake, terrorism, fire, riot, labor disturbances, strikes or work stoppages for any reason, embargo, epidemics or pandemics or other health crises, government action, including any laws, ordinances, regulations or the like (whether domestic, federal, state, county or municipal or foreign) which delay, restrict or prohibit the providing of the services contemplated by this Agreement or the other Note Documents, or the unavailability of communications or computer facilities, the failure of equipment or interruption of communications or computer facilities, or the unavailability of the Federal Reserve Bank wire or telex or other wire or communication facility, or any other causes beyond the Administrative Agent's control whether or not of the same class or kind as specified above.

(g) The rights, privileges, protections, immunities and benefits given to the Administrative Agent, including, without limitation, its right to be indemnified, in this Agreement are extended to, and shall be enforceable: (i) by the Administrative Agent in each Note Document and any other document related hereto or thereto to which it is a party and (ii) the entity serving as the Administrative Agent in each of its capacities hereunder and in each of its capacities under any Note Document whether or not specifically set forth therein and each agent, custodian and other Person employed to act hereunder and under any Note Document or related document, as the case may be.

(h) Notwithstanding anything contained in this Agreement or any other Note Document to the contrary, in no event shall the Administrative Agent be responsible or liable for confirming or monitoring any Issuer's or any Noteholder's compliance with state or federal securities laws, including without limitation, (i) whether any Noteholder is an "accredited investor" as that term is defined in Rule 501 of the Securities Act, (ii) whether the Issuer or any Noteholder has complied with any transfer restrictions required by state or federal securities laws, or (iii) whether any conditions of Rule 144 promulgated under the Securities Act have been complied with.

## 13.3 General Immunity.

(a) <u>No Responsibility for Certain Matters</u>. The Administrative Agent shall not be responsible to any Noteholder for the value, execution, effectiveness, genuineness, validity, enforceability, collectability or sufficiency hereof or of any other Note Document or for any representations, warranties, recitals or statements made herein or therein or made in any written or oral statements or in any financial or other statements, instruments, reports or certificates or any other documents furnished or made by the Administrative Agent to the Noteholders or by or on behalf of the Issuer to the Administrative Agent or any Noteholder in connection with the Note Documents and the transactions contemplated thereby or for the financial condition or business affairs of the Issuer or any other Person liable for the payment of any Obligations, nor shall the Administrative Agent be required to ascertain or inquire as to the performance or observance of any of the terms, conditions, provisions, covenants or agreements contained in any of the Note Documents or as to the use of the proceeds of the Notes or as to the existence or possible existence of any Event of Default or Default. Anything contained herein to the contrary notwithstanding, the Administrative Agent shall not have any liability arising from confirmations of the amount of outstanding Notes or the component amounts thereof.

(b) Notice of Default. The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default, except with respect to Defaults or Events of Default in the payment of principal and interest required to be paid to the Administrative Agent for the account of the Noteholders, unless (x) Administrative Agent shall have received written notice from a Noteholder or the Issuer referring to this Agreement, describing such Default or Event of Default and stating that such notice is a "notice of default" or (y) an officer of the Administrative Agent with direct responsibility for the administration of this Agreement has actual knowledge of such Default or Event of Default. The Administrative Agent will notify the Noteholders of its receipt of any such notice within one (1) Business Day of such receipt of such written notice. The Administrative Agent shall take such action with respect to any such Default or Event of Default as may be directed by the Required Noteholders (or such other Noteholders as may be required to give such instructions pursuant to the terms hereof); provided, however, that unless and until Administrative Agent has received any such direction, the Administrative Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable.

**13.4 Administrative Agent Entitled to Act as Noteholder**. The agency hereby created shall in no way impair or affect any of the rights and powers of, or impose any duties or obligations upon, the Administrative Agent in its individual capacity as a Noteholder hereunder, if applicable. With respect to its participation in the Notes, the Administrative Agent shall have the same rights and powers hereunder as any other Noteholder and may exercise the same as if it were not performing the duties and functions delegated to it hereunder, and the term "Noteholder" shall, unless the context clearly otherwise indicates, include Administrative Agent in its individual capacity. Administrative Agent and its Affiliates may accept deposits from, lend money to, own securities of, and generally engage in any kind of banking, trust, financial advisory or other business with the Issuer or any of its Affiliates as if it were not performing the duties specified herein, and may accept fees and other consideration from the Issuer for services in connection herewith and otherwise without having to account for the same to Noteholders.

13.5 Right to Indemnity. EACH NOTEHOLDER, IN PROPORTION TO ITS PRO RATA SHARE, SEVERALLY AGREES TO INDEMNIFY THE ADMINISTRATIVE AGENT, ITS AFFILIATES AND ITS RESPECTIVE OFFICERS, PARTNERS, DIRECTORS, TRUSTEES, EMPLOYEES AND AGENTS OF ADMINISTRATIVE AGENT (EACH, AN "INDEMNITEE AGENT PARTY"), TO THE EXTENT THAT SUCH INDEMNITEE AGENT PARTY SHALL NOT HAVE BEEN REIMBURSED BY THE ISSUER, FOR AND AGAINST ANY AND ALL LIABILITIES, OBLIGATIONS, LOSSES, DAMAGES, PENALTIES, ACTIONS, JUDGMENTS, SUITS, COSTS, EXPENSES (INCLUDING COUNSEL FEES AND DISBURSEMENTS) OR DISBURSEMENTS OF ANY KIND OR NATURE WHATSOEVER WHICH MAY BE IMPOSED ON, INCURRED BY OR ASSERTED AGAINST SUCH INDEMNITEE AGENT PARTY IN EXERCISING ITS POWERS, RIGHTS AND REMEDIES OR PERFORMING ITS DUTIES HEREUNDER OR UNDER THE OTHER NOTE DOCUMENTS OR OTHERWISE IN ITS CAPACITY AS SUCH INDEMNITEE AGENT PARTY IN ANY WAY RELATING TO OR ARISING OUT OF THIS AGREEMENT OR THE OTHER NOTE DOCUMENTS, IN ALL CASES, WHETHER OR NOT CAUSED BY OR ARISING, IN WHOLE OR IN PART, OUT OF THE COMPARATIVE, CONTRIBUTORY, OR SOLE NEGLIGENCE OF SUCH INDEMNITEE AGENT PARTY; <u>PROVIDED</u>, THAT NO NOTEHOLDER SHALL BE LIABLE FOR ANY PORTION OF SUCH LIABILITIES, OBLIGATIONS, LOSSES, DAMAGES, PENALTIES,

ACTIONS, JUDGMENTS, SUITS, COSTS, EXPENSES OR DISBURSEMENTS RESULTING FROM SUCH INDEMNITEE AGENT PARTY'S GROSS NEGLIGENCE, BAD FAITH OR WILLFUL MISCONDUCT, AS DETERMINED BY A COURT OF COMPETENT JURISDICTION IN A FINAL, NON-APPEALABLE ORDER. IF ANY INDEMNITY FURNISHED TO ANY INDEMNITEE AGENT PARTY FOR ANY PURPOSE SHALL, IN THE OPINION OF SUCH INDEMNITEE AGENT PARTY, BE INSUFFICIENT OR BECOME IMPAIRED, SUCH INDEMNITEE AGENT PARTY MAY CALL FOR ADDITIONAL INDEMNITY AND CEASE, OR NOT COMMENCE, TO DO THE ACTS INDEMNIFIED AGAINST UNTIL SUCH ADDITIONAL INDEMNITY IS FURNISHED; <u>PROVIDED</u>, IN NO EVENT SHALL THIS SENTENCE REQUIRE ANY NOTEHOLDER TO INDEMNIFY ANY INDEMNITEE AGENT PARTY AGAINST ANY LIABILITY, OBLIGATION, LOSS, DAMAGE, PENALTY, ACTION, JUDGMENT, SUIT, COST, EXPENSE OR DISBURSEMENT IN EXCESS OF SUCH NOTEHOLDER'S PRO RATA SHARE THEREOF; AND <u>PROVIDED FURTHER</u>, THIS SENTENCE SHALL NOT BE DEEMED TO REQUIRE ANY NOTEHOLDER TO INDEMNIFY ANY INDEMNITEE AGENT PARTY AGAINST ANY LIABILITY, OBLIGATION, LOSS, DAMAGE, PENALTY, ACTION, JUDGMENT, SUIT, COST, EXPENSE OR DISBURSEMENT IN EXCESS OF SUCH NOTEHOLDER TO INDEMNIFY ANY INDEMNITEE AGENT PARTY AGAINST ANY LIABILITY, OBLIGATION, LOSS, DAMAGE, PENALTY, ACTION, JUDGMENT, SUIT, COST, EXPENSE OR DISBURSEMENT DESCRIBED IN THE PROVISO IN THE IMMEDIATELY PRECEDING SENTENCE. THE AGREEMENTS IN THIS <u>SECTION 13.5</u> SHALL SURVIVE REPAYMENT OF THE OBLIGATIONS AND ALL OTHER AMOUNTS PAYABLE UNDER THE NOTE DOCUMENTS, THE TERMINATION OF THIS AGREEMENT AND THE RESIGNATION OR REMOVAL OF THE ADMINISTRATIVE AGENT.

## 13.6 Successor Administrative Agent.

(a) The Administrative Agent may resign at any time by giving sixty (60) days' (or such shorter period as shall be agreed by the Required Noteholders) prior written notice thereof to the Noteholders and the Issuer. Upon any such notice of resignation, the Required Noteholders shall have the right, upon five (5) Business Days' notice to the Issuer, to appoint a successor Administrative Agent subject (to the extent that such successor Administrative Agent is not a Noteholder or an Affiliate of an Noteholder and so long as no Event of Default has occurred and is continuing) to the written consent of the Issuer (not to be unreasonably withheld, conditioned or delayed). If no successor shall have been so appointed by the Required Noteholders and shall have accepted such appointment within thirty (30) days after the retiring Administrative Agent gives notice of its resignation, then the retiring Administrative Agent may, on behalf of the Noteholders appoint a successor Administrative Agent from among the Noteholders. Upon the acceptance of any appointment as the Administrative Agent hereunder by a successor Administrative Agent, that successor Administrative Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Administrative Agent, and the retiring Administrative Agent shall promptly (i) transfer to such successor Administrative Agent all sums, securities or equity interests and other items of Collateral in its possession, together with all records and other documents necessary or appropriate in connection with the performance of the duties of the successor Administrative Agent under the Note Documents and (ii) at the reasonable request for the Issuer or the Required Noteholders, execute and deliver to such successor Administrative Agent such amendments to financing statements (as prepared by the Issuer or the Required Noteholders), and take such other actions as reasonably requested by the Issuer or the Required Noteholders, as may be necessary or appropriate in connection with the assignment to such successor Administrative Agent of the security interests created under the Collateral Documents, whereupon such retiring Administrative Agent shall be discharged from its duties and obligations hereunder. After any retiring Administrative Agent's resignation hereunder as Administrative Agent, the provisions of this Agreement shall inure to its benefit as to any actions taken or omitted to be taken by it while it was the Administrative Agent hereunder.

(b) Notwithstanding anything herein to the contrary, the Administrative Agent may assign its rights and duties as the Administrative Agent, as applicable, hereunder to an Affiliate of the Administrative Agent without the prior written consent of, or prior written notice to, the Issuer or the Noteholders; <u>provided</u> that the Issuer and the Noteholders may deem and treat such assigning Administrative Agent as the Administrative Agent for all purposes hereof, unless and until such assigning Administrative Agent provides written notice to the Issuer and the Noteholders of such assignment. Upon such assignment such Affiliate shall succeed to and become vested with all rights, powers, privileges and duties as the Administrative Agent hereunder and under the other Note Documents. Notwithstanding anything to the contrary contained herein or in any related document, any corporation or entity into which the Administrative Agent may be merged or converted or with which it may be consolidated, or any corporation or entity resulting from any merger, conversion or consolidation to which the Administrative Agent shall be a party, or any corporation or entity succeeding to the business of the Administrative Agent shall be the successor of the Administrative Agent hereunder and under any related document to which the Administrative Agent is a party without the execution or filing of any paper with any Person or any further act on the part of any Person.

(c) The Administrative Agent may perform any and all of its duties and exercise its rights and powers under this Agreement or under any other Note Document by or through any one or more sub-agents appointed by the Administrative Agent. The Administrative Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through their respective Affiliates. The exculpatory, indemnification and other provisions of this Agreement shall apply to any of the Affiliates and sub-agents of the Administrative Agent and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as the Administrative Agent. All of the rights, benefits and privileges (including the exculpatory and indemnification provisions) of this Agreement shall apply to any such sub-agent and to the Affiliates of any such sub-agent, and shall apply to their respective activities as sub-agent and Affiliates were named herein. Notwithstanding anything herein to the contrary, with respect to each sub-agent appointed by the Administrative Agent, (i) such sub-agent shall be a third party beneficiary under this Agreement with respect to all such rights, benefits and privileges (including exculpatory and rights to indemnification) and shall have all of the rights, benefits and privileges of a third party beneficiary, without the consent or joinder of any other Person, against any or all of the Issuer and the Noteholders, (ii) such rights, benefits and privileges (including exculpatory rights and rights to indemnification) shall not be modified or amended without the consent of such sub-agent, and (iii) such sub-agent shall only have obligations to Administrative Agent and not to any Issuer, Noteholder or any other Person and no Issuer, Noteholder or any other Person shall have the rights, directly or indirectly, as a third party beneficiary or otherwise, against such sub-agent.

## **13.7** Collateral Documents.

(a) <u>Administrative Agent under Collateral Documents</u>. Each Noteholder hereby further authorizes the Administrative Agent, on behalf of and for the benefit of the Secured Parties, to be the agent for and representative of the Secured Parties with respect to the Collateral and the Collateral Documents. Without further written consent or authorization from the Noteholders (other than as specified herein), the Administrative Agent may execute any documents or instruments prepared by the Issuer (and in form and substance satisfactory to the Administrative Agent) necessary to release any Lien encumbering any item of Collateral that is the subject of a sale or other disposition of assets permitted hereby or to which the Required Noteholders have otherwise consented. Notwithstanding anything contained in this Agreement or any other Note Document to the contrary, in no event shall the Administrative Agent be obligated to execute or deliver any document evidencing any release, subordination or reconveyance without receipt of a certificate executed by an authorized officer of Eargo certifying that such release, subordination or re-conveyance have been complied with. Upon request by the Administrative Agent at any time, the Required Noteholders will confirm in writing the Administrative Agent's authority to release, subordinate or re-convey its interest in particular types or items of property pursuant to this <u>Section 13.7</u>.

(b) <u>Right to Realize on Collateral</u>. Anything contained in any of the Note Documents to the contrary notwithstanding, the Issuer, other than as set forth in <u>Section 12.16</u>, the Administrative Agent and each Noteholder hereby agree that (i) no Noteholder shall have any right individually to realize upon any of the Collateral, it being understood and agreed that all powers, rights and remedies hereunder may be exercised solely by the Administrative Agent, and (ii) in the event of a foreclosure by the Administrative Agent on any of the Collateral pursuant to a public or private sale or any sale of the Collateral in a case under the Bankruptcy Code, the Administrative Agent or any Noteholder may be the purchaser of any or all of such Collateral at any such sale, and the Administrative Agent, as agent for and representative of the Secured Parties (but not any Noteholder or Noteholders in its or their respective individual capacities unless the Required Noteholders, shall be entitled, for the purpose of bidding and making settlement or payment of the purchase price for all or any portion of the Collateral sold at any such public sale, to use and apply any or all of the Obligations (other than Obligations owing to the Administrative Agent) as a credit on account of the purchase price for any Collateral payable by the Administrative Agent at such sale

**13.8 Agency for Perfection.** The Administrative Agent and each Noteholder hereby appoints each other Noteholder as agent and bailee for the purpose of perfecting the security interests in and Liens upon the Collateral in assets which, in accordance with Article 9 of the UCC, can be perfected only by possession or control (or where the security interest of a secured party with possession or control has priority over the security interest of another secured party), and the Administrative Agent and each Noteholder hereby acknowledges that it holds possession of or otherwise controls any such Collateral for the benefit of the Secured Parties. Should any Noteholder obtain possession or control of any such Collateral, such Noteholder shall notify the Administrative Agent thereof, and, promptly upon Administrative Agent's request therefore, shall deliver such Collateral to the Administrative Agent or in accordance with the Administrative Agent's instructions. In addition, Administrative Agent shall also have the power and authority hereunder to appoint such other sub-agents as may be necessary or required under applicable state Law or otherwise to perform its duties and enforce its rights with respect to the Collateral and under the Note Documents. Each Issuer by its execution and delivery of this Agreement hereby consents to the foregoing.

13.9 Reports and Other Information; Confidentiality; Disclaimers. By becoming a party to this Agreement, each Noteholder:

(a) is deemed to have requested that the Administrative Agent furnish such Noteholder, promptly after it is received by the Administrative Agent from any Issuer, a copy of each audit, examination or report with respect to the Issuer or its Subsidiaries (each a "**Report**" and collectively, "**Reports**") prepared by the Issuer or its Subsidiaries at the request of the Administrative Agent shall so furnish each Noteholder with such Reports;

(b) expressly agrees and acknowledges that the Administrative Agent does not (i) make any representation or warranty as to the accuracy of any Report, and (ii) shall not be liable for any information contained in any Report;

(c) expressly agrees and acknowledges that the Reports are not comprehensive audits or examinations, that any party performing any audit or examination will inspect only specific information regarding the Issuer and its Subsidiaries and will rely significantly upon the Issuer and its Subsidiaries' books and records, as well as on representations of such Person's personnel;

(d) agrees to keep all Reports and other material, non-public information regarding the Issuer and its Subsidiaries and their operations, assets, and existing and contemplated business plans in a confidential manner in accordance with <u>Section 12.8</u>; and

(e) without limiting the generality of any other indemnification provision contained in this Agreement, agrees: (i) to hold the Administrative Agent and any other Noteholder preparing a Report harmless from any action the indemnifying Noteholder may take or fail to take or any conclusion the indemnifying Noteholder may reach or draw from any Report in connection with any loans or other credit accommodations that the indemnifying Noteholder has made or may make to the Issuer, or the indemnifying Noteholder's participation in, or the indemnifying Noteholder's purchase of, a loan or loans or notes of the Issuer, and (ii) to pay and protect, and indemnify, defend and hold Administrative Agent, and any such other Noteholder preparing a Report harmless from and against, the claims, actions, proceedings, damages, costs, expenses, and other amounts (including attorneys' fees and costs) incurred by Administrative Agent and any such other Noteholder preparing a Report as the direct or indirect result of any third parties who might obtain all or part of any Report through the indemnifying Noteholder.

**13.10 Erroneous Payments.** If a payment is made by the Administrative Agent (or its Affiliates) in error (whether known to the recipient or not) or if a Noteholder or another recipient of funds is not otherwise entitled to receive such funds at such time of such payment or from such Person in accordance with the Note Documents, then such Noteholder or recipient shall forthwith on demand repay to the Administrative Agent the portion of such payment that was made in error (or otherwise not intended (as determined by the Administrative Agent) to be received) in the amount made available by the Administrative Agent (or its Affiliate) to such Noteholder or recipient. Each Noteholder and other party hereto waives the discharge for value defense in respect of any such payment.

# 14. NOTEHOLDER REPRESENTATIONS.

Each Noteholder party hereto represents and warrants to the Issuer on the Closing Date and, solely with respect to the any Noteholder purchasing a Note issued hereunder after the Closing Date, at the time of each Note issuance after the Closing Date that:

14.1 Accredited Investor and Qualified Purchaser. It is an "accredited investor" as that term is defined in Rule 501 of the Securities Act and that, in making the purchases contemplated herein, it is specifically understood and agreed that such Noteholder is acquiring the Notes for the purpose of investment and not with a view towards the sale or distribution thereof within the meaning of the Securities Act.

**14.2 Securities Act**. It understands that the Notes will not be registered under the Securities Act by reason of their issuance by the Issuer in a transaction exempt from the registration requirements of the Securities Act, and that it may have to hold the Notes indefinitely unless a subsequent disposition thereof is registered under the Securities Act and applicable state securities Laws or is exempt from registration or qualification by prospectus. The Noteholder has such knowledge and experience in financial and business matters as to be capable of evaluating the merits and risks of its prospective investment in the Notes and has the ability to bear the economic risks of the prospective investment.

**14.3 Due Authorization**. It has full power and authority to enter into this Agreement. This Agreement, when executed and delivered by it, will constitute valid and legally binding obligations of each Noteholder, enforceable in accordance with their terms, except as limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance, and any other Laws of general application affecting enforcement of creditors' rights generally, and as limited by Laws relating to the availability of specific performance, injunctive relief, or other equitable remedies.

**14.4 Rule 144**. Such Noteholder further understands that the exemption from registration afforded by Rule 144 promulgated under the Securities Act depends on the satisfaction of various conditions, and that, if applicable, Rule 144 may afford the basis for sales only in limited amounts.

14.5 Brokerage. Such Noteholder did not employ any broker or finder in connection with the transactions contemplated in this Agreement.

**14.6 Independent Investigation**. It has made its own independent investigation of the financial condition and affairs of Issuer and its Subsidiaries in connection with the purchase of the Notes and that it has made and shall continue to make its own appraisal of the creditworthiness of Issuer and its Subsidiaries.

# 15. DEFINITIONS

**15.1 Definitions**. As used in the Note Documents, the word "shall" is mandatory, the word "may" is permissive, the word "or" is not exclusive, the words "includes" and "including" are not limiting, the singular includes the plural, and numbers denoting amounts that are set off in brackets are negative. As used in this Agreement, the following capitalized terms have the following meanings:

"Account" means any "account" as defined in the Code with such additions to such term as may hereafter be made, and includes, without limitation, all accounts receivable and other sums owing to the Issuer.

"Account Debtor" means any "account debtor" as defined in the Code with such additions to such term as may hereafter be made.

"Affiliate" means, with respect to any Person, each other Person that owns or controls directly or indirectly the Person, any Person that controls or is controlled by or is under common control with the Person, and each of that Person's senior executive officers, directors, partners and, for any Person that is a limited liability company, that Person's managers and members.

"Agent Fee Letter" means that certain fee letter dated as of the date hereof between the Administrative Agent and the Issuer.

"Agreement" has the meaning provided to such term in the preamble hereof.

"Antitrust Law" means any Law that is designed or intended to prohibit, restrict or regulate actions or transactions having the purpose or effect of monopolization, restraint of trade, harm to competition or effectuating foreign investment.

"Bankruptcy Code" means Title 11 of the United States Code entitled "Bankruptcy," as now and hereafter in effect, or any successor statute.

"**Books**" means all the Issuer's books and records including ledgers, federal and state tax returns, records regarding the Issuer's assets or liabilities, the Collateral, business operations or financial condition, and all computer programs or storage or any equipment containing such information.

"**Borrowing Resolutions**" means, with respect to any Person, those resolutions adopted by such Person's board of directors (and, if required under the terms of such Person's Operating Documents, stockholders) and delivered by such Person to the Administrative Agent and the Noteholders approving the Note Documents to which such Person is a party and the transactions contemplated thereby (including the issuance of the Notes), together with a certificate executed by its secretary on behalf of such Person certifying (a) such Person has the authority to execute, deliver, and perform its obligations under each of the Note Documents to which it is a party, (b) that set forth as a part of or attached as an exhibit to such certificate is a true, correct, and complete copy of the resolutions then in full force and effect authorizing and ratifying the execution, delivery, and performance by such Person of the Note Documents to which it is a party, (c) the name(s) of the Person(s) authorized to execute the Note Documents on behalf of such Person, together with a sample of the true signature(s) of such Person(s), and (d) that the Administrative Agent may conclusively rely on such certificate unless and until such Person shall have delivered to the Administrative Agent a further certificate canceling or amending such prior certificate.

"Business Day" means any day that is not a Saturday, Sunday or a day on which commercial banks in New York City and any other city in which the office of the Administrative Agent is authorized or required by law to remain closed.

"CARES Act" means (a) the Coronavirus Aid, Relief, and Economic Security Act (Pub. L. 116-136) and any administrative or other guidance published with respect thereto by any Governmental Authority (including IRS Notices 2020-22 and 2020-65), or any other Law or executive order or executive memorandum (including the Memorandum on Deferring Payroll Tax Obligations in Light of the Ongoing COVID-19 Disaster, dated August 8, 2020) intended to address the consequences of the COVID-19 Pandemic (in each case, including any comparable provisions of federal, state, local or non-U.S. Law and including any related or similar orders or declarations from any Governmental Authority) and (b) any extension of, amendment, supplement, correction, revision or similar treatment to any provision of the CARES Act contained in the Consolidated Appropriations Act, 2021, H.R. 133.

"Cash Burn" means, as of any date of determination, the amount equal to the Issuer's and its Subsidiaries' total unrestricted cash as of the first (1st) day of the immediately preceding three (3) month period <u>minus</u> the Issuer's and its Subsidiaries' total unrestricted cash balance as of the last day of such three (3) month period, as adjusted to exclude any changes in the cash balance as a result of debt and equity financing activities; *provided*, that the calculation of Cash Burn, and any determination of compliance with <u>Section 7.9</u>, shall not give effect to any cash transactions occurring prior to the Closing Date such that the calculation of Cash Burn for all periods prior to the Closing Date shall be deemed to be \$0.

"Cash Equivalents" means (a) marketable direct obligations issued or unconditionally guaranteed by the United States or any agency or any State thereof having maturities of not more than one (1) year from the date of acquisition; (b) commercial paper maturing no more than one (1) year after its creation and having the highest rating from either Standard & Poor's Ratings Group or Moody's Investors Service, Inc.; and (c) money market funds at least ninety-five percent (95%) of the assets of which constitute Cash Equivalents of the kinds described in <u>clauses (a)</u> and (b) of this definition.

"Change in Control" means (a) at any time, any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act")), shall become, or obtain rights (whether by means or warrants, options or otherwise) to become, the "beneficial owner" (as defined in Rules 13(d)-3 and 13(d)-5 under the Exchange Act), directly or indirectly, of forty-nine percent (49.0%) or more of the ordinary voting power for the election of directors of Eargo (determined on a fully diluted basis); (b) during any period of twelve (12) consecutive months, a majority of the members of the board of directors or other equivalent governing body of Eargo cease to be composed of individuals (i) who were members of that board or equivalent governing body on the first day of such

period, (ii) whose election or nomination to that board or equivalent governing body was approved by individuals referred to in clause (i) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body or (iii) whose election or nomination to that board or other equivalent governing body was approved by individuals referred to in clauses (i) and (ii) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body; (c) at any time, Eargo shall cease to own and control, of record and beneficially, directly or indirectly, one hundred percent (100%) of each class of outstanding capital stock of each Subsidiary of Eargo (unless such Subsidiary is dissolved, merged, consolidated, or liquidated into Issuer) free and clear of all Liens (except Liens created by this Agreement); or (d) a sale of all or substantially all of the assets of the Eargo.

## "Closing Date" means June 24, 2022.

"**Code**" means the Uniform Commercial Code, as the same may, from time to time, be enacted and in effect in the State of New York; <u>provided</u>, that, to the extent that the Code is used to define any term herein or in any Note Document and such term is defined differently in different Articles or Divisions of the Code, the definition of such term contained in Article or Division 9 shall govern; <u>provided</u>, <u>further</u>, that in the event that, by reason of mandatory provisions of Law, any or all of the attachment, perfection, or priority of, or remedies with respect to, the Administrative Agent's Lien on any Collateral is governed by the Uniform Commercial Code in effect in a jurisdiction other than the State of New York, the term "**Code**" shall mean the Uniform Commercial Code as enacted and in effect in such other jurisdiction solely for purposes of the provisions thereof relating to such attachment, perfection, priority, or remedies and for purposes of definitions relating to such provisions.

"Collateral" means any and all properties, rights and assets of Issuer described on Exhibit A.

"Collateral Documents" means this Agreement, the IP Agreement, any Control Agreement, and all other instruments, documents and agreements delivered by the Issuer pursuant to this Agreement or any of the other Note Documents in order to grant to Administrative Agent, for the benefit of Secured Parties, a Lien on any real, personal or mixed property of the Issuer constituting Collateral as security for the Obligations, in each case, as such Collateral Documents may be amended or otherwise modified from time to time.

"Collateral Account" means any Deposit Account, Securities Account, or Commodity Account.

"Commitment" means, with respect to each Noteholder, such Noteholder's First Tranche Commitment and Second Tranche Commitment.

"Commodity Account" means any "commodity account" as defined in the Code with such additions to such term as may hereafter be made.

"Common Stock" has the meaning provided to such term in Section 5.1(c).

"Compliance Statement" means that certain statement in the form attached hereto as Exhibit C.

"Contingent Obligation" means, for any Person, any direct or indirect liability, contingent or not, of that Person for (a) any indebtedness, lease, dividend, letter of credit or other obligation of another such as an obligation, in each case, directly or indirectly guaranteed, endorsed, co-made, discounted or sold with recourse by that Person, or for which that Person is directly or indirectly liable; (b) any obligations for undrawn letters of credit for the account of that Person; and (c) all obligations from any interest rate, currency or commodity swap agreement, interest rate cap or collar agreement, or other agreement or

arrangement designated to protect a Person against fluctuation in interest rates, currency exchange rates or commodity prices; but "Contingent Obligation" does not include endorsements, warranties, or indemnities in the ordinary course of business. The amount of a Contingent Obligation is the stated or determined amount of the primary obligation for which the Contingent Obligation is made or, if not determinable, the maximum reasonably anticipated liability for it determined by the Person in good faith; but the amount may not exceed the maximum of the obligations under any guarantee or other support arrangement.

"Control Agreement" means any control agreement, in form and substance satisfactory to the Administrative Agent and the Required Noteholders, entered into among the depository institution at which the Issuer maintains a Deposit Account or the securities intermediary or commodity intermediary at which Issuer maintains a Securities Account or a Commodity Account, in each case, pursuant to which the Administrative Agent obtains control (within the meaning of the Code) over such Deposit Account, Securities Account, or Commodity Account.

"Conversion Date" has the meaning provided to such term in Section 2.1.1(b)(ii).

"Copyrights" means any and all copyright rights, copyright applications, copyright registrations and like protections in each work of authorship and derivative work thereof, whether published or unpublished and whether or not the same also constitutes a trade secret.

"COVID-19 Actions" means all reasonable actions taken, planned or planned to be taken in response to events, occurrences, conditions, circumstances, or developments arising directly or indirectly as a result of the COVID-19 Pandemic.

"COVID-19 Measures" means any quarantine, "shelter in place," "stay at home," workforce reduction, social distancing, shut down, closure, sequester, safety or similar Law, directive or guideline issued by any Governmental Authority having authority over the Issuer, including the Centers for Disease Control and Prevention and the World Health Organization, in each case, in connection with or in response to the COVID-19 Pandemic, including the CARES Act, the Families First Act and any other COVID-19 Pandemic relief measure hereafter adopted by any Governmental Authority.

"COVID-19 Pandemic" means the SARS-Cov2 or COVID-19 pandemic, including any future resurgence or evolutions, mutations or variations thereof and/or any related or associated disease outbreaks, epidemics, pandemics or public health emergencies.

"Data Protection Obligations" has the meaning provided to such term in Section 5.13.

"Default" means a condition or event that, after notice or lapse of time or both, would constitute an Event of Default.

"Default Rate" has the meaning provided to such term in Section 2.2(b).

"Default Interest" has the meaning provided to such term in Section 2.2(b).

"Deposit Account" means any "deposit account" as defined in the Code with such additions to such term as may hereafter be made.

"**Division**" means, in reference to any Person which is an entity, the division of such Person into two (2) or more separate Persons, with the dividing Person either continuing or terminating its existence as part of such division, including as contemplated under Section 18-217 of the Delaware Limited Liability Company Act for limited liability companies formed under Delaware Law, or any analogous action taken pursuant to any other applicable Law with respect to any corporation, limited liability company, partnership or other entity.

"**Dollars**," "**dollars**" or use of the sign "\$" means only lawful money of the United States and not any other currency, regardless of whether that currency uses the "\$" sign to denote its currency or may be readily converted into lawful money of the United States.

"Domain Name" means any domain name, uniform resource locator or social media account or handle.

"Environmental Permit" means any permit, license, approval or other authorization under any applicable Law, rule or regulations of the United States or of any state, municipality or other subdivision thereof relating to pollution or protection of health or the environment, including Laws, regulations or other requirements relating to emissions, discharges, releases or threatened releases of pollutants, contaminants or Hazardous Substances or toxic materials or wastes into ambient air, surface water, ground water or land, or otherwise relating to the manufacture, processing, distribution, recycling, presence, use, treatment, storage, disposal, transport, or handling of, wastes, pollutants, contaminants or Hazardous Substances.

"Equipment" means all "equipment" as defined in the Code with such additions to such term as may hereafter be made, and includes without limitation all machinery, fixtures, goods, vehicles (including motor vehicles and trailers), and any interest in any of the foregoing.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Documents" means all material "employment benefit plans" as defined in Section 3(3) of ERISA that are maintained or sponsored by the Issuer or its Subsidiaries for the benefit of their respective current or former employees and with respect to which the Issuer or its Subsidiaries have any liability.

"Event of Default" has the meaning provided to such term in Section 8.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Excluded Taxes" means any of the following Taxes imposed on or with respect to a Secured Party or required to be withheld or deducted from a payment to a Secured Party, (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes, and branch profits Taxes, in each case, (i) imposed as a result of such Secured Party being organized under the laws of, or having its principal office or, in the case of any Noteholder, its applicable lending office located in, the jurisdiction imposing such Tax (or any political subdivision thereof) or (ii) that are Other Connection Taxes, (b) in the case of a Noteholder, U.S. federal withholding Taxes imposed on amounts payable to or for the account of such Noteholder with respect to an applicable interest in a Note or Commitment pursuant to a law in effect on the date on which (i) such Noteholder acquires such interest in the Note or Commitment or (ii) such Noteholder changes its lending office, except in each case to the extent that, pursuant to Section 2.7 amounts with respect to such Taxes were payable either to such Noteholder's assignor immediately before such Noteholder became a party hereto or to such Noteholder immediately before it changed its lending office, (c) Taxes attributable to such Secured Party's failure to comply with <u>Section 2.7.4</u> and (d) any withholding Taxes imposed under FATCA. If any Secured Party is required to withhold or deduct any Tax from a direct or indirect owner of such Secured Party in respect of any payment or accrual relating to amounts received from the Issuer under this Agreement, such Tax shall be considered an "Excluded Tax" if such Tax would be an Excluded Tax had such direct or indirect owner of such Secured Party been a Secured Party hereunder and complied with the provisions of <u>Section 2.7.4</u>.

"Families First Act" means the Families First Coronavirus Response Act, as signed into Law by the President of the United States on March 18, 2020.

"FATCA" means Sections 1471 through 1474 of the IRC, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future regulations or official interpretations thereof, any agreements entered into pursuant to Section 1471(b)(1) of the IRC and any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreement, treaty or convention among Governmental Authorities and implementing such Sections of the IRC.

"First Tranche Closing" has the meaning provided to such term in Section 2.1.1(a)(ii).

"First Tranche Closing Date" has the meaning provided to such term in Section 2.1.1(a)(ii).

"First Tranche Commitment" means, with respect to each Noteholder, the commitment of such Noteholder to purchase the First Tranche Notes in the amount set forth opposite such Noteholder's name on <u>Schedule I</u> hereto under the heading "First Tranche Commitment" on the First Tranche Closing Date. The aggregate amount of the First Tranche Commitments as of the Closing Date is \$100,000,000.

"First Tranche Notes" means the Notes issued pursuant to Section 2.1.1(a) on the First Tranche Closing Date.

"GAAP" means generally accepted accounting principles set forth in the opinions and pronouncements of the Accounting Principles Board of the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or in such other statements by such other Person as may be approved by a significant segment of the accounting profession, which are applicable to the circumstances as of the date of determination.

"General Intangibles" means all "general intangibles" as defined in the Code in effect on the date hereof with such additions to such term as may hereafter be made, and includes without limitation, all Intellectual Property, claims, income and other tax refunds, security and other deposits, payment intangibles, contract rights, options to purchase or sell real or personal property, rights in all litigation presently or hereafter pending (whether in contract, tort or otherwise), insurance policies (including key man, property damage, and business interruption insurance), payments of insurance and rights to payment of any kind.

"Governmental Approval" means any consent, authorization, approval, order, license, franchise, permit, certificate, accreditation, registration, filing or notice, of, issued by, from or to, or other act by or in respect of, any Governmental Authority.

"Governmental Authority" means any nation or government, any state or other political subdivision thereof, any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative functions of or pertaining to government, any securities exchange and any self-regulatory organization.

"Government Official" means any officer or employee of a foreign governmental authority or any department, agency, or instrumentality thereof, or of a public international organization, or any person acting in an official capacity for or on behalf of any such foreign governmental authority or department, agency, or instrumentality, or for or on behalf of any such public international organization, or any political party, party official, or candidate thereof, excluding officials of the governments of the United States, the several states thereof, any local subdivision of any of them or any agency, department or unit of any of the foregoing.

"Hazardous Substance" means any waste, substance, product or material defined or regulated as "hazardous" or "toxic" by any applicable Law, rule, regulation or order described in the definition of "Requirements of Environmental Law", including petroleum and any fraction thereof, and any radioactive materials and waste.

"Health Care Laws" means the Federal Food, Drug, and Cosmetic Act, the Anti-Kickback Statute (42 U.S.C. Section 1320a-7b(b)), the Civil Monetary Penalties Law (42 U.S.C. § 1320a-7a), the Civil False Claims Act (31 U.S.C. Section 3729 et seq.), the criminal False Claims Law (42 U.S.C. § 1320a-7b(a)), all criminal Laws relating to health care fraud and abuse, including 18 U.S.C. Sections 286 and 287 and the health care fraud criminal provisions under HIPAA, the exclusion Laws (42 U.S.C. § 1320a-7), and any and all other similar state, local or federal health care Laws and the regulations promulgated pursuant to such Laws, each as amended from time to time.

"HIPAA" has the meaning provided to such term in Section 5.13.

"Indebtedness" means (a) indebtedness for borrowed money or the deferred price of property or services, such as reimbursement and other obligations for surety bonds and letters of credit, (b) obligations evidenced by notes, bonds, debentures or similar instruments, (c) capital lease obligations, and (d) Contingent Obligations.

"Indemnified Liabilities" means, collectively, any and all liabilities, obligations, losses, damages, penalties, claims, documented out-of-pocket costs, expenses and disbursements of any kind or nature whatsoever (including the reasonable fees and disbursements of counsel for Indemnitees and any fees or expenses incurred by Indemnitees in enforcing this indemnity), whether direct, indirect or consequential and whether based on any federal, state or foreign Laws, statutes, rules or regulations (including securities and commercial Laws, statutes, rules or regulations and environmental Laws), on common Law or equitable cause or on contract or otherwise, that may be imposed on, incurred by, or asserted against any such Indemnitee, in any manner relating to or arising out of this Agreement or the other Note Documents or the transactions contemplated hereby or thereby (including the Noteholders' agreement to make purchases or the use or intended use of the proceeds thereof, or any enforcement of any of the Note Documents (including any sale of, collection from, or other realization upon, any of the Collateral)).

"Indemnified Taxes" means (a) Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of the Issuer under any Note Document and (b) to the extent not otherwise described in <u>clause (a)</u>, other taxes as described in <u>Section 2.7.5</u>.

"**Insolvency Proceeding**" means any proceeding by or against any Person under the United States Bankruptcy Code, or any other bankruptcy or insolvency Law, including assignments for the benefit of creditors, compositions, extensions generally with its creditors, or proceedings seeking reorganization, arrangement, or other relief.

"Intellectual Property" means, with respect to any Person, all of such Person's right, title, and interest in and to the following:

(a) its Copyrights, Trademarks, Patents and Domain Names;

(b) any and all Trade Secrets; ;

(c) any and all software including source code, object code, algorithms, application programming interfaces and documentation;

(d) any and all design rights and mask work rights which may be available to such Person;

(e) any and all other intellectual property or proprietary rights;

(f) any and all claims for damages by way of past, present and future infringement, misappropriation or violation of any of the foregoing <u>clauses</u> (a) through (f), with the right, but not the obligation, to sue for and collect such damages for said use, infringement, misappropriation or other violation of the rights identified above in clauses (a) through (f); and

(g) all amendments, renewals and extensions of any of the foregoing.

"Interest Payment Date" has the meaning provided to such term in Section 2.2(b).

"**Inventory**" means all "inventory" as defined in the Code in effect on the date hereof with such additions to such term as may hereafter be made, and includes without limitation all merchandise, raw materials, parts, supplies, packing and shipping materials, work in process and finished products, including such inventory as is temporarily out of Issuer's custody or possession or in transit and including any returned goods and any documents of title representing any of the above.

"Investment" means any beneficial ownership interest in any Person (including stock, partnership interest or other securities), and any loan, advance or capital contribution to any Person.

"Investor Rights Agreement" means that certain Investor Rights Agreement, dated of even date herewith, by and among Eargo and the Investors (as defined therein).

"IP Agreement" means that certain Intellectual Property Security Agreement between Issuer and the Administrative Agent, dated as of the Closing Date, as may be amended, modified, supplemented or restated from time to time.

"IRC" means the Internal Revenue Code of 1986, as amended.

"Issuance Request" means an irrevocable written notice substantially in the form of Exhibit D.

"Issuer" is defined in the preamble hereof. References herein to "Issuer" mean each Issuer individually and collectively with each other Issuer. For purposes of <u>Section 5</u>, <u>Section 6</u>, <u>Section 7</u>, <u>Section 8</u>, and <u>Section 9</u>, any reference to "Issuer" shall be deemed to include any Subsidiary of the Issuer.

"IT Systems" has the meaning provided to such term in Section 5.13.

"Key Person" means any of the Chief Executive Officer, Chief Financial Officer, Chief Technology Officer, Chief Operating Officer, or any other senior vice president or above, in each case, of Eargo's management team, whether such Person is a Key Person on the Closing Date or subsequently becomes a Key Person after the Closing Date. As of the Closing Date, each Key Person of Eargo is set forth on Section 7.2(c) of the Disclosure Schedule.

"Law" means any law (including common law), constitution, statute, treaty, regulation, rule, ordinance, code, ruling, or order of, including the administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, or any agreement with, any Governmental Authority.

"Lien" means a claim, mortgage, deed of trust, levy, charge, pledge, security interest or other encumbrance of any kind, whether voluntarily incurred or arising by operation of Law or otherwise against any property.

"Material Adverse Effect" means an effect that reasonably would be expected to be a material adverse effect upon the financial condition, assets or results of operations of the Issuer and its Subsidiaries, taken as a whole; provided, however, that any such effect resulting or arising from or relating to any of the following matters shall not be considered when determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur: (a) any change, development, occurrence or event affecting the industry in which the Issuer and its Subsidiaries operate; (b) any conditions affecting the United States general economy or the general economy in any geographic area in which the Issuer or its Subsidiaries operate or developments or changes therein or the financial and securities markets and credit markets in the United States or elsewhere in the world; (c) political conditions, including the continuation, occurrence, escalation, outbreak or worsening of any hostilities, war, political action, acts of terrorism, sabotage or military conflicts, whether or not pursuant to the declaration of an emergency or war; (d) any conditions resulting from the existence, occurrence or continuation of any force majeure events, including any earthquakes, floods, hurricanes, tropical storms, fires or other natural or manmade disasters, or any other national, international ore regional calamity; (e) conditions or effects relating to any epidemic, pandemic or disease outbreak (including the COVID-19 Pandemic), or any COVID-19 Action, COVID-19 Measures or any change in COVID-19 Measures; (f) changes in any Law, rule, regulation or GAAP; (g) any action taken or omitted to be taken by or at the written request or with the written consent of a Purchaser; (h) any announcement of this Agreement or the transactions contemplated hereby; (i) changes in the market price or trading volume of Common Stock or any other equity, equityrelated or debt securities of the Issuer or its Affiliates (it being understood that the underlying circumstances, events or reasons giving rise to any such change can be taken into account in determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur); (j) any failure to meet any internal or public projections, forecasts, estimates or guidance for any period (it being understood that the underlying circumstances, events or reasons giving rise to any such failure can be taken into account in determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur); or (k) any effect arising out of or resulting from any claims or proceedings made by any of the Issuer's stockholders arising out of or related to this Agreement; provided, that any of the matters described in clauses (a), (b), (c), (d), (e), or (f), will be taken into account for purposes of determining whether or not a Material Adverse Effect has occurred to the extent that such matter disproportionately and adversely affects the Issuer and its Subsidiaries, taken as a whole, as compared with other companies operating in the industry in which the Issuer and its Subsidiaries operate.

"Maturity Date" means the earlier of (a) the one year anniversary of the First Tranche Closing Date *plus* one day and (b) the date that the Notes shall become due and payable in full hereunder, whether by acceleration or otherwise.

"Nasdaq" means The Nasdaq Stock Market LLC.

"Note" means each of the First Tranche Notes and the Second Tranche Notes, in each case, substantially in the form of Exhibit E attached hereto.

"Note Documents" means, collectively, this Agreement, each Collateral Document, each Issuance Request, the Agent Fee Letter, the Investor Rights Agreement, the Registration Rights Agreement, and any schedules, exhibits, certificates, notices, and any other documents related to this Agreement, any subordination agreement, any note, or notes or guaranties executed by Issuer, and any other present or future agreement by Issuer with or for the benefit of the Administrative Agent in connection with this Agreement, all as amended, restated, or otherwise modified.

"Noteholder" means each Person listed on <u>Schedule I</u> hereto under the heading "Noteholder", and each such Person's successors and assigns as permitted hereunder, each of which is referred to herein as a "Noteholder."

"**Obligations**" means the Issuer's obligation to (x) pay when due any debts, principal, interest, fees, expenses, the Repayment Value and other amounts Issuer owes to the Secured Parties now or later, whether under this Agreement, the other Note Documents or otherwise, including any amounts owing to the Administrative Agent under the Note Documents and interest accruing after Insolvency Proceedings begin and debts, liabilities, or obligations of Issuer assigned to the Administrative Agent, and (y) perform Issuer's duties under the Note Documents. Notwithstanding the foregoing, <u>clause (x)</u> of the immediately preceding sentence shall not include any obligations (i) with respect to the Conversion Shares, (ii) under the Investor Rights Agreement, (iii) under the Registration Rights Agreement (iv) under any directors and officers indemnification agreement or similar instrument or (v) otherwise with respect to any equity instruments. For the avoidance of doubt, "Obligations" does not include any interest of the Noteholders in the equity interests of the Issuer.

"Operating Documents" means, for any Person, such Person's formation documents, as certified by the Secretary of State (or equivalent agency) of such Person's jurisdiction of organization on a date that is no earlier than thirty (30) days prior to the Closing Date, the First Tranche Closing Date or the Second Tranche Closing Date, as applicable, and, (a) if such Person is a corporation, its bylaws in current form, (b) if such Person is a limited liability company, its limited liability company agreement (or similar agreement), and (c) if such Person is a partnership, its partnership agreement (or similar agreement), each of the foregoing with all current amendments or modifications thereto.

"Other Connection Taxes" means, with respect to any Secured Party, Taxes imposed as a result of a present or former connection between such Secured Party and the jurisdiction imposing such Tax (other than connections arising from such Secured Party having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced any Note Document, or sold or assigned an interest in any Note or Note Document).

"**Patents**" means all patents, patent applications and like protections including improvements, divisions, provisionals, continuations, renewals, re-examinations, substitutions, reissues, extensions, restorations and continuations-in-part of the same and supplementary protection certifications and inventor's certificates.

"Perfection Certificate" has the meaning provided to such term in Section 5.1(a).

"Permitted Indebtedness" means:

(a) the Issuer's Indebtedness to the Secured Parties under this Agreement and the other Note Documents;

(b) unsecured Indebtedness to trade creditors incurred in the ordinary course of business;

(c) Indebtedness incurred as a result of endorsing negotiable instruments received in the ordinary course of business;

(d) Indebtedness secured by Liens permitted under <u>clause (c)</u> of the definition of "Permitted Liens" hereunder, and any extensions, refinancings, modifications, amendments and restatements thereof, <u>provided</u> that the principal amount thereof is not increased or the terms thereof are not modified to impose more burdensome terms upon Borrower or its Subsidiary, as the case may be;

(e) Indebtedness in an aggregate amount not to exceed \$100,000 with respect to surety, indemnity, or appeal bonds and similar obligations in the ordinary course of business, and any extensions, refinancings, modifications, amendments and restatements thereof, <u>provided</u> that the principal amount thereof is not increased or the terms thereof are not modified to impose more burdensome terms upon Borrower or its Subsidiary, as the case may be;

(f) letters of credit obtained with respect to leases of real property, contract manufacturers or supply-chain matters, in each case, in the ordinary course of business;

(g) Investments by one Issuer in or to another Issuer, including intercompany Indebtedness;

(h) customer deposits and advance payments received in the ordinary course of business;

(i) other unsecured Indebtedness not to exceed \$100,000 in the aggregate amount outstanding at any time;

(j) any other Indebtedness approved in writing by the Noteholders;

(k) prior to the First Tranche Closing Date, the "Obligations" under (and as defined in) the Existing Loan Agreement; and

(1) unsecured Indebtedness under business credit cards used in the ordinary course of business in an aggregate amount at any time outstanding not exceeding \$500,000.

"Permitted Investments" means:

(a) Investments existing on the Closing Date, which are listed on the Perfection Certificate;

(b) Investments consisting of Cash Equivalents and any Investments permitted by the Issuer's investment policy, as amended from time to time, provided that such investment policy (and any such amendment thereto) has been approved in writing by the Required Noteholders;

(c) Investments consisting of the endorsement of negotiable instruments for deposit or collection or similar transactions in the ordinary course of the Issuer;

(d) Investments consisting of deposit accounts in which the Administrative Agent has a perfected security interest;

(e) Investments accepted in connection with Transfers permitted by Section 7.1;

(f) Investments by one Issuer in or to another Issuer, including intercompany Indebtedness;

(g) Investments consisting of (i) travel advances and employee relocation loans and other employee loans and advances in the ordinary course of business and (ii) loans to employees, officers, or directors of the Issuer relating to the purchase of equity securities of the Issuer pursuant to employee stock purchase plans or agreements approved by the Issuer's board of directors and approved in writing by the Required Noteholders;

(h) Investments (including debt obligations) received in connection with the bankruptcy or reorganization of customers or suppliers and in settlement of delinquent obligations of, and other disputes with, customers or suppliers arising in the ordinary course of business;

(i) Investments consisting of notes receivable of, or prepaid royalties and other credit extensions, to customers and suppliers who are not Affiliates in the ordinary course of business; *provided* that this <u>paragraph (i)</u> shall not apply to Investments of any Issuer in any other Issuer;

(j) non-cash loans to employees, officers, or directors relating to the purchase of equity securities of the Issuer pursuant to employee stock purchase plans or equity compensation arrangements approved by the Issuer's board of directors;

(k) Investments in connection with joint ventures or strategic alliances or collaboration of Borrower or a Subsidiary in the ordinary course of Borrower's business consisting of the non-exclusive licensing of technology, the development of technology, or the providing of technical support, provided that any cash invested in connection therewith shall not exceed \$250,000 in the aggregate in any fiscal year; and

(1) other Investments not otherwise permitted by this Agreement not exceeding \$100,000 in the aggregate outstanding at any time.

#### "Permitted Liens" means:

(a) Liens arising under this Agreement and the other Note Documents;

(b) Liens for taxes, fees, assessments or other government charges or levies either (i) not due and payable or (ii) being contested in good faith and for which the Issuer maintains adequate reserves in accordance with GAAP; *provided* that no notice of any such Lien has been filed or recorded under the IRC and the Treasury Regulations adopted thereunder;

(c) purchase money Liens or capital leases (i) on Equipment (including additions, accessions, and improvements thereto and the proceeds thereof) acquired or held by the Issuer incurred for financing the acquisition of the Equipment securing no more than \$500,000 in the aggregate at any time outstanding or (ii) existing on Equipment when acquired, if such Lien is confined to such Equipment and additions, accessions, and improvements thereto and the proceeds thereof;

(d) Liens of carriers, warehousemen, suppliers, or other Persons that are possessory in nature and arising in the ordinary course of business, so long as such Liens attach only to Inventory and secure liabilities in an aggregate amount not exceeding \$150,000 and that either (x) are not delinquent or (y) are being contested in good faith and by appropriate proceedings which proceedings have the effect of preventing the forfeiture or sale of the property subject thereto and for which the Issuer maintains adequate reserved in accordance with GAAP;

(e) Liens to secure payment of workers' compensation, employment insurance, old-age pensions, social security and other like obligations incurred in the ordinary course of business (other than Liens imposed by ERISA);

(f) leases of real property granted in the ordinary course of the Issuer's business, and leases, subleases, non-exclusive licenses or sublicenses of personal property (other than Intellectual Property) granted in the ordinary course of the Issuer's business, so long as such leases, subleases, licenses and sublicenses do not prohibit granting the Secured Parties a security interest therein;

(g) non-exclusive licenses of Intellectual Property granted in the ordinary course of business consistent with past practice;

(h) Liens arising from attachments or judgments, orders, or decrees in circumstances not constituting an Event of Default under <u>Sections 8.4</u> and <u>8.7</u>;

(i) Liens in favor of financial institutions arising in connection with the Issuer's deposit and/or securities accounts held at such institutions; *provided*, that the Administrative Agent has a perfected security interest in such accounts;

(j) deposits in an aggregate outstanding amount not to exceed \$150,000 to secure the security, indemnity, or appeal bonds and similar obligations arising in the ordinary course of business described in <u>clause (e)</u> of the definition of "Permitted Indebtedness" or to secure the performance of bids, trade contracts (other than for borrowed money), contracts for the purchase of property, and leases in the ordinary course of business;

(k) Liens in favor of customs and revenue authorities arising as a matter of Law to secure payments of custom duties in connection with the importation of goods;

(l) Liens on property (other than Intellectual Property) securing obligations in an aggregate principal amount at any time outstanding not to exceed \$100,000;

(m) Liens in respect of any cash collateral securing a letter of credit permitted pursuant to <u>clause (f)</u> of the definition of "Permitted Indebtedness"; and

(n) prior to the First Tranche Closing Date, Liens securing the "Obligations" under (and as defined in) the Existing Loan Agreement.

"**Person**" means any individual, sole proprietorship, partnership, limited liability company, joint venture, company, trust, unincorporated organization, association, corporation, institution, public benefit corporation, firm, joint stock company, estate, entity or government agency.

"Personal Data" has the meaning provided to such term in Section 5.13.

"Preferred Stock" has the meaning provided to such term in Section 5.1(c).

"Redemption Date" has the meaning provided to such term in Section 2.6.

"Registered Organization" means any "registered organization" as defined in the Code with such additions to such term as may hereafter be made.

"Registration Rights Agreement" means that certain Registration Rights Agreement, dated as of even date herewith, by and among Eargo and the Investors (as defined therein).

"**Repayment Value**" means, as to any Note on any applicable date, an amount equal to the greater of (x) an amount such that the annualized return (excluding any Default Interest) on the principal amount (excluding any PIK Interest) of such Note held by such Noteholder on such date is no less than 12.00% and (y) the principal amount of such Note (excluding any PIK Interest) *multiplied by* 1.50. For the avoidance of doubt, references to the Repayment Value with respect to all outstanding Notes means the Repayment Value of all Notes outstanding at such time, excluding the amount of any principal comprised of PIK Interest.

"Repayment Value Excess" means, as to any Note on any applicable date, an amount equal to the Repayment Value with respect to such Note *minus* the principal amount of such Note prior to the Repayment Value becoming payable.

"**Required Noteholders**" means (i) for purposes of <u>Section 9.1</u>, Noteholders holding at least 25.0% of the aggregate amount of Notes outstanding on any date of determination and (ii) for all other purposes, Noteholders holding at least 50.1% of the aggregate amount of Notes outstanding on any date of determination.

"**Requirement of Law**" means, as to any Person, the organizational or governing documents of such Person, and any Law (statutory or common), treaty, rule or regulation or determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject.

"Requirements of Environmental Law" means all requirements imposed by any Law (including the Resource Conservation and Recovery Act, the Comprehensive Environmental Response, Compensation, and Liability Act, the Clean Water Act, the Clean Air Act, and any state analogues of any of the foregoing), rule, regulation, or order of any governmental authority which relate to (a) pollution, protection or clean-up of the air, surface water, ground water or land; (b) solid, gaseous or liquid waste or Hazardous Substance generation, recycling, reclamation, release, threatened release, treatment, storage, disposal or transportation; (c) exposure of Persons or property to Hazardous Substances; or (d) the manufacture, presence, processing, distribution in commerce, use, discharge, releases, threatened releases, emissions or storage of Hazardous Substances into the environment.

"Responsible Officer" means any of the Chief Executive Officer, President, Chief Financial Officer, and Chief Operating Officer, in each case, of the Issuer, or any other duly appointed officer identified in writing from time to time by the Issuer to the Administrative Agent.

"**Restricted License**" means any material license or other material agreement with respect to which Issuer is the licensee (a) that prohibits or otherwise restricts Issuer from granting a security interest in Issuer's interest in such license or agreement or any other property (to the extent such restriction is enforceable under the Code), or (b) for which a default under or termination of could interfere with the Administrative Agent's right to sell any Collateral.

"**Rights**" means non-Transferable subscription rights to purchase Common Stock; *provided*, that, in the case of holders of record of Common Stock that are closed-end funds, such rights may be Transferred to funds Affiliated with such holders.

"**Rights Offering**" means (1) the distribution of Rights by Eargo, at no charge, to holders of record of its Common Stock, or, in the case of its stockholders who are closed funds, such stockholders' affiliated funds or other vehicles with substantially the same underlying beneficial owners as such closed funds, to subscribe for an aggregate of 375,000,000 shares (as adjusted in connection with stock divisions or combinations pursuant to <u>Section 2.5</u> hereof) of Common Stock at the Rights Offering Price and (2) the issuance and sale of Common Stock pursuant to the exercise of such Rights.

"Rights Offering Price" means \$0.50 per share of Common Stock (as adjusted for stock divisions or combinations pursuant to <u>Section 2.5</u> hereof).

"Rights Offering Shortfall Amount" has the meaning provided to such term in Section 2.5.3.

"Safety Notice" has the meaning provided to such term in Section 5.14(b).

"SEC" means the Securities and Exchange Commission, any successor thereto, and any analogous Governmental Authority.

"SEC Documents" means all reports, schedules, registration statements, proxy statements and other documents (including all amendments, exhibits and schedules thereto) filed by Eargo with the SEC on or after January 1, 2021.

"Second Closing" has the meaning provided to such term in Section 2.1.1(b)(ii).

"Second Tranche Closing Date" has the meaning provided to such term in Section 2.1.1(b)(ii).

"Second Tranche Commitment" means, with respect to each Noteholder, the commitment of such Noteholder to purchase the Second Tranche Notes in the amount set forth opposite such Noteholder's name on <u>Schedule I</u> hereto under the heading "Second Tranche Commitment" on the Second Tranche Closing Date. The aggregate principal amount of the Second Tranche Commitment on the Closing Date shall be equal to \$25,000,000; *provided*, that if the condition to the Second Tranche Commitment set forth in <u>Section 3.3(a)(ii)</u> has been satisfied (i.e., a Rights Offering is consummated within one hundred fifty (150) days of Closing Date and less than 75,000,000 shares are subscribed for and purchased in the Rights Offering), then such commitment and the corresponding principal amount of Notes issued at the Second Tranche Closing will be reduced by an amount equal to the quotient of the total gross proceeds from the Rights Offering *divided by* 1.50 (in which case, the commitments of each Noteholder will be reduced on a pro rata basis). Attached hereto as <u>Annex I</u> under the heading "\$187.5 Million Rights Offering Before Month Five" is an illustrative calculation of the foregoing.

"Second Tranche Commitment Termination Date" has the meaning provided to such term in Section 2.1.1(b)(i).

"Second Tranche Notes" means the Notes issued pursuant to Section 2.1.1(b) on the Second Tranche Closing Date.

"Secured Parties" means, collectively, the Administrative Agent and each Noteholder.

"Securities Account" means any "securities account" as defined in the Code with such additions to such term as may hereafter be made.

"Securities Act" means the Securities Act of 1933, as amended from time to time, and any successor statute.

"Stock Plans" means the Eargo, Inc. 2010 Equity Incentive Plan, the Eargo Inc. 2020 Incentive Award Plan and the Eargo Inc. 2020 Employee Stock Purchase Plan.

"Stockholder Approvals" means the Stockholder Charter Approval, the Stockholder Nasdaq Approval and any other approval of the Issuer's stockholders that is or would reasonably be expected to be required to give effect to (1) the transactions contemplated hereby and (2) the Rights Offering at every level of exercise by holders of Rights (i.e., no Rights are exercised, some Rights are exercised and all Rights are exercised).

"Stockholder Charter Approval" means the requisite approval of the Issuer's stockholders to increase the number of authorized shares of Common Stock under the Issuer's certificate of incorporation to an amount that is at least the number of authorized shares of Common Stock needed to consummate the Rights Offering (i.e., the number of issued and outstanding shares of Common Stock *plus* the number of shares issuable pursuant to outstanding securities, awards or other instruments that are convertible into or exercisable for shares of Common Stock *plus* 375,000,000 shares of Common Stock, in each case as of the closing of the Rights Offering).

"Stockholder Nasdaq Approval" means the requisite approval of Eargo's stockholders to issue the full potential amount of Conversion Shares hereunder pursuant to Rule 5635(b) and Rule 5635(d) of Nasdaq (i.e., the full amount of Conversion Shares that would be issued without regard to the application of such rules).

"Subsidiary" means, as to any Person, a corporation, partnership, limited liability company or other entity of which shares of stock or other ownership interests having ordinary voting power (other than stock or such other ownership interests having such power only by reason of the happening of a contingency) to elect a majority of the board of directors or other managers of such corporation, partnership or other entity are at the time owned, or the management of which is otherwise controlled, directly or indirectly through one or more intermediaries, or both, by such Person. Unless the context otherwise requires, each reference to a Subsidiary herein shall be a reference to a Subsidiary of Issuer.

"**Trade Secrets**" means any and all trade secrets, including any rights to unpatented inventions, discoveries, know how, confidential information, research, developments, materials, data and databases, processes, protocols, methods, techniques, models, specifications, instructions, technology, business plans, marketing strategies and operating manuals.

"**Trademarks**" means any trademarks, service marks, trade names, trade dress, brands, logos, slogans and other indicia of origin, whether registered or not, applications to register and registrations of the same and like protections, and the entire goodwill of the business of Borrower connected with and symbolized by the foregoing, but excluding any intent-to-use trademark application prior to the filing of a "Statement of Use" or "Amendment to Allege Use" with respect thereto, to the extent, if any, that, and solely during the period, if any, in which, the grant of a security interest therein would impair the validity or enforceability of such intent-to-use trademark application under applicable Law.

"Transfer" has the meaning provided to such term in Section 7.1.

[Signature page follows]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the Closing Date.

#### **ISSUER:**

### EARGO, INC.

By:	/s/ Christian Gormsen
Name:	Christian Gormsen
Title:	Chief Executive Officer and President

### EARGO HEARING, INC.

By:	/s/ Christian Gormsen
Name:	Christian Gormsen
Title:	Chief Executive Officer and President

#### EARGO SCREENING, LLC

By: /s/ Mark Thorpe Name: Mark Thorpe Title: President and Treasurer

[Signature Page to Note Purchase Agreement]

# **DRIVETRAIN AGENCY SERVICES, LLC,** as Administrative Agent

/s/ Timothy Daileader By: Name: Timothy Daileader Title: Authorized Signatory

[Signature Page to Note Purchase Agreement]

**PSC ECHO, LP,** as a Noteholder

By: /s/ Adam Fliss Name: Adam Fliss Title: General Counsel and Chief Compliance Officer

[Signature Page to Note Purchase Agreement]

<u>SCHEDULE I</u>

<u>EXHIBIT A</u>

EXHIBIT B

.

EXHIBIT C

.

<u>EXHIBIT D</u>

<u>EXHIBIT E</u>

<u>Annex I</u>

## Certain confidential information contained in this document, marked by [\*\*\*], has been omitted pursuant to Item 601(b)(10)(iv) of Regulation S-K because it is both (i) not material and (ii) the type of information that the registrant treats as private or confidential.

#### **INDEMNIFICATION AND ADVANCEMENT AGREEMENT**

This Indemnification and Advancement Agreement ("<u>Agreement</u>") is made as of \_\_\_\_\_\_, 2022 by and between Eargo, Inc., a Delaware corporation (the "<u>Company</u>"), and \_\_\_\_\_\_, a member of the Board of Directors of the Company ("<u>Indemnitee</u>"). This Agreement supersedes and replaces any and all previous Agreements between the Company and Indemnitee covering indemnification and advancement.

#### RECITALS

WHEREAS, the Board of Directors of the Company (the "<u>Board</u>") believes that highly competent persons have become more reluctant to serve publicly-held corporations as directors, officers, or in other capacities unless they are provided with adequate protection through insurance or adequate indemnification and advancement of expenses against inordinate risks of claims and actions against them arising out of their service to and activities on behalf of the corporation;

WHEREAS, the Board has determined that, in order to attract and retain qualified individuals, the Company will attempt to maintain on an ongoing basis, at its sole expense, liability insurance to protect persons serving the Company and its subsidiaries from certain liabilities. Although the furnishing of such insurance has been a customary and widespread practice among United States-based corporations and other business enterprises, the Company believes that, given current market conditions and trends, such insurance may be available to it in the future only at higher premiums and with more exclusions. At the same time, directors, officers, and other persons in service to corporations or business enterprises are being increasingly subjected to expensive and time-consuming litigation relating to, among other things, matters that traditionally would have been brought only against the Company or business enterprise itself. The Bylaws and Certificate of Incorporation of the Company require indemnification of the officers and directors of the Company. Indemnitee may also be entitled to indemnification pursuant to the General Corporation Law of the State of Delaware (the "<u>DGCL</u>"). The Bylaws, Certificate of Incorporation, and the DGCL expressly provide that the indemnification provisions set forth therein are not exclusive, and thereby contemplate that contracts may be entered into between the Company and members of the board of directors, officers and other persons with respect to indemnification and advancement of expenses;

WHEREAS, the uncertainties relating to such insurance, to indemnification, and to advancement of expenses may increase the difficulty of attracting and retaining such persons;

WHEREAS, the Board has determined that the increased difficulty in attracting and retaining such persons is detrimental to the best interests of the Company and its stockholders and that the Company should act to assure such persons that there will be increased certainty of such protection in the future;

WHEREAS, it is reasonable, prudent and necessary for the Company contractually to obligate itself to indemnify, and to advance expenses on behalf of, such persons to the fullest extent permitted by applicable law so that they will serve or continue to serve the Company free from undue concern that they will not be so indemnified;

WHEREAS, this Agreement is a supplement to and in furtherance of the Bylaws, Certificate of Incorporation and any resolutions adopted pursuant thereto, and is not a substitute therefor, nor diminishes or abrogates any rights of Indemnitee thereunder; and

WHEREAS, Indemnitee does not regard the protection available under the Bylaws, Certificate of Incorporation, DGCL and insurance as adequate in the present circumstances, and may not be willing to serve or continue to serve as an officer or director without adequate additional protection, and the Company desires Indemnitee to serve or continue to serve in such capacity. Indemnitee is willing to serve, continue to serve and to take on additional service for or on behalf of the Company on the condition that Indemnitee be so indemnified and be advanced expenses.

NOW, THEREFORE, in consideration of the premises and the covenants contained herein, the Company and Indemnitee do hereby covenant and agree as follows:

Section 1. <u>Services to the Company</u>. Indemnitee agrees to serve as a director of the Company. Indemnitee may at any time and for any reason resign from such position (subject to any other contractual obligation or any obligation imposed by operation of law).

Section 2. Definitions. As used in this Agreement:

(a) "Agent" means any person who is authorized by the Company or an Enterprise to act for or represent the interests of the Company or an Enterprise, respectively.

(b) A "Change in Control" occurs upon the earliest to occur after the date of this Agreement of any of the following events:

i. <u>Acquisition of Stock by Third Party</u>. Any Person (as defined below) is or becomes the Beneficial Owner (as defined below), directly or indirectly, of securities of the Company representing fifteen percent (15%) or more of the combined voting power of the Company's then outstanding securities unless the change in relative beneficial ownership of the Company's securities by any Person results solely from a reduction in the aggregate number of outstanding shares of securities entitled to vote generally in the election of directors;

ii. <u>Change in Board of Directors</u>. During any period of two (2) consecutive years (not including any period prior to the execution of this Agreement), individuals who at the beginning of such period constitute the Board, and any new director (other than a director designated by a person who has entered into an agreement with the Company to effect a transaction described in Sections 2(b)(i), 2(b)(iii) or 2(b)(iv)) whose election by the Board or nomination for election by the Company's stockholders was approved by a vote of at least two-thirds of the directors then still in office who either were directors at the beginning of the period or whose election or nomination for election was previously so approved, cease for any reason to constitute at least a majority of the members of the Board;

iii. <u>Corporate Transactions</u>. The effective date of a merger or consolidation of the Company with any other entity, other than a merger or consolidation which would result in the voting securities of the Company outstanding immediately prior to such merger or consolidation continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving entity) more than 51% of the combined voting power of the voting securities of the surviving entity outstanding immediately after such merger or consolidation and with the power to elect at least a majority of the board of directors or other governing body of such surviving entity;

iv. <u>Liquidation</u>. The approval by the stockholders of the Company of a complete liquidation of the Company or an agreement for the sale or disposition by the Company of all or substantially all of the Company's assets; and

v. <u>Other Events</u>. There occurs any other event of a nature that would be required to be reported in response to Item 6(e) of Schedule 14A of Regulation 14A (or a response to any similar item on any similar schedule or form) promulgated under the Exchange Act (as defined below), whether or not the Company is then subject to such reporting requirement.

vi. For purposes of this Section 2(b), the following terms have the following meanings:

- 1) "Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time.
- 2) "<u>Person</u>" has the meaning as set forth in Sections 13(d) and 14(d) of the Exchange Act; provided, however, that Person excludes (i) the Company, (ii) any trustee or other fiduciary holding securities under an employee benefit plan of the Company, and (iii) any corporation owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their ownership of stock of the Company.
- 3) "<u>Beneficial Owner</u>" has the meaning given to such term in Rule 13d-3 under the Exchange Act; provided, however, that Beneficial Owner excludes any Person otherwise becoming a Beneficial Owner by reason of the stockholders of the Company approving a merger of the Company with another entity.

(c) "Corporate Status" describes the status of a person who is or was acting as a director, officer, employee, fiduciary, or Agent of the Company or an Enterprise.

(d) "Disinterested Director" means a director of the Company who is not and was not a party to the Proceeding in respect of which indemnification is sought by Indemnitee.

(e) "<u>Enterprise</u>" means any other corporation, limited liability company, partnership, joint venture, trust, employee benefit plan or other entity for which Indemnitee is or was serving at the request of the Company as a director, officer, employee, or Agent.

(f) "Expenses" includes all reasonable attorneys' fees, retainers, court costs, transcript costs, fees of experts and other professionals, witness fees, travel expenses, duplicating costs, printing and binding costs, telephone charges, postage, delivery service fees, any federal, state, local or foreign taxes imposed on Indemnitee as a result of the actual or deemed receipt of any payments under this Agreement, ERISA excise taxes and penalties, and all other disbursements or expenses of the types customarily incurred in connection with prosecuting, defending, preparing to prosecute or defend, investigating, being or preparing to be a witness in, or otherwise participating in, a Proceeding. Expenses also include (i) Expenses incurred in connection with any appeal resulting from any Proceeding, including without limitation the premium, security for, and other costs relating to any cost bond, supersedeas bond, or other appeal bond or its equivalent, and (ii) Expenses incurred by Indemnitee in connection with the interpretation, enforcement or defense of Indemnitee's rights under this Agreement, by litigation or otherwise. The parties agree that for the purposes of any advancement of Expenses for which Indemnitee has made written demand to the Company in accordance with this Agreement, all Expenses included in such demand that are certified by affidavit of Indemnitee's counsel as being reasonable in the good faith judgment of such counsel will be presumed conclusively to be reasonable. Expenses, however, do not include amounts paid in settlement by Indemnitee or the amount of judgments or fines against Indemnitee.

(g) "<u>Independent Counsel</u>" means a law firm, or a member of a law firm, that is experienced in matters of corporation law and neither presently is, nor in the past five years has been, retained to represent: (i) the Company or Indemnitee in any matter material to either such party (other than with respect to matters concerning the Indemnitee under this Agreement, or of other indemnitees under similar indemnification agreements), or (ii) any other party to the Proceeding giving rise to a claim for indemnification hereunder. Notwithstanding the foregoing, the term "Independent Counsel" does not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in representing either the Company or Indemnitee in an action to determine Indemnitee's rights under this Agreement.

(h) "<u>Potential Change in Control</u>" means the occurrence of any of the following events: (i) the Company enters into any written or oral agreement, undertaking or arrangement, the consummation of which would result in the occurrence of a Change in Control; (ii) any Person or the Company publicly announces an intention to take or consider taking actions which if consummated would constitute a Change in Control; (iii) any Person who becomes the Beneficial Owner, directly or indirectly, of securities of the Company representing 5% or more of the combined voting power of the Company's then outstanding securities entitled to vote generally in the election of directors increases its beneficial ownership of such securities by 5% or more over the percentage so owned by such Person on the date hereof; or (iv) the Board adopts a resolution to the effect that, for purposes of this Agreement, a Potential Change in Control has occurred.

(i) The term "<u>Proceeding</u>" includes any threatened, pending or completed action, suit, claim, counterclaim, cross claim, arbitration, mediation, alternate dispute resolution mechanism, investigation, inquiry, administrative hearing or any other actual, threatened or completed proceeding, whether brought in the right of the Company or otherwise and whether of a civil, criminal, administrative, legislative, or investigative (formal or informal) nature, including any appeal therefrom, in which Indemnitee was, is or will be involved as a party, potential party, non-party witness or otherwise by reason of Indemnitee's Corporate Status or by reason of any action taken by Indemnitee (or a failure to take action by Indemnitee) or of any action (or failure to act) on Indemnitee's part while acting pursuant to Indemnitee's Corporate Status, in each case whether or not serving in such capacity at the time any liability or Expense is incurred for which indemnification, reimbursement, or advancement of Expenses can be provided under this Agreement. A Proceeding also includes a situation the Indemnitee believes in good faith may lead to or culminate in the institution of a Proceeding.

Section 3. Indemnity in Third-Party Proceedings. The Company will indemnify Indemnitee in accordance with the provisions of this Section 3 if Indemnitee is, or is threatened to be made, a party to or a participant in any Proceeding, other than a Proceeding by or in the right of the Company to procure a judgment in its favor. Pursuant to this Section 3, the Company will indemnify Indemnitee to the fullest extent permitted by applicable law against all Expenses, judgments, fines and amounts paid in settlement (including all interest, assessments and other charges paid or payable in connection with or in respect of such Expenses, judgments, fines and amounts paid in settlement) actually and reasonably incurred by Indemnitee or on Indemnitee's behalf in connection with such Proceeding or any claim, issue or matter therein, if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in or not opposed to the best interests of the Company and, in the case of a criminal Proceeding had no reasonable cause to believe that Indemnitee's conduct was unlawful.

Section 4. <u>Indemnity in Proceedings by or in the Right of the Company.</u> The Company will indemnify Indemnitee in accordance with the provisions of this Section 4 if Indemnitee is, or is threatened to be made, a party to or a participant in any Proceeding by or in the right of the Company to procure a judgment in its favor. Pursuant to this Section 4, the Company will indemnify Indemnitee to the fullest extent permitted by applicable law against all Expenses actually and reasonably incurred by Indemnitee or on Indemnitee's behalf in connection with such Proceeding or any claim, issue or matter therein, if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in or not opposed to the best interests of the Company. The Company will not indemnify Indemnitee for Expenses under this Section 4 related to any claim, issue or matter in a Proceeding for which Indemnitee has been finally adjudged by a court to be liable to the Company, unless, and only to the extent that, the Delaware Court or any court in which the Proceeding was brought determines upon application by Indemnitee that, despite the adjudication of liability but in view of all the circumstances of the case, Indemnitee is fairly and reasonably entitled to indemnification.

Section 5. Indemnification for Expenses of a Party Who is Wholly or Partly Successful. Notwithstanding any other provisions of this Agreement, to the fullest extent permitted by applicable law, the Company will indemnify Indemnitee against all Expenses actually and reasonably incurred by Indemnitee in connection with any Proceeding the extent that Indemnitee is successful, on the merits or otherwise. If Indemnitee is not wholly successful in such Proceeding but is successful, on the merits or otherwise, as to one or more but less than all claims, issues or matters in such Proceeding, the Company will indemnify Indemnitee against all Expenses actually and reasonably incurred by Indemnitee's behalf in connection with or related to each successfully resolved claim, issue or matter to the fullest extent permitted by law. For purposes of this Section 5 and without limitation, the termination of any claim, issue or matter in such a Proceeding by dismissal, with or without prejudice, will be deemed to be a successful result as to such claim, issue or matter. In the event that any Proceeding to which Indemnitee is a party is resolved in any manner other than by adverse judgment against Indemnitee (including, without limitation, settlement of such action, suit or proceeding with or without payment of money or other consideration) it shall be presumed that Indemnitee has been successful on the merits or otherwise in such action, suit or proceeding. Anyone seeking to overcome this presumption shall have the burden of proof and the burden of persuasion by clear and convincing evidence.

Section 6. <u>Indemnification For Expenses of a Witness</u>. Notwithstanding any other provision of this Agreement and to the fullest extent permitted by applicable law, the Company will indemnify Indemnitee against all Expenses actually and reasonably incurred by Indemnitee or on Indemnitee's behalf in connection with any Proceeding to which Indemnitee is not a party but to which Indemnitee is a witness, deponent, interviewee, or otherwise asked to participate.

Section 7. <u>Partial Indemnification</u>. If Indemnitee is entitled under any provision of this Agreement to indemnification by the Company for some or a portion of Expenses, but not, however, for the total amount thereof, the Company will indemnify Indemnitee for the portion thereof to which Indemnitee is entitled.

Section 8. <u>Additional Indemnification</u>. Notwithstanding any limitation in Sections 3, 4, or 5, the Company will indemnify Indemnitee to the fullest extent permitted by applicable law (including but not limited to, the DGCL and any amendments to or replacements of the DGCL adopted after the date of this Agreement that expand the Company's ability to indemnify its officers and directors) if Indemnitee is a party to or threatened to be made a party to any Proceeding (including a Proceeding by or in the right of the Company to procure a judgment in its favor).

Section 9. <u>Exclusions</u>. Notwithstanding any provision in this Agreement, the Company is not obligated under this Agreement to make any indemnification payment to Indemnitee in connection with any Proceeding:

(a) for which payment has actually been made to or on behalf of Indemnitee under any insurance policy or other indemnity provision, except to the extent provided in Section 16(b) of the Exchange Act and except with respect to any excess beyond the amount paid under any insurance policy or other indemnity provision; or

(b) for (i) an accounting of profits made from the purchase and sale (or sale and purchase) by Indemnitee of securities of the Company within the meaning of Section 16(b) of the Exchange Act or similar provisions of state statutory law or common law, (ii) any reimbursement of the Company by the Indemnitee of any bonus or other incentive-based or equity-based compensation or of any profits realized by the Indemnitee from the sale of securities of the Company pursuant to Section 304 of the Sarbanes-Oxley Act of 2002 (the "<u>Sarbanes-Oxley Act</u>"), or the payment to the Company of profits arising from the purchase and sale by Indemnitee of securities in violation of Section 306 of the Sarbanes-Oxley Act) or (iii) any reimbursement of the Company by Indemnitee of any compensation pursuant to any compensation recoupment or clawback policy adopted by the Board or the compensation committee of the Board, including but not limited to any such policy adopted to comply with stock exchange listing requirements implementing Section 10D of the Exchange Act; or

(c) initiated by Indemnitee, including any Proceeding (or any part of any Proceeding) initiated by Indemnitee against the Company or its directors, officers, employees or other indemnitees, unless (i) the Proceeding or part of any Proceeding is to enforce Indemnitee's rights to indemnification or advancement, of Expenses, including a Proceeding (or any part of any Proceeding) initiated pursuant to Section 14 of this Agreement, (ii) the Board authorized the Proceeding (or any part of any Proceeding) prior to its initiation or (iii) the Company provides the indemnification, in its sole discretion, pursuant to the powers vested in the Company under applicable law.

#### Section 10. Advances of Expenses.

(a) The Company will advance, to the extent not prohibited by law, the Expenses incurred by Indemnitee in connection with any Proceeding (or any part of any Proceeding) not initiated by Indemnitee or any Proceeding (or any part of any Proceeding) initiated by Indemnitee if (i) the Proceeding or part of any Proceeding is to enforce Indemnitee's rights to obtain indemnification or advancement of Expenses from the Company or Enterprise, including a proceeding initiated pursuant to Section 14 or (ii) the Board authorized the Proceeding (or any part of any Proceeding) prior to its initiation. The Company will advance the Expenses within thirty (30) days after the receipt by the Company of a statement or statements requesting such advances from time to time, whether prior to or after final disposition of any Proceeding.

(b) Advances will be unsecured and interest free. Indemnitee undertakes to repay the amounts advanced (without interest) to the extent that it is ultimately determined that Indemnitee is not entitled to be indemnified by the Company, but that Indemnitee qualifies for advances upon the execution of this Agreement and delivery to the Company. No other form of undertaking is required other than the execution of this Agreement. The Company will make advances without regard to Indemnitee's ability to repay the Expenses and without regard to Indemnitee's ultimate entitlement to indemnification under the other provisions of this Agreement.

(c) To the extent requested by Indemnitee and approved by the Board, the Company may at any time and from time to time provide security to Indemnitee for the Company's obligations hereunder through an irrevocable bank line of credit, funded trust or other collateral. Any such security, once provided to Indemnitee, may not be revoked or released without the prior written consent of the Indemnitee.

(d) The Company shall not seek from a court, or agree to, a "bar order" which would have the effect of prohibiting or limiting the Indemnitee's rights to receive advancement of Expenses under this Agreement.

#### Section 11. Procedure for Notification of Claim for Indemnification or Advancement.

(a) Indemnitee will notify the Company in writing of any Proceeding with respect to which Indemnitee intends to seek indemnification or advancement of Expenses hereunder as soon as reasonably practicable following the receipt by Indemnitee of written notice thereof. Indemnitee will include in the written notification to the Company a description of the nature of the Proceeding and the facts underlying the Proceeding and provide such documentation and information as is reasonably available to Indemnitee and is reasonably necessary to determine

whether and to what extent Indemnitee is entitled to indemnification following the final disposition of such Proceeding. Indemnitee's failure to notify the Company will not relieve the Company from any obligation it may have to Indemnitee under this Agreement, and any delay in so notifying the Company will not constitute a waiver by Indemnitee of any rights under this Agreement. The Secretary of the Company will, promptly upon receipt of such a request for indemnification, advise the Board in writing that Indemnitee has requested indemnification or advancement.

(b) The Company will be entitled to participate in (but not control the defense with respect to) the Proceeding at its own expense.

#### Section 12. Procedure Upon Application for Indemnification.

(a) Unless a Change of Control has occurred, the determination of Indemnitee's entitlement to indemnification will be made:

i. by a majority vote of the Disinterested Directors, even though less than a quorum of the Board;

ii. by a committee of Disinterested Directors designated by a majority vote of the Disinterested Directors, even though less than a quorum of the Board;

iii. if there are no such Disinterested Directors or, if such Disinterested Directors so direct, by written opinion provided by Independent Counsel selected by the Board; or

iv. if so directed by the Board, by the stockholders of the Company.

(b) If a Change in Control has occurred, the determination of Indemnitee's entitlement to indemnification will be made by written opinion provided by Independent Counsel selected by Indemnitee (unless Indemnitee requests such selection be made by the Board).

(c) The party selecting Independent Counsel pursuant to subsection (a)(iii) or (b) of this Section 12 will provide written notice of the selection to the other party. The notified party may, within ten (10) days after receiving written notice of the selection of Independent Counsel, deliver to the selecting party a written objection to such selection; provided, however, that such objection may be asserted only on the ground that the Independent Counsel so selected does not meet the requirements of "Independent Counsel" as defined in Section 2 of this Agreement, and the objection will set forth with particularity the factual basis of such assertion. Absent a proper and timely objection, the person so selected will act as Independent Counsel. If such written objection is so made and substantiated, the Independent Counsel so selected may not serve as Independent Counsel unless and until such objection is withdrawn or the Delaware Court has determined that such objection is without merit. If, within thirty (30) days after the later of submission by Indemnitee of a written request for indemnification pursuant to Section 11(a) hereof and the final disposition of the Proceeding, Independent Counsel has not been resolved, either the Company or Indemnitee may petition the Delaware Court for the appointment as Independent Counsel of a person selected by such court or by such other person as such court designates. Upon the due commencement of any judicial proceeding or arbitration pursuant to Section 14(a) of this Agreement, Independent Counsel will be discharged and relieved of any further responsibility in such capacity (subject to the applicable standards of professional conduct then prevailing).

(d) Indemnitee will cooperate with the person, persons or entity making the determination with respect to Indemnitee's entitlement to indemnification, including providing to such person, persons or entity upon reasonable advance request any documentation or information which is not privileged or otherwise protected from disclosure and which is reasonably available to Indemnitee and reasonably necessary to such determination. The Company will advance and pay any Expenses incurred by Indemnitee in so cooperating with the person, persons or entity making the indemnification determination irrespective of the determination as to Indemnitee's entitlement to indemnification and the Company hereby indemnifies and agrees to hold Indemnitee harmless therefrom. The Company promptly will advise Indemnitee in writing of the determination that Indemnitee is or is not entitled to indemnification, including a description of any reason or basis for which indemnification has been denied and providing a copy of any written opinion provided to the Board by Independent Counsel.

(e) If it is determined that Indemnitee is entitled to indemnification, the Company will make payment to Indemnitee within ten (10) days after such determination.

#### Section 13. Presumptions and Effect of Certain Proceedings.

(a) In making a determination with respect to entitlement to indemnification hereunder, the person or persons or entity making such determination will, to the fullest extent not prohibited by law, presume Indemnitee is entitled to indemnification under this Agreement if Indemnitee has submitted a request for indemnification in accordance with Section 11(a) of this Agreement, and the Company will, to the fullest extent not prohibited by law, have the burden of proof to overcome that presumption. Neither the failure of the Company (including by its directors or Independent Counsel) to have made a determination prior to the commencement of any action pursuant to this Agreement that indemnification is proper in the circumstances because Indemnitee has met the applicable standard of conduct, nor an actual determination by the Company (including by its directors or Independent Counsel) that Indemnitee has not met such applicable standard of conduct, will be a defense to the action or create a presumption that Indemnitee has not met the applicable standard of conduct.

(b) If the determination of the Indemnitee's entitlement to indemnification has not made pursuant to Section 12 within sixty (60) days after the latter of (i) receipt by the Company of Indemnitee's request for indemnification pursuant to Section 11(a) and (ii) the final disposition of the Proceeding for which Indemnitee requested Indemnification (the "Determination Period"), the requisite determination of entitlement to indemnification will, to the fullest extent not prohibited by law, be deemed to have been made and Indemnitee will be entitled to such indemnification, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law. The Determination Period may be extended for a reasonable time, not to exceed an additional thirty (30) days, if the person, persons or entity making the determination with respect to entitlement to indemnification in good faith requires such additional time for the obtaining or evaluating of documentation and/or information relating thereto; and provided, further, the Determination Period may be extended an additional fifteen (15) days if the determination of entitlement to indemnification is to be made by the stockholders pursuant to Section 12(a)(iv) of this Agreement; provided, that the Board will promptly cause a special meeting of the stockholders to be convened and held for the purpose of making such determination.

(c) The termination of any Proceeding or of any claim, issue or matter therein, by judgment, order, settlement or conviction, or upon a plea of <u>nolo contendere</u> or its equivalent, will not (except as otherwise expressly provided in this Agreement) of itself adversely affect the right of Indemnitee to indemnification or create a presumption that Indemnitee did not act in good faith and in a manner which Indemnitee reasonably believed to be in or not opposed to the best interests of the Company or, with respect to any criminal Proceeding, that Indemnitee had reasonable cause to believe that Indemnitee's conduct was unlawful.

(d) For purposes of any determination of good faith, Indemnitee will be deemed to have acted in good faith if Indemnitee acted based on the records or books of account of the Company, its subsidiaries, or an Enterprise, including financial statements, or on information supplied to Indemnitee by the directors or officers of the Company, its subsidiaries, or an Enterprise in the course of their duties, or on the advice of legal counsel for the Company, its subsidiaries, or an Enterprise in the course of their duties, or on the advice of legal counsel for the Company, its subsidiaries, or an Enterprise or or or the course of the Company or an Enterprise by an independent certified public accountant or by an appraiser, financial advisor or other expert selected with reasonable care by or on behalf of the Company, its subsidiaries, or an Enterprise. Further, Indemnitee will be deemed to have acted in a manner "not opposed to the best interests of the Company," as referred to in this Agreement if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in the best interests of the participants and beneficiaries of an employee benefit plan. The provisions of this Section 13(d) is not exclusive and does not limit in any way the other circumstances in which the Indemnitee may be deemed to have met the applicable standard of conduct set forth in this Agreement.

(e) The knowledge and/or actions, or failure to act, of any director, officer, trustee, partner, managing member, fiduciary, agent or employee of the Enterprise may not be imputed to Indemnitee for purposes of determining Indemnitee's right to indemnification under this Agreement.

#### Section 14. Remedies of Indemnitee.

(a) Indemnitee may commence litigation against the Company in the Delaware Court to obtain indemnification or advancement of Expenses provided by this Agreement in the event that (i) a determination is made pursuant to Section 12 of this Agreement that Indemnitee is not entitled to indemnification under this Agreement, (ii) the Company does not advance Expenses pursuant to this Agreement, (iii) the determination of entitlement to indemnification is not made pursuant to Section 12 of this Agreement within the Determination Period, (iv) the Company does not indemnify Indemnitee pursuant to Section 5 or 6 or the second to last sentence of Section 12(d) of this Agreement within ten (10) days after receipt by the Company of a written request therefor, (v) the Company does not indemnify Indemnitee pursuant to Section 3, 4, 7, or 8 of this Agreement within ten (10) days after a determination has been made that Indemnitee is entitled to indemnification, or (vi) in the event that the Company or any other person takes or threatens to take any action to declare this Agreement void or unenforceable, or institutes any litigation or other

action or Proceeding designed to deny, or to recover from, the Indemnitee the benefits provided or intended to be provided to the Indemnitee hereunder. Alternatively, Indemnitee, at Indemnitee's option, may seek an award in arbitration to be conducted by a single arbitrator pursuant to the Commercial Arbitration Rules of the American Arbitration Association. Indemnitee must commence such Proceeding seeking an adjudication or an award in arbitration within 180 days following the date on which Indemnitee first has the right to commence such Proceeding pursuant to this Section 14(a); <u>provided</u>, <u>however</u>, that the foregoing clause does not apply in respect of a Proceeding brought by Indemnitee to enforce Indemnitee's rights under Section 5 of this Agreement. The Company will not oppose Indemnitee's right to seek any such adjudication or award in arbitration.

(b) If a determination is made pursuant to Section 12 of this Agreement that Indemnitee is not entitled to indemnification, any judicial proceeding or arbitration commenced pursuant to this Section 14 will be conducted in all respects as a *de novo* trial, or arbitration, on the merits and Indemnitee may not be prejudiced by reason of that adverse determination. In any judicial proceeding or arbitration commenced pursuant to this Section 14 the Company will have the burden of proving Indemnitee is not entitled to indemnification or advancement of Expenses, as the case may be and will not introduce evidence of the determination made pursuant to Section 12 of this Agreement.

(c) If a determination is made pursuant to Section 12 of this Agreement that Indemnitee is entitled to indemnification, the Company will be bound by such determination in any judicial proceeding or arbitration commenced pursuant to this Section 14, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law.

(d) The Company is, to the fullest extent not prohibited by law, precluded from asserting in any judicial proceeding or arbitration commenced pursuant to this Section 14 that the procedures and presumptions of this Agreement are not valid, binding and enforceable and will stipulate in any such court or before any such arbitrator that the Company is bound by all the provisions of this Agreement.

(e) It is the intent of the Company that, to the fullest extent permitted by law, the Indemnitee not be required to incur legal fees or other Expenses associated with the interpretation, enforcement or defense of Indemnitee's rights under this Agreement by litigation or otherwise because the cost and expense thereof would substantially detract from the benefits intended to be extended to the Indemnitee hereunder. The Company, to the fullest extent permitted by law, will (within ten (10) days after receipt by the Company of a written request therefor) advance to Indemnitee such Expenses which are incurred by Indemnitee in connection with any action concerning this Agreement, Indemnitee's right to indemnification or advancement of Expenses from the Company, or concerning any directors' and officers' liability insurance policies maintained by the Company, and will indemnify Indemnitee against any and all such Expenses unless the court determines that each of the Indemnitee's claims in such Proceeding were made in bad faith or were frivolous or are prohibited by law. Without limitation of the foregoing, in the event that Indemnitee seeks a judicial adjudication of his or her rights under, or to recover under any directors' and officers' liability insurance policies, the Company shall pay on the Indemnitee's behalf, in advance, any and all expenses (of the types described in the definition of Expenses) actually and reasonably incurred by Indemnitee in such judicial adjudication, regardless of whether Indemnitee ultimately is determined to be entitled to such indemnification, advancement of expenses or insurance recovery.

#### Section 15. Establishment of Trust.

(a) In the event of a Potential Change in Control or a Change in Control, the Company will, upon written request by Indemnitee, create a trust for the benefit of Indemnitee (the "<u>Trust</u>") and from time to time upon written request of Indemnitee will fund such Trust in an amount sufficient to satisfy the reasonably anticipated indemnification and advancement obligations of the Company to the Indemnitee in connection with any Proceeding for which Indemnitee has demanded indemnification and/or advancement prior to the Potential Change in Control or Change in Control (the "<u>Funding</u> <u>Obligation</u>"). The trustee of the Trust (the "<u>Trustee</u>") will be a bank or trust company or other individual or entity chosen by the Indemnitee and reasonably acceptable to the Company. Nothing in this Section 15 relieves the Company of any of its obligations under this Agreement.

(b) The amount or amounts to be deposited in the Trust pursuant to the Funding Obligation will be determined by mutual agreement of the Indemnitee and the Company or, if the Company and the Indemnitee are unable to reach such an agreement, by Independent Counsel selected in accordance with Section 12(b) of this Agreement. The terms of the Trust will provide that, except upon the consent of both the Indemnitee and the Company, upon a Change in Control: (i) the Trust may not be revoked, or the principal thereof invaded, without the written consent of the Indemnitee; (ii) the Trustee will advance Expenses, to the fullest extent permitted by applicable law, within two (2) business days of a request by the Indemnitee; (iii) the Company will continue to fund the Trust in accordance with the Funding Obligation; (iv) the Trustee will promptly pay to the Indemnitee all amounts for which the Indemnitee is entitled to indemnification pursuant to this Agreement or otherwise; and (v) all unexpended funds in such Trust revert to the Company upon mutual agreement by the Indemnitee and the Company or, if the Indemnitee and the Company upon mutual agreement by the Indemnitee and the Company or, if the Indemnitee has been fully indemnified under the terms of this Agreement. New York law (without regard to its conflicts of laws rules) governs the Trust and the Trustee will consent to the exclusive jurisdiction of Delaware Court, in accordance with Section 25 of this Agreement.

#### Section 16. Non-exclusivity; Survival of Rights; Insurance; Subrogation.

(a) The indemnification and advancement of Expenses provided by this Agreement are not exclusive of any other rights to which Indemnitee may at any time be entitled under applicable law, the Certificate of Incorporation, the Bylaws, any agreement, a vote of stockholders or a resolution of directors, or otherwise. The indemnification and advancement of Expenses provided by this Agreement may not be limited or restricted by any amendment, alteration or repeal of this Agreement in any way with respect to any action taken or omitted by Indemnitee in Indemnitee's Corporate Status occurring prior to any amendment, alteration or repeal of this Agreement. To the extent that a change in Delaware law, whether by statute or judicial decision, permits greater indemnification or advancement of Expenses than would be

afforded currently under the Bylaws, Certificate of Incorporation, or this Agreement, it is the intent of the parties hereto that Indemnitee enjoy by this Agreement the greater benefits so afforded by such change. No right or remedy herein conferred is intended to be exclusive of any other right or remedy, and every other right and remedy is cumulative and in addition to every other right and remedy given hereunder or now or hereafter existing at law or in equity or otherwise. The assertion or employment of any right or remedy hereunder, or otherwise, will not prevent the concurrent assertion or employment of any other right or remedy.

(b) The Company hereby acknowledges that Indemnitee may have certain rights to indemnification, advancement of Expenses and/or insurance provided by one or more Persons with whom or which Indemnitee may be associated (including, without limitation, Patient Square Capital, LP and certain of its affiliates, collectively, the "Fund Indemnitors").

i. The Company hereby acknowledges and agrees:

1) the Company is the indemnitor of first resort with respect to any request for indemnification or advancement of Expenses made pursuant to this Agreement concerning any Proceeding arising from or related to Indemnitee's Corporate Status with the Company;

2) the Company is primarily liable for all indemnification and indemnification or advancement of Expenses obligations for any Proceeding arising from or related to Indemnitee's Corporate Status, whether created by law, organizational or constituent documents, contract (including this Agreement) or otherwise;

3) any obligation of any other Persons with whom or which Indemnitee may be associated (including, without limitation, any Fund Indemnitor) to indemnify Indemnitee and/or advance Expenses to Indemnitee in respect of any proceeding are secondary to the obligations of the Company's obligations;

4) the Company will indemnify Indemnitee and advance Expenses to Indemnitee hereunder to the fullest extent provided herein without regard to any rights Indemnitee may have against any other Person with whom or which Indemnitee may be associated (including, any Fund Indemnitor) or insurer of any such Person; and

ii. the Company irrevocably waives, relinquishes and releases (A) any other Person with whom or which Indemnitee may be associated (including, without limitation, any Fund Indemnitor) from any claim of contribution, subrogation, reimbursement, exoneration or indemnification, or any other recovery of any kind in respect of amounts paid by the Company to Indemnitee pursuant to this Agreement and (B) any right to participate in any claim or remedy of Indemnitee against any Fund Indemnitor (or former Fund Indemnitor), whether or not such claim, remedy or right arises in equity or under contract, statute or common law, including, without limitation, the right to take or receive from any Fund Indemnitor (or former Fund Indemnitor), directly or indirectly, in cash or other property or by set-off or in any other manner, payment or security on account of such claim, remedy or right.

iii. In the event any other Person with whom or which Indemnitee may be associated (including, without limitation, any Fund Indemnitor) or their insurers advances or extinguishes any liability or loss for Indemnitee, the payor has a right of subrogation against the Company or its insurers for all amounts so paid which would otherwise be payable by the Company or its insurers under this Agreement. In no event will payment by any other Person with whom or which Indemnitee may be associated (including, without limitation, any Fund Indemnitor) or their insurers affect the obligations of the Company hereunder or shift primary liability for the Company's obligation to indemnify or advance of Expenses to any other Person with whom or which Indemnitee may be associated (including, without limitation, any Fund Indemnitor).

iv. Any indemnification or advancement of Expenses provided by any other Person with whom or which Indemnitee may be associated (including, without limitation, any Fund Indemnitor) is specifically in excess over the Company's obligation to indemnify and advance Expenses or any valid and collectible insurance (including but not limited to any malpractice insurance or professional errors and omissions insurance) provided by the Company.

v. The Company and Indemnitee agree that the Fund Indemnitors are express third party beneficiaries of the terms of this Section 16(b).

(c) To the extent that the Company maintains an insurance policy or policies providing liability insurance for directors, officers, employees, or agents of the Enterprise, the Company will obtain a policy or policies covering Indemnitee to the maximum extent of the coverage available for any such director, officer, employee or agent under such policy or policies, including coverage in the event the Company does not or cannot, for any reason, indemnify or advance Expenses to Indemnitee as required by this Agreement. If, at the time of the receipt of a notice of a claim pursuant to this Agreement, the Company has director and officer liability insurance in effect, the Company will give prompt notice of such claim or of the commencement of a Proceeding, as the case may be, to the insurers in accordance with the procedures set forth in the respective policies. The Company will thereafter take all necessary or desirable action to cause such insurers to pay, on behalf of the Indemnitee, all amounts payable as a result of such Proceeding in accordance with the terms of such policies. Indemnitee agrees to assist the Company efforts to cause the insurers to pay such amounts and will comply with the terms of such policies, including selection of approved panel counsel, if required.

(d) The Company's obligation to indemnify or advance Expenses hereunder to Indemnitee for any Proceeding concerning Indemnitee's Corporate Status with an Enterprise will be reduced by any amount Indemnitee has actually received as indemnification or advancement of Expenses from such Enterprise. The Company and Indemnitee intend that any such Enterprise (and its insurers) be the indemnitor of first resort with respect to indemnification and advancement of Expenses for any Proceeding related to or arising from Indemnitee's Corporate Status with such Enterprise. The Company's obligation to indemnify and advance Expenses to Indemnitee is secondary to the obligations the Enterprise or its insurers owe to Indemnitee. Indemnitee agrees to take all reasonably necessary and desirable action to obtain from an Enterprise indemnification and advancement of Expenses for any Proceeding related to or arising from Indemnification and advancement of Expenses for any Proceeding related to obtain from an Enterprise indemnification and advancement of Expenses for any Proceeding related to obtain from an Enterprise indemnification and advancement of Expenses for any Proceeding related to or arising from Indemnification and advancement of Expenses for any Proceeding related to or arising from Indemnification and advancement of Expenses for any Proceeding related to or arising from Indemnification and advancement of Expenses for any Proceeding related to or arising from Indemnitie's Corporate Status with such Enterprise.

(e) In the event of any payment made by the Company under this Agreement, the Company will be subrogated to the extent of such payment to all of the rights of recovery of Indemnitee from any Enterprise or insurance carrier. Indemnitee will execute all papers required and take all action necessary to secure such rights, including execution of such documents as are necessary to enable the Company to bring suit to enforce such rights.

Section 17. <u>Duration of Agreement.</u> This Agreement continues until and terminates upon the later of: (a) ten (10) years after the date that Indemnitee ceases to serve as a director of the Company or (b) one (1) year after the final termination of any Proceeding then pending in respect of which Indemnitee is granted rights of indemnification or advancement of Expenses hereunder and of any Proceeding commenced by Indemnitee pursuant to Section 14 of this Agreement relating thereto. The indemnification and advancement of Expenses rights provided by or granted pursuant to this Agreement are binding upon and be enforceable by the parties hereto and their respective successors and assigns (including any direct or indirect successor by purchase, merger, consolidation or otherwise to all or substantially all of the business or assets of the Company), continue as to an Indemnitee who has ceased to be a director, officer, employee or agent of the Company or of any other Enterprise, and inure to the benefit of Indemnitee and Indemnitee's spouse, assigns, heirs, devisees, executors and administrators and other legal representatives.

Section 18. <u>Severability</u>. If any provision or provisions of this Agreement is held to be invalid, illegal or unenforceable for any reason whatsoever: (a) the validity, legality and enforceability of the remaining provisions of this Agreement (including without limitation, each portion of any Section of this Agreement containing any such provision held to be invalid, illegal or unenforceable, that is not itself invalid, illegal or unenforceable) will not in any way be affected or impaired thereby and remain enforceable to the fullest extent permitted by law; (b) such provision or provisions will be deemed reformed to the extent necessary to conform to applicable law and to give the maximum effect to the intent of the parties hereto; and (c) to the fullest extent possible, the provisions of this Agreement (including, without limitation, each portion of any Section of this Agreement containing any such provision held to be invalid, illegal or unenforceable) will be construed so as to give effect to the intent manifested thereby.

Section 19. <u>Interpretation</u>. Any ambiguity in the terms of this Agreement will be resolved in favor of Indemnitee and in a manner to provide the maximum indemnification and advancement of Expenses permitted by law. The Company and Indemnitee intend that this Agreement provide to the fullest extent permitted by law for indemnification in excess of that expressly provided, without limitation, by the Certificate of Incorporation, the Bylaws, vote of the Company stockholders or disinterested directors, or applicable law.

#### Section 20. Enforcement.

(a) The Company expressly confirms and agrees that it has entered into this Agreement and assumed the obligations imposed on it hereby in order to induce Indemnitee to serve as a director or officer of the Company, and the Company acknowledges that Indemnitee is relying upon this Agreement in serving or continuing to serve as a director or officer of the Company.

(b) This Agreement constitutes the entire agreement between the parties hereto with respect to the subject matter hereof and supersedes all prior agreements and understandings, oral, written and implied, between the parties hereto with respect to the subject matter hereof; provided, however, that this Agreement is a supplement to and in furtherance of the Certificate of Incorporation, the Bylaws and applicable law, and is not a substitute therefor, nor to diminish or abrogate any rights of Indemnitee thereunder.

Section 21. <u>Modification and Waiver</u>. No supplement, modification or amendment of this Agreement is binding unless executed in writing by the parties hereto. No waiver of any of the provisions of this Agreement will be deemed or constitutes a waiver of any other provisions of this Agreement nor will any waiver constitute a continuing waiver.

Section 22. <u>Notice by Indemnitee.</u> Indemnitee agrees promptly to notify the Company in writing upon being served with any summons, citation, subpoena, complaint, indictment, information or other document relating to any Proceeding or matter which may be subject to indemnification or advancement of Expenses covered hereunder. The failure of Indemnitee to so notify the Company does not relieve the Company of any obligation which it may have to the Indemnitee under this Agreement or otherwise.

Section 23. <u>Notices.</u> All notices, requests, demands and other communications under this Agreement will be in writing and will be deemed to have been duly given if (a) delivered by hand to the other party, (b) sent by reputable overnight courier to the other party or (c) sent by facsimile transmission or electronic mail, with receipt of oral confirmation that such communication has been received:

(a) If to Indemnitee, at the address indicated on the signature page of this Agreement, or such other address as Indemnitee provides to the Company.

(b) If to the Company to:

Name:	Eargo, Inc.
Address:	1600 Technology Drive, 6th Floor
	San Jose, CA 95110
Email:	[***]
Attention:	Chief Legal Officer

or to any other address as may have been furnished to Indemnitee by the Company.

#### Section 24. Contribution.

(a) Whether or not the indemnification provided for herein is available, in respect of any threatened, pending or completed action, suit or proceeding in which the Company is jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), the Company shall pay, in the first instance, the entire amount of any judgment or settlement of such action, suit or proceeding without requiring Indemnitee to contribute to such payment and the Company hereby waives and relinquishes any right of contribution it may have against Indemnitee.

(b) Without diminishing or impairing the obligations of the Company set forth in the preceding subparagraph, if, for any reason, Indemnitee shall elect or be required to pay all or any portion of any judgment or settlement in any threatened, pending or completed action, suit or proceeding in which the Company is jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), the Company shall contribute to the amount of Expenses, judgments, fines and amounts paid in settlement actually and reasonably incurred and paid or payable by Indemnitee in proportion to the relative benefits received by the Company and all officers, directors or employees of the Company, other than Indemnitee, who are jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), on the one hand, and Indemnitee, on the other hand, from the transaction or events from which such action, suit or proceeding arose; <u>provided</u>, <u>however</u>, that the proportion determined on the basis of relative benefit may, to the extent necessary to conform to law, be further adjusted by reference to the relative fault of the Company and all officers, directors or employees of the Company other than Indemnitee who are jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), on the one hand, and Indemnitee, on the other hand, in connection with the transaction or events that resulted in such expenses, judgments, fines or settlement amounts, as well as any other equitable considerations which applicable law may require to be considered. The relative fault of the Company and all officers, directors or employees of the Company, other than Indemnitee, who are jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), on the one hand, and Indemnitee, who are jointly liable with Indemnitee (or would be if joined in such action, suit or proceeding), on the one hand, and Indemnitee, on the other hand, shall be determined by reference to, among other t

(c) The Company hereby agrees to fully indemnify and hold Indemnitee harmless from any claims of contribution which may be brought by officers, directors, or employees of the Company, other than Indemnitee, who may be jointly liable with Indemnitee.

(d) To the fullest extent permissible under applicable law, if the indemnification provided for in this Agreement is unavailable to Indemnitee for any reason whatsoever, the Company, in lieu of indemnifying Indemnitee, will contribute to the amount incurred by Indemnitee, whether for judgments, fines, penalties, excise taxes, amounts paid or to be paid in settlement and/or for Expenses, in connection with any claim relating to an indemnifiable event under this Agreement, in such proportion as is deemed fair and reasonable in light of all of the circumstances of such Proceeding in order to reflect (i) the relative benefits received by the Company and Indemnitee as a result of the event(s) and/or transaction(s) giving cause to such Proceeding; and/or (ii) the relative fault of the Company (and its directors, officers, employees and agents) and Indemnitee in connection with such event(s) and/or transaction(s).

Section 25. <u>Applicable Law and Consent to Jurisdiction</u>. This Agreement and the legal relations among the parties are governed by, and construed and enforced in accordance with, the laws of the State of Delaware, without regard to its conflict of laws rules. Except with respect to any arbitration commenced by Indemnitee pursuant to Section 14(a) of this Agreement, the Company and Indemnitee hereby irrevocably and unconditionally (i) agree that any action or Proceeding arising out of or in connection with this Agreement may be brought only in the Chancery Court of the State of Delaware and any state appellate court therefrom within the State of Delaware (or if the Chancery Court of the State of Delaware declines to accept jurisdiction over a particular matter, any federal court sitting in the State of Delaware and any federal appellate court therefrom) (the "Delaware Court") and not in any other state or federal court in the United States of America or any court in any other country, (ii) consent to submit to the exclusive jurisdiction of the Delaware Court for purposes of any action or Proceeding arising out of or in connection with this Agreement, with this Agreement, (iii) waive any objection to the laying of venue of any such action or Proceeding in the Delaware Court, and (iv) waive, and agree not to plead or to make, any claim that any such action or Proceeding brought in the Delaware Court has been brought in an improper or inconvenient forum.

Section 26. <u>Identical Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which will for all purposes be deemed to be an original but all of which together constitutes one and the same Agreement. Only one such counterpart signed by the party against whom enforceability is sought needs to be produced to evidence the existence of this Agreement.

Section 27. <u>Headings</u>. The headings of this Agreement are inserted for convenience only and do not constitute part of this Agreement or affect the construction thereof.

IN WITNESS WHEREOF, the parties have caused this Agreement to be signed as of the day and year first above written.

EARGO,	INC.
--------	------

#### INDEMNITEE

By: \_\_\_\_\_\_Name: \_\_\_\_\_\_Office:

Name: Address:

nuur

# Certain confidential information contained in this document, marked by [\*\*\*], has been omitted pursuant to Item 601(b)(10)(iv) of Regulation S-K because it is both (i) not material and (ii) the type of information that the registrant treats as private or confidential.

#### **BOARD OBSERVER AGREEMENT**

THIS BOARD OBSERVER AGREEMENT, dated as of June 24, 2022 (this "<u>Agreement</u>"), is made by and between Eargo, Inc., a Delaware corporation (the "<u>Company</u>"), and PSC Echo, LP (the "<u>Investor</u>").

WHEREAS, the Company desires to provide the Investor with the right to nominate a board observer (the "<u>Observer</u>") to the Company's board of directors (the "<u>Board</u>") to attend all meetings of the Board and any and all committees and sub-committees thereof, including without limitation, any executive committee thereof, but excluding any committee established to review a transaction between the Company, on the one hand, and the Investor Parties, the Observer or an Affiliate of the Investor Parties (as such terms are defined in the Investor Rights Agreement (as defined below)) or the Observer on the other hand (collectively, the "<u>Committees</u>");

**NOW, THEREFORE**, in consideration of the foregoing, the Company and the Investor hereby agree as follows, effective as of the First Tranche Closing Date (as defined the Purchase Agreement (as defined below)):

#### 1. Board Observer Rights.

(a) Pursuant to Section 2.1(a) of the Investor Rights Agreement by and between the Company and the Investor dated as of June 24, 2022 (as amended, amended and restated, supplemented or otherwise modified from time to time, the "Investor Rights Agreement"), the Company shall cause one person designated by the Investor to be appointed as a non-voting observer to the Board and the Committees. On the date hereof, the Investor designates Justin Sabet-Peyman as the Observer. Except as set forth herein, the Company shall invite Observer to attend, in a non-voting observer capacity, all meetings of the Board and any and all Committees (as determined by Observer from time to time). Except as set forth herein, Observer shall have the right to participate in any such meeting, but in no event shall Observer: (i) be deemed to be a member of the Board or any Committee; (ii) have the right to vote on any matter under consideration by the Board or any Committee or otherwise have any power to cause the Company to take, or not to take, any action; (iii) have or be deemed to have, or otherwise be subject to, any duties, fiduciary or otherwise, to the Company or its stockholders or any duties, fiduciary or otherwise, otherwise applicable to the directors of the Company; or (iv) have any right or possess any authority to bind the Company or any of its subsidiaries in any way whatsoever. As a non-voting observer, and except as set forth herein, Observer will also be provided, concurrently with delivery to the directors of the Company and in the same manner delivery is made to them, copies of all notices, agenda, minutes, consents, and all other materials or information, financial or otherwise, that are provided to the directors, including with respect to any Board or Committee meeting or any written consent in lieu of a Board or Committee meeting. If a meeting of the Board or a Committee is conducted via telephone or other electronic medium, except as set forth herein, Observer may attend such meeting via the same medium. The presence of Observer shall not be taken into account or required for purposes of establishing a quorum. Notwithstanding anything in this Agreement to the contrary, the Board or any Committee may exclude the Observer from all or any portion of any meeting or discussions, and can withhold or redact materials otherwise distributed to the members of the Board or such Committee, solely to the extent that (i) the Board reasonably determines based on the advice of counsel that such action is reasonably necessary to prevent the loss of attorney-client privilege,

work product or similar privilege; <u>provided</u> that, in such case, (x) any such exclusion shall only apply to such portion of such meeting or material which would be required to preserve such privilege and not to any other portion thereof; and (y) Observer shall be informed of the general subject matter of the materials or meetings (or portion thereof) from which Observer is excluded to the extent it would not result in the loss of such privilege, or (ii) the Board reasonably determines in good faith that the information being discussed at all or a portion of such meetings or included in such materials, (A) would give rise to a conflict of interest or otherwise relates to any actual or potential transactions between or involving the Company or its affiliates, on the one hand, and the Investor Parties, the Observer or an Affiliate of the Investor Parties or the Observer, on the other hand, (B) would result in a breach of the Company's confidentiality obligations to any third party or (C) would result in a violation of applicable laws; <u>provided</u> that, in any such case, (x) any such exclusion shall only apply to such portion of such meeting or material which would be required to avoid such conflict, breach of confidentiality or violation of applicable laws, and not to any other portion thereof; and (y) Observer shall be informed of the general subject matter of the materials or meetings (or portion thereof) from which Observer is excluded to the extent it would not result in a potential breach of such confidentiality obligations to any other portion thereof; and (y) Observer shall be informed of the general subject matter of the materials or meetings (or portion thereof) from which Observer is excluded to the extent it would not result in a potential breach of such confidentiality obligations to any other portion thereof; and (y) Observer shall be informed of the general subject matter of the materials or meetings (or portion thereof) from which Observer is excluded to the extent it would not result in a potential

(b) The rights described in this <u>Section 1</u> shall terminate with respect to such Observer upon: (i) such time as the Investor no longer has the right to appoint an Observer pursuant to Section 2.1(a) of the Investor Rights Agreement; (ii) written notice of termination to the Company from the Investor; (iii) any material violation of the terms of this Agreement by Investor which (A) remains uncured within ten business days after receipt of notice thereof (which notice must be delivered to the Investor by the Company in accordance with <u>Section 11</u>), or (B) if such violation is not subject to cure, may reasonably lead to material harm to the Company; or (iv) the death or disability of Observer, *provided that* upon the occurrence of any event described in the foregoing clause (iv), the Investor shall have the right to appoint a replacement Observer and such replacement Observer shall have the rights described in this <u>Section 1</u>. For the avoidance of doubt, the provisions of this <u>Section 1(b)</u> (including any potential termination of rights thereunder) shall not affect, or be construed as affecting, the rights or obligations of the Investor or the Company under the Investor Rights Agreement or the Note Purchase Agreement by and among the Company, the Investor and DriveTrain dated as of June 24, 2022 (as amended, amended and restated, supplemented or otherwise modified from time to time, the "<u>Purchase Agreement</u>").

#### 2. Confidential Treatment of Company Information.

(a) In consideration of the Company's disclosure to the Investor, including through the Observer as one of the Investor's Representatives (defined below), of information that is not publicly available concerning the Company, the Investor agrees that this Agreement will apply to all Board and Committee materials and all other non-public, confidential and/or proprietary information, in any form whatsoever, disclosed or made available by the Company or its advisors to the Observer or to the Investor via Observer, in his or her capacity as the Observer ("Information"). The Investor may disclose the Information to its affiliates, and its affiliates' officers, employees, directors, managers, agents, advisors, affiliates, limited partners, prospective limited partners and representatives ("Representatives"; which term shall, for the avoidance of doubt, include the Observer) who either (x) are bound by a professional, ethical, contractual or other obligation of confidentiality to the Investor or the Company no less restrictive than the terms of this Section 2 or (y) have agreed to act in accordance with the terms and conditions of this Section 2.

(b) Except as otherwise provided herein, the Investor agrees: (i) to hold Information in strict confidence and (ii) not to disclose Information to any third parties. The Investor agrees to instruct all such Representatives, with whom it shares information, that such Information is confidential and not to disclose such Information to third parties without the prior written permission of the Company or as otherwise permitted herein. The Investor shall be responsible for any non-compliance with this <u>Section 2</u> by its Representatives, including the Observer.

3. Exempted Disclosure. The term Information and the foregoing restriction on the disclosure of Information will not include information which: (i) was, is, or hereafter becomes generally known or available to the public other than as the result of the Investor's or its Representatives' disclosure in violation of this Agreement; (ii) was or is hereafter acquired by the Investor or its Representatives from a source other than the Company without, to the Investor's knowledge, restriction as to disclosure; (iii) was independently developed by or for the Investor or its Representatives without use of or reference to the Information or Investor acquired such information from the Company in connection with the Investor or its affiliates being a lender to the Company or its affiliate, including pursuant to the Purchase Agreement (subject to the confidentiality provisions thereof); (iv) is subsequently and expressly granted permission to disclose by the Company in writing (with email being sufficient); or (v) is required or requested to be disclosed by the Investor or any of its Representatives pursuant to law, regulation, judicial, regulatory or administrative process or court order, including, without limitation, through or in connection with any legal proceeding by subpoena or discovery request or otherwise, subject to all applicable defenses; provided, that to the extent permitted by law, rule or regulation and reasonably practicable under the circumstances, the Investor shall give the Company reasonably prompt notice of such required disclosure so that the Company may, at its sole cost and expense, seek a protective order or other appropriate remedy and/or waive compliance with the provisions of this Agreement; provided that no such notice shall be required for disclosure in connection with an audit or to any governmental agency, regulatory authority or self-regulatory authority (including, without limitation, bank and securities examiners) having or claiming to have authority to regulate or oversee the Investor.

4. <u>Return of Information</u>. Following the termination of the rights of the Observer described in <u>Section 1</u> and the termination of any other rights of the Investor to appoint members to the Board pursuant to Section 2.1 of the Investor Rights Agreement and upon written request of the Company, the Investor and its Representatives will promptly: (i) destroy all physical materials containing or consisting of Information and all hard copies thereof; and (ii) to the extent technically practicable, destroy all electronically stored Information in the possession or control of the Investor or its Representatives. Notwithstanding the foregoing, the Investor and its Representatives may retain in its confidential files copies of any item of Information in order to comply with any legal, compliance, regulatory or document retention requirements, any internal compliance policy, or in order to satisfy audit requirements. The Investor and its Representatives may also retain and need not destroy any Information which is retained pursuant to automated backup or archiving procedures, and that portion of any Information that consists of copies, electronic copies, notes, analyses, compilations, studies, analyses,

compilations, studies or other documents prepared by or for the benefit of the Investor or its Representatives to the extent the foregoing contains Information, will remain subject to the restriction on nondisclosure set forth in this Agreement for eighteen months after Observer's termination pursuant to <u>Section 1(b)</u> of this Agreement.

5. <u>Company Ownership of Information; Remedies</u>. Neither the Investor nor any of its affiliates, or their Representatives shall, by virtue of the Company's disclosure of, or such person's use of any Information, acquire any rights with respect thereto, all of which rights (including intellectual property rights) shall remain exclusively with the Company. The Investor acknowledges that all of the Information is owned solely by the Company (or its licensors) and that the unauthorized disclosure or use of such Information may cause irreparable harm and significant injury, the degree of which may be difficult to ascertain. The Company, on the one hand, and the Investor on the other hand, each acknowledge that monetary damages would not be a sufficient remedy for any breach or threatened breach of this Agreement and that, in the event of any breach of this Agreement by any party hereto, the non-breaching party(ies) is entitled to seek all forms of equitable relief (including an injunction and order for specific performance), in addition to all other remedies available at law or in equity, without the requirement that a bond be posted by the non-breaching party(ies). Such remedies will not be exclusive but will be in addition to other remedies that may be available to the non-breaching party(ies) at law or in equity.

6. <u>Compliance with Securities Laws</u>. The Investor agrees that the Information is given in confidence in accordance with the terms of this Agreement, and the Investor acknowledges that any trading in securities while in possession of material non-public information (within the meaning of Regulation FD under the Securities Exchange Act of 1934, as amended, "<u>MNPI</u>") or communicating such information to any other person who trades in such securities could subject the Investor to liability under applicable securities laws and the rules and regulations promulgated thereunder, and agrees to instruct all of its Representatives to whom it discloses Information that such Information may contain or constitute MNPI and that any trading in securities while in possession of MNPI or communicating such information to any other person who trades in such securities could subject the Representative to liability under applicable securities and regulations promulgated thereunder. Notwithstanding anything herein to the contrary, nothing in this Agreement shall be construed as (i) making the Investor or the Observer an insider of the Company, (ii) creating any duties or restrictions on the Investor's or the Observer's ability to trade any debt or equity issued by the Company beyond those that may exist under applicable law, or (iii) creating any fiduciary relationship between the Observer, the Investor or either of their Representatives, on the one hand, and any other holder of securities issued by the Company, on the other hand.

7. <u>Information Barriers</u>. The Company (i) acknowledges that the Investor has advised it that the Investor and its affiliates are in the business of holding, investing and trading in securities, and (ii) so long as no information is disclosed in violation of <u>Sections 2</u> and <u>3</u>, agrees that this Agreement shall not in any way restrict or limit the activities of the Investor, any of its affiliates or any of their respective employees (or, for the avoidance of doubt, the Observer) in the business of holding, investing and trading in securities so long as such activities are in compliance with applicable securities laws, including, but not limited to, where (A) such activities are undertaken by the Investor, employees and/or their affiliates (or, for the avoidance of doubt, the Observer) that are not then in possession of Information constituting material non-public

information of the Company, (B) Observer is subject to information barriers designed to protect the flow of Information constituting material non-public information to the Investor, other employees and affiliates or (C) an affirmative defense pursuant to paragraph (c) of Rule 10b5-1 under the Securities Exchange Act of 1934, as amended, is otherwise available.

8. Indemnification; Advancement of Expenses. Observer shall be entitled to advancement of expenses and rights to indemnification from the Company, and the Company agrees to advance expenses to and indemnify the Observer, in each case to the same extent provided by the Company in the Indemnification and Advancement Agreement attached to the Investor Rights Agreement as Exhibit B. The foregoing rights to indemnification and advancement of expenses constitute third-party rights contractually extended to Observer by the Company and do not constitute rights to indemnification or advancement of expenses as a result of Observer serving as a director, manager, officer, employee, or agent of the Company. The Company shall reimburse Observer, on the same basis as directors of the Company, for all reasonable out-of-pocket expenses incurred by Observer in connection with attendance at Board and Committee meetings or any other matter which Observer expressly agrees thereto in his or her sole discretion). All reimbursements payable by the Company pursuant to this <u>Section 8</u> shall be paid to Observer in accordance with the Company's policies and practices with respect to director expense reimbursement then in effect; provided, however, that any such reimbursement shall be paid to Observer no later than comparable reimbursement is paid to the members of the Board. Observer shall not be entitled to receive any other payment or remuneration hereunder or otherwise in connection with Observer's involvement with the Board or the Committee.

9. <u>Insurance</u>. For the duration of Observer's appointment as Observer of the Company, and thereafter for the duration of the applicable statute of limitations, the Company shall cause to be maintained in effect a policy of liability insurance coverage for Observer against liability that may be asserted against or incurred by him or her in his or her capacity as Observer which is no worse in scope and amount to that provided to the members of the Board. Upon request, the Company shall provide Observer or his or her counsel with a copy of all directors'/officers' liability insurance applications, binders, policies, declarations, endorsements, and other related materials.

10. <u>Governing Law: Venue for Disputes; Waiver of Jury Trial</u>. This Agreement shall be governed by, and construed in accordance with, the laws of the state of New York, without giving effect to any choice of law or conflict of law rules or provisions (whether of the state of New York or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the state of New York. Each of the parties hereto consents to the exclusive jurisdiction and venue of any state or federal court located in New York in which appeal from the New York Court of Appeals may validly be taken under the laws of the State of New York and hereby waives, to the maximum extent permitted by law, any objection, including any objection based on forum non conveniens, to the bringing of any such proceeding in such jurisdictions. Each of the parties hereto irrevocably waives the right to trial by jury.

11. Notices. All communications hereunder shall be in writing and shall be delivered either by certified or registered mail, postage pre-paid, return receipt requested, email or nationally recognized overnight courier, and shall be addressed to the following addresses:

To the Company:	Eargo, Inc. 2665 North First Street, Suite 300 San Jose, CA 95134 Attention: Chief Legal Officer Email: [***]
<i>With a copy (which shall not constitute notice) to:</i>	Latham & Watkins LLP 10250 Constellation Blvd., Suite 1100 Los Angeles, CA 90067 Attention: David Zaheer, Charles Ruck, Kathleen Wells Email: [***]
To Investor:	c/o Patient Square Equity Advisors, LP 2884 Sand Hill Road, Suite 100 Menlo Park, CA 94025 E-mail: [***] Attention: Adam Fliss, Justin Sabet-Peyman
<i>With a copy (which shall not constitute notice to):</i>	Ropes & Gray LLP Three Embarcadero Center San Francisco, CA 94111 Email: [***] Attention: Jason Freedman and
	Ropes & Gray LLP Prudential Tower 800 Boylston Street Boston, MA 02199 E-mail: [***] Attention: Tom Fraser

12. <u>Entire Agreement</u>. This Agreement, together with the Investor Rights Agreement, constitutes the complete and exclusive statement regarding the subject matter of this Agreement and supersedes all prior agreements, understandings and communications, oral or written, between the parties regarding the subject matter of this Agreement.

13. <u>Term</u>. The provisions of <u>Section 1</u> hereof shall terminate and be of no further force or effect pursuant to <u>Section 1(b)</u> hereof. Notwithstanding the provisions of this <u>Section 13</u>, the provisions of <u>Sections 2, 3, 5, 6, 9, 11, 13, 14, 15, 16</u> and <u>17</u> shall terminate two (2) years following the termination of <u>Section 1</u> hereof and the provisions of <u>Sections 7, 8</u> and <u>10</u> shall survive indefinitely following the termination of this Agreement.

14. <u>Amendments</u>. No provision of this Agreement may be amended, modified, or waived, except in a writing signed by the Investor, the Company and, in accordance with <u>Section 19</u>, the Observer.

15. Assignment. This Agreement may not be assigned by any party hereto without the prior written consent of the other party.

16. <u>Severability</u>. The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provision, and if any restriction in this Agreement is found by a court to be unreasonable or unenforceable, then such court may amend or modify the restriction so it can be enforced to the fullest extent permitted by law.

17. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all of which taken together shall constitute one single agreement between the parties. A signed copy of this Agreement delivered by email (including via DocuSign or equivalent software) or other means of electronic transmission shall be deemed to have the same legal effect as delivery of an original signed copy of this Agreement. The exchange of copies of this Agreement and of signature pages by e-mail transmission (including via DocuSign or equivalent software) or other means of electronic transmission shall constitute effective execution and delivery of this Agreement as to the parties and may be used in lieu of the original Agreement for all purposes.

18. <u>Waiver</u>. Any waiver by any party hereto of a breach of any provision of this Agreement shall not operate or be construed to be a waiver of any other breach of such provision or of any breach of any other provision of this Agreement. The failure of a party to insist on strict adherence to any term of this Agreement on one or more occasions shall not be construed as a waiver or deprive such party of the right to thereafter insist on strict adherence to that term or any other term of this Agreement.

19. <u>No Third Party Beneficiary</u>. This Agreement will be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns. Nothing in this Agreement, express or implied, is intended to confer upon any person other than the parties hereto or their respective successors or permitted assigns any rights or remedies under or by reason of this Agreement; <u>provided</u>, <u>however</u>, that the Observer (and his successors, heirs and Representatives) is an intended third-party beneficiary of, and may enforce, this Agreement.

[signature page follows]

IN WITNESS WHEREOF, the undersigned have hereto executed this Agreement as of the date first above written.

### EARGO, INC.

By: /s/ Christian Gormsen

Name: Christian Gormsen Title: Chief Executive Officer and President

[Signature Page to Board Observer Agreement]

IN WITNESS WHEREOF, the undersigned has hereto executed this Agreement as of the date first above written.

## PSC ECHO, LP

By: /s/ Adam Fliss

Name: Adam Fliss Title: General Counsel and Chief Compliance Officer

[Signature Page to Board Observer Agreement]

Certain confidential information contained in this document, marked by [\*\*\*], has been omitted pursuant to Item 601(b)(10)(iv) of Regulation S-K because it is both (i) not material and (ii) the type of information that the registrant treats as private or confidential.

### EARGO, INC.

### INVESTOR RIGHTS AGREEMENT

June 24, 2022

SECTION 1 DEFINITIONS		Page 1
1.1	Certain Definitions	
1.1	Certain Definitions	1
SECTION	2 COVENANTS OF THE PARTIES	4
2.1	Board of Directors	4
2.2	Restrictions on Transfer	6
2.3	Restrictive Legends	7
2.4	Standstill	8
2.5	Confidentiality	9
2.6	Financial Statements and Other Information	10
2.7	Information Rights	11
2.8	Section 16 Matters	12
2.9	Voting Agreement	12
2.10	Nasdaq Listing	13
2.11	Information Barriers	13
SECTION	3 MISCELLANEOUS	14
3.1	Counterparts	14
3.2	Governing Law	14
3.3	Entire Agreement; No Third Party Beneficiary	15
3.4	Expenses	15
3.5	Notices	15
3.6	Successors and Assigns	16
3.7	Headings	16
3.8	Amendments and Waivers	16
3.9	Interpretation; Absence of Presumption	17
3.10	Severability	17
3.11	Specific Performance	17
3.12	Corporate Opportunities	18
3.13	Public Announcement	18
3.14	Non-Recourse Parties	18

## EXHIBITS

Exhibit A Exhibit B Board Observer Agreement Indemnification and Advancement Agreement

#### EARGO, INC.

#### **INVESTOR RIGHTS AGREEMENT**

This INVESTOR RIGHTS AGREEMENT dated as of June 24, 2022 (this "*Agreement*") is by and between Eargo, Inc., a Delaware corporation (the "*Company*"), and the investor(s) listed on the signature page hereto ("*Investor*").

#### RECITALS

The Company proposes to issue the Notes and, following conversion of the Notes, shares of the Company's Common Stock, to the Investor, in each case as contemplated by the Note Purchase Agreement (as may be amended from time to time, the "*Purchase Agreement*"), of even date herewith (the "*Financing*").

As a condition to the Financing, the parties have agreed to enter into this Agreement.

The parties therefore agree as follows:

#### SECTION 1 DEFINITIONS

1.1 Certain Definitions. As used in this Agreement, the following terms shall have the meanings set forth below:

(a) "Activist Investor" means any Person that (a) in the reasonable judgment of the Board of Directors, holds or will acquire and hold securities of the Company with the purpose or the effect of changing or influencing control of the Company, as such terms are used in Rule 13d-1 of the Exchange Act, (b) any Person identified on the most recently available "SharkWatch 50" list (or, if "SharkWatch 50" is no longer available, then the prevailing comparable list as reasonably determined by the Company), or (c) has been previously identified in writing to the Investor by the Company as a Person that is acting with the purpose or the effect of changing or influencing control of the Company or that within the three (3) years preceding the date of the proposed Transfer has acted with the purpose or the effect of changing or influencing control of the Company. An "Activist Investor" shall also include any other Person that, to the knowledge of the Investor, is an Affiliate of a Person referenced in any of the foregoing clauses (a) through (c), and shall exclude any Person with respect to which the Company has agreed in writing shall not be considered an "Activist Investor" (it being understood that a Person shall be considered not to be an Activist Investor if such Person shall have agreed in writing with the Company that such Person will acquire and hold securities of the Company in the ordinary course of business and not with the purpose or the effect of changing or influencing control of the Company that such Person will acquire and hold securities of the Company in the ordinary course of business and not with the purpose or the effect of changing or influencing control of the Company that such Person will acquire and hold securities of the Company in the ordinary course of business and not with the purpose or the effect of changing or influencing control of the Company).

such Person.

(b) "Affiliate" of any Person means any Person, directly or indirectly, Controlling, Controlled by or under common Control with

(c) "Board of Directors" means the Company's board of directors.

(d) "Business Day" means any day other than a Saturday, a Sunday or any day on which the Federal Reserve Banks of New York and San Francisco are authorized or required by law or executive order to close or be closed.

(e) "*Code*" means the Internal Revenue Code of 1986, as amended.

(f) "Company Competitor" means any Person that is primarily engaged in the business of selling hearing aids.

(g) "*Confidential Information*" means information regarding the Company or its Subsidiaries that is non-public, confidential or proprietary in nature, together with all analyses, compilations, forecasts, studies or other documents prepared by the Investor or its Representatives which contain or otherwise reflect such information. "Confidential Information" shall not include such portions of the Confidential Information that (a) are or become generally available to the public other than as a result of the Investors' or its Affiliates' or their respective Representatives' disclosure in violation of this Agreement, (b) become available to the Investors or its Affiliates or their respective Representatives on a non-confidential basis from a source other than the Company or its Subsidiaries without any known breach of any confidentiality obligations, (c) was already in the Investor's or its Affiliate's or their respective Representatives' possession prior to the date of this Agreement and which was not obtained from the Company or its Subsidiaries or their respective Representatives or (d) are independently developed by the Investor or its Affiliates without use of or reference to the Confidential Information.

(h) "*Control*" (including its correlative meanings "*under common Control with*" and "*Controlled by*") means, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through ownership of securities or partnership or other interests, by contract or otherwise.

(i) "*Conversion Shares*" means the shares of Common Stock issuable upon conversion of the Notes as contemplated by the Purchase Agreement.

(j) "Exchange Act" means the Securities Exchange Act of 1934, as amended.

(k) "*First Tranche Investment*" means the investment of \$100 million in senior secured notes of the Company by the Investor as contemplated by the Purchase Agreement.

(1) "*Governmental Entity*" means any supranational, national, state, municipal, local or foreign government, any court, tribunal, arbitrator, administrative agency, commission or other governmental official, authority or instrumentality.

(m) "*Indebtedness*" means, without duplication, all indebtedness for borrowed money, including such obligations evidenced by notes, bonds, debentures or similar instruments.

(n) "Investor Parties" means the Investor and its Permitted Transferees.

(o) "Notes" means the senior secured notes of the Company issued to the Investor pursuant to the Purchase Agreement

(p) "Other Holders" means any Person that holds Notes other than the Investor Parties.

(q) "*Person*" means an individual, corporation, partnership, limited liability company, joint venture, trust or unincorporated organization or a government or agency or political subdivision thereof.

(r) "Representatives" means a Persons' Affiliates, employees, agents, consultants, accountants, attorneys or financial advisors.

(s) "*Restricted Securities*" means Conversion Shares required to bear the legend set forth in <u>Section 2.3(a)</u> under the applicable provisions of the Securities Act.

(t) "Rights Offering" has the meaning set forth in the Purchase Agreement.

(u) "SEC" means the Securities and Exchange Commission.

(v) "Securities Act" means the Securities Act of 1933, as amended.

(w) "*Subsidiary*" means, when used with reference to a party, any corporation or other organization, whether incorporated or unincorporated, of which such party or any other Subsidiary of such party is a general partner or serves in a similar capacity, or, with respect to such corporation or other organization, at least a majority of the securities or other interests having by their terms ordinary voting power to elect a majority of the board of directors or others performing similar functions is directly or indirectly owned or controlled by such party or by any one or more of its Subsidiaries.

(x) "*Tax*" and "*Taxes*" means all federal, state, local and foreign taxes (including, without limitation, income, franchise, property, sales, withholding, payroll and employment taxes).

(y) "*Tax Return*" means any return, report or similar filing (including the attached schedules) filed or required to be filed with respect to Taxes (and any amendments thereto), including any information return, claim for refund or declaration of estimated Taxes.

(z) "*Transfer*" means any direct or indirect (a) sale, transfer, hypothecation, assignment, gift, bequest or disposition by any other means, whether for value or no value and whether voluntary or involuntary (including, without limitation, by realization upon any lien or by operation of law or by judgment, levy, attachment, garnishment, bankruptcy or other legal or equitable proceedings) or (b) grant of any option, warrant or other right to purchase or the entry into any hedge, swap or any other agreement or any transaction that transfers, in whole or in part, directly or indirectly, the economic consequence of ownership of Common Stock. The term "*Transferred*" shall have a correlative meaning.

The following terms are defined in the Sections of this Agreement indicated:

Term	Section
Agreement	Preamble
Applicable Matters	3.2(b)
Chosen Court	3.2(b)
Company	Preamble
Confidentiality Agreement	2.4(a)
Financing	Preamble
Financing Arrangement	2.2(a)
Investor	Preamble
Investor Group	3.12
Investor Nominee	2.1(b)
Issuer Agreement	2.2(a)
Non-Recourse Parties	3.14
Observer	2.1(a)
Other Investments	3.12
Permitted Transferees	2.2(a)
Purchase Agreement	Preamble
Renounced Business Opportunity	3.12

### SECTION 2 COVENANTS OF THE PARTIES

### 2.1 Board of Directors.

(a) Prior to consummation of a Rights Offering (as contemplated by the Purchase Agreement), for as long as the Investor Parties hold any Notes, the Investor Parties shall have the right to appoint one observer to the Board of Directors (the "*Observer*"); provided, that such Observer shall be a senior professional, no less senior than Vice President, associated with Patient Square Capital, L.P. The initial Observer shall be Justin Sabet-Peyman. The Company and the Investor shall enter into the Board Observer Agreement in the form and substance attached hereto as Exhibit A. Following the consummation of the Rights Offering, or in the event the Investor Parties no longer hold any Notes, all rights of the Investor Parties to designate an Observer and for any previously designated Observer to observe under this Agreement shall terminate without the requirement of further action by the Company or any other person or entity.

(b) Effective immediately following the Rights Offering and any acquisition of shares of Common Stock by the Investor Parties pursuant to the Note Purchase Agreement, the Investor Parties shall have the right to nominate the number of directors (each, an "*Investor Nominee*" and collectively, "*Investor Nominees*") to the Board of Directors that is proportionate to Investor's ownership of the Company, which will be rounded up to the nearest whole number (for the avoidance of doubt, such number of Investor Nominees shall in no event be less than one). If the Investor Parties have the right to so nominate an Investor Nominee or Investor Nominees, subject to <u>Section 2.1(d)</u>, the Company shall take commercially reasonable

actions to cause to be elected to the Board of Directors such individual or individuals nominated by the Investor, including, without limitation, by (i) expanding the size of the Board of Directors and causing the Investor Nominee or Investor Nominees to be appointed as directors and (ii) including such Investor Nominee or Investor Nominees on the Company's recommended slate of directors that are up for election at the Company's Annual Meeting of Stockholders in the Company's proxy statement to its stockholders announcing the Company's Annual Meeting of Stockholders and nominating the Investor Nominee or Investor Nominees for election to the Board of Directors. If, following election to the Board of Directors, an Investor Nominee resigns, is removed, is not re-elected or is otherwise unable to serve for any reason and the Investor Parties still have the right to nominate such Investor Nominee or Investor Nominees, then, subject to Section 2.1(e), the Investor Parties shall be entitled to designate a replacement Investor Nominee or Investor Nominees, and the Board of Directors shall use commercially reasonable efforts to elect such replacement Investor Nominee or Investor Nominees to the Board of Directors, including, without limitation, by including such Investor Nominee or Investor Nominees on the Company's recommended slate of directors that are up for election at the Company's Annual Meeting of Stockholders in the Company's proxy statement to its stockholders announcing the Company's Annual Meeting of Stockholders and nominating the Investor Nominees for election to the Board of Directors. In the event that the Investor Parties cease to hold any Conversion Shares that entitle it to nominate the Investor Nominee or Investor Nominees as provided above, if requested by the Board of Directors, the Investor Parties shall use reasonable best efforts to have its Investor Nominee or Investor Nominees resign as a director. As a pre-condition to the nomination of any Investor Nominee, such Investor Nominee shall execute and deliver to the Board of Directors an irrevocable letter of resignation to be deemed tendered at the time the Investor is required to use reasonable best efforts to have him or her so resign and the Board of Directors requests such resignation. If the Board of Directors so requests, the Observer shall resign at such time as an Investor Nominee or Investor Nominees are elected to the Board of Directors.

(c) The foregoing rights in <u>Section 2.1(a)</u> and <u>Section 2.1(b)</u> will inure to the Investor and will not be exercisable by any transferee that acquires Notes or Conversion Shares other than Permitted Transferees of the Investor.

(d) Notwithstanding anything to the contrary herein, as a condition to his or her nomination to the Board of Directors, each Investor Nominee must (i) be either (x) a senior professional, no less senior than Vice President, associated with Patient Square Capital, L.P. or (y) otherwise reasonably acceptable to the Nominating and Corporate Governance Committee of the Board of Directors applying the Company's standard practices and the same considerations to the Investor Nominee as would be applied by such committee to any other director appointee, nominee or applicant, (ii) comply with the Corporate Governance Guidelines of the Company as in effect from time to time to the extent such compliance is required from all of the other director appointees, nominees or applicants and (iii) satisfy all then-applicable independence and other requirements of membership on a Board of Directors, Nominating and Corporate Governance Committee of the Board of Directors and Compensation Committee of the Board of Directors (or similar committees) imposed by the rules of Nasdaq (other than the requirements of Rule 10A-3(b)(1)(ii) under the Exchange Act).

(e) Subject to all applicable legal and stock exchange listing requirements, at the request of the Investor Parties, the Company shall take all action reasonably necessary to ensure an Investor Nominee serves as a member of each committee of the Board of Directors (other than the Audit Committee and any committee established to review a transaction between the Company, on the one hand, and the Investor Parties, Investor Nominees or an Affiliate of the Investor Parties or the Investor Nominee, on the other hand).

(f) For the duration of Observer's appointment as Observer of the Company and for the duration of any Investor Nominee's service on the Board of Directors, and thereafter for the duration of the applicable statute of limitations, the Company shall cause to be maintained in effect a policy of liability insurance coverage for Observer and any Investor Nominee against liability that may be asserted against or incurred by him or her in his or her capacity as Observer or Investor Nominee, as applicable, which is no worse in scope and amount to that provided to the members of the Board of Directors. Upon request, the Company shall provide Observer and any Investor Nominee or his or her counsel with a copy of all directors'/officers' liability insurance applications, binders, policies, declarations, endorsements, and other related materials.

(g) The Company and any Investor Nominee elected to the Company's Board of Directors shall enter into the Indemnification and Advancement Agreement in the form and substance attached hereto as Exhibit B.

#### 2.2 Restrictions on Transfer.

(a) The Investor Parties shall not Transfer any of the Notes prior to the Maturity Date thereof to any Person without the consent of the Company, unless an Event of Default (as defined in the Purchase Agreement) has occurred and is continuing. Notwithstanding the foregoing, without the consent of the Company, the Investor may Transfer Notes (i) to its direct or indirect limited partners or other investors, in each case, to which Notes are transferred by way of distribution, the Investor's affiliated funds or controlled Affiliates (such controlled Affiliates, limited partners and other investors that hold Notes, whether during or after such restricted period, collectively, the "Permitted Transferees"), (ii) in connection with customary pledges in connection with financing arrangements by the Investor or to the extent that all of the net proceeds of such sale are solely used to satisfy a margin call (i.e. posted as collateral) or repay bona fide loan or lending transaction entered into by the Investor (a "Financing Arrangement") to the extent necessary to satisfy a bona fide margin call on such Financing Arrangement or to avoid a bona fide margin call on such Financing Arrangement that is reasonably likely to occur; provided, that, in each such Financing Arrangement, the lender has executed a customary issuer agreement (an "Issuer Agreement") with each lender or agent therefor with respect to such Financing Arrangements on commercially reasonable terms in form and substance reasonably satisfactory to the Company, and subject to the consent of the Company (which will not be unreasonably withheld, conditioned or delayed), or (iii) following the date the Company commences a voluntary case under Title 11 of the United States Bankruptcy Code or any other similar insolvency laws; provided, however, that as a condition to such Transfer any such transferee must agree to be bound by (x) the terms of this <u>Section 2</u> pursuant to a written agreement in form and substance reasonably satisfactory to the Company and (y) the terms and conditions of the Registration Rights Agreement (and upon such Transfer, whether during or after such restricted period, any transferee pursuant to clause (i) above shall become an "Investor Party" and any transferee pursuant to clauses (ii) and (iii) above shall become an "Other Holder," in each case, for purposes of this Section 2). For the avoidance of doubt, the foregoing restriction shall not apply to conversions of Notes, as contemplated by the Purchase Agreement.

(b) In connection with any Transfer of Notes that is effected pursuant to a transaction in which the Investor Parties (or any of their representatives) negotiate the terms of such Transfer directly with the third party purchaser (other that any underwriter, dealer (including a dealer acting as a block positioner), market maker, placement agent or initial purchaser thereof), at no time shall the Investor Parties and Other Holders, as applicable, knowingly (after reasonable inquiry) Transfer any Notes to (i) any Company Competitor, (ii) any Activist Investor or Person who the Investor Party or Other Holder, as applicable, reasonably should know is an Activist Investor or has been an Activist Investor in the last three (3) years. In no event shall the foregoing limitations apply to, or limit in any way, sales by the Investor Parties in transactions effected on any stock exchange.

(c) In any event, Restricted Securities shall not be Transferred except upon satisfaction of the conditions specified in <u>Section 2.3</u>, which conditions are intended to ensure compliance with the provisions of the Securities Act. Any attempted Transfer in violation of this <u>Section 2.2</u> or applicable securities laws shall be void *ab initio*.

### 2.3 Restrictive Legends.

(a) The Conversion Shares issued upon conversion of the Notes shall be represented by a certificate and shall (unless otherwise permitted by the provisions of <u>Section 2.3(c)</u>) be stamped or otherwise imprinted with a legend in substantially the following form (in addition to any legend required under applicable state securities laws):

"THE OFFER AND SALE OF THIS SECURITY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "<u>SECURITIES ACT</u>"), AND THIS SECURITY MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED EXCEPT (A) PURSUANT TO A REGISTRATION STATEMENT THAT IS EFFECTIVE UNDER THE SECURITIES ACT; OR (B) PURSUANT TO AN EXEMPTION FROM, OR IN A TRANSACTION NOT SUBJECT TO, THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT."

(b) The Investor Parties or Other Holders, as applicable, consent to the Company making a notation on its records and giving instructions to any transfer agent of the Conversion Shares in order to implement the restrictions on transfer set forth in this <u>Section 2.3</u>.

(c) Prior to any proposed Transfer of any Restricted Securities, the Investor Party and Other Holder shall give written notice to the Company of such Investor Party's or Other Holder's, as applicable, intention to effect such Transfer. Each such notice shall describe the manner and circumstances of the proposed Transfer in sufficient detail, and, unless there is in effect a registration statement under the Securities Act covering the proposed Transfer, shall be accompanied by either (i) an opinion of legal counsel reasonably satisfactory to the Company and

its transfer agent to the effect that the proposed Transfer of the Restricted Securities may be effected without registration under the Securities Act, or (ii) any other evidence reasonably satisfactory to counsel to the Company and its transfer agent, whereupon such Investor Party or Other Holder, as applicable, shall be entitled to Transfer such Restricted Securities in accordance with the terms of the notice delivered by such Investor Party or Other Holder, as applicable, to the Company. Each certificate evidencing the Restricted Securities transferred shall bear the appropriate restrictive legend set forth in <u>Section 2.3</u> above, except that such certificate shall not bear the first such restrictive legend if such legend is not required in order to establish compliance with any provisions of the Securities Act. Upon the request of an Investor Party or Other Holder, as applicable, of a certificate bearing the first such restrictive legend from such certificate and from the certificate to be issued to the applicable transfere if such legend is not required in order to establish compliance with any provisions of the Securities Act and the Investor promptly Transfers the Conversion Shares. If an Investor Party or Other Holder holds a certificate bearing the second restrictive legend, upon the request of such Investor Party or Other Holder, as applicable, the Company shall remove such restrictive legend from such certificate when the provisions of <u>Section 2.2</u> are no longer applicable to the applicable conversion Shares.

#### 2.4 Standstill.

(a) The Investor Parties and any Other Holders, as applicable, agree (x) so long as the Investor Parties or Other Holders have record and beneficial ownership (within the meaning of Rule 13d-3 under the Exchange Act) of Conversion Shares representing more than five percent (5%) of the outstanding Common Stock of the Company or (y) the Investor Parties have the right to nominate a director to the Board of Directors (or any person so nominated continues to serve on the Board of Directors) pursuant to Section 2.1, the Investor Parties and Other Holders will not at any time, nor will it cause or permit any of its Affiliates to: (a) effect or seek, offer or propose (whether publicly or otherwise) to effect, or announce any intention to effect or cause or participate in or in any way assist, facilitate or encourage any other Person to effect or seek, offer or propose (whether publicly or otherwise) to effect or participate in, (i) any acquisition of any securities (or beneficial ownership thereof), or rights or options to acquire any securities (or beneficial ownership thereof), or any assets, indebtedness or businesses of the Company or its Subsidiaries, other than as a result of the Rights Offering contemplated by the Purchase Agreement, (ii) any tender or exchange offer, merger, consolidation, recapitalization or other disposition involving the Company or its Subsidiaries or assets of the Company or its Subsidiaries constituting a significant portion of the consolidated assets of the Company and its Subsidiaries, or (iii) any "solicitation" of "proxies" (as such terms are used in the proxy rules of the SEC) or consents to vote any voting securities of the Company or any of its Affiliates; (b) form, join or in any way participate in a "group" (as defined under the Exchange Act) with respect to the Company or otherwise act in concert with any person in respect of any such securities; (c) otherwise act, alone or in concert with others, to seek representation on or to control or influence the management, Board of Directors or policies of the Company or to obtain representation on the Board of Directors except in accordance with Section 2.1; (d) take any action which would or would reasonably be expected to force the Company to make a public announcement regarding any of the types of matters set forth in clause (a) above; or (e) enter into any discussions or arrangements with any third party with respect to any of the foregoing; it being understood that nothing in this Section 2.4

shall (x) restrict or prohibit the Investor Nominee or Investor Nominees, as applicable, from taking any action, or refraining from taking any action, which he or she determines, in his or her reasonable discretion, is necessary to fulfill his or her fiduciary duties as a member of the Board of Directors or (y) restrict the Investor Parties' or Other Holders' acquisition of the Notes or Common Stock issuable upon conversion of the Notes in connection with the Rights Offering contemplated by the Purchase Agreement (including the accretion of dividends thereon and any dividends payable in any other security), in each case, in accordance with the terms of this Agreement and the Purchase Agreement; provided, further, that the restrictions set forth in clause (a)(i) above shall not apply (i) with respect to acquisitions of Common Stock consented to in writing by the Board of Directors, or (ii) following the public announcement of a tender or exchange offer, merger, consolidation, recapitalization or other disposition of all or substantially all of the Company's stock or assets, in each case, that has been approved by the Board of Directors, and the restrictions set forth in this Section 2.4 shall not apply (iii) following the date the Company commences proceedings under Title 11 of the United States Bankruptcy Code or any other similar insolvency laws. Upon closing of the First Tranche Investment contemplated by the Purchase Agreement, the restrictions set forth in Section 6 of that certain Confidentiality Agreement, dated as of March 7, 2022, by and between the Company and Patient Square Capital, L.P. (the "Confidentiality Agreement"), shall terminate and cease to apply. Notwithstanding the foregoing, it is understood and agreed that the Investor Parties shall not be prohibited from entering into an agreement and having confidential discussions with legal, accounting or financial advisors for the limited purposes of evaluating any of the transactions contemplated by this Section 2.4, and the Investor Parties and/or its Affiliates may initiate private discussions with the Company that could result in a negotiated transaction otherwise prohibited by this Section 2.4; provided, however, that, in each case, any such discussions under this sentence are not reasonably be expected to require public disclosure.

(b) The provisions in <u>Section 2.4(a)</u> will terminate and cease to have effect if, following the Rights Offering contemplated by the Purchase Agreement, the Investor Parties acquire control of a majority of the voting power of the Company.

#### 2.5 Confidentiality.

(a) So long as the Investor Parties have the right to appoint an Observer or nominate a director to the Board of Directors pursuant to <u>Section 2.1</u> or an Observer or Investor Nominee or Investor Nominees, as applicable, are still acting as an Observer or serving on the Board of Directors, as applicable, (i) the Investor Parties and Other Holders shall keep all Confidential Information confidential and shall not, without the Company's prior written consent, disclose any Confidential Information in any manner whatsoever, in whole or in part and (ii) the Investor Parties and Other Holders shall not use any Confidential Information, other than in connection with the performance of its obligations hereunder. During such period, the Investor Parties and Other Holders shall disclose the Confidential Information only to such of its respective Representatives who need to know the Confidential Information for such purpose, who are informed by the applicable Investor Party or Other Holder, as applicable, of the confidential nature of the Confidential Information and who either (x) are otherwise bound by an obligation of confidentiality to the Company to ensure compliance with the terms of this <u>Section 2.5</u> or (y) shall agree to act in accordance with the terms and conditions of this <u>Section 2.5</u>. The applicable Investor Party or Other Holder shall be responsible for any non-compliance with this <u>Section 2.5</u>.

by its Representatives. The applicable Investor Party or Other Holder shall return to the Company or, insofar as reasonably practicable, destroy all copies of Confidential Information upon the Company's written request. Notwithstanding the foregoing, the Investor Parties and Other Holders may retain such materials to the extent required by applicable law or in accordance with internal compliance procedures; *provided, however*, that the Investor Parties and Other Holders shall keep all such copies confidential in accordance with this <u>Section 2.5</u>. In addition, notwithstanding the foregoing, Confidential Information may be disclosed by the Investor Parties to their limited partners, prospective limited partners, members or other owners, who are subject to customary confidentiality obligations, as part of ordinary course reporting or review procedure or in connection with ordinary course fundraising, transactional or reporting activities, as the case may be. The Confidentiality Agreement shall terminate simultaneously with the closing of the First Tranche Investment.

(b) In the event that an Investor Party or any of its Representatives is required or requested by applicable law (including oral questions, interrogatories, requests for information or documents, subpoena, civil investigative demand or other process) to disclose any of the Confidential Information, the applicable Investor Party will provide the Company with prompt notice (unless such notification is prohibited by applicable law and other than in connection with a routine audit or examination by, or a blanket document request from, a regulatory or governmental entity that does not reference the Company or this Agreement) so that the Company may seek a protective order or other appropriate remedy and/or waive compliance with the provisions of this <u>Section 2.5</u>. In the event that such a protective order or other remedy is not obtained, that no such notice is required to be provided to the Company or that the Company waives compliance with the provisions of this <u>Section 2.5</u>, the applicable Investor Party may disclose such Confidential Information without liability hereunder.

### 2.6 Financial Statements and Other Information.

(a) Subject to <u>Section 2.6(b)</u>, and for so long as the Investor Parties hold record and beneficial ownership (within the meaning of Rule 13d-3 under the Exchange Act) of Notes or of Conversion Shares representing more than five percent (5%) of the outstanding Common Stock of the Company, the Company shall deliver to the Investor Parties:

(i) within 90 days after the end of each fiscal year of the Company commencing with the fiscal year ending December 31, 2022, (A) an audited, consolidated balance sheet of the Company and its Subsidiaries as of the end of such fiscal year, (B) an audited, consolidated income statement of the Company and its Subsidiaries for such fiscal year and (C) an audited, consolidated statement of cash flows of the Company and its Subsidiaries for such fiscal year;

(ii) Within 60 days after the last day of each fiscal year of the Company, (A) annual operating budgets (including income statements, balance sheets and cash flow statements, by month) for the upcoming fiscal year of the Company and (B) annual financial projections for the following fiscal year (on a quarterly basis) as approved by the board of directors of the Company, together with any related business forecasts used in the preparation of such annual financial projections, in each case, in form and substance reasonably acceptable to the Investor Parties and approved in accordance with this Agreement;

(iii) within 45 days after the end of each quarter commencing with the quarter ending June 30, 2022, (A) an unaudited, consolidated balance sheet of the Company and its Subsidiaries as of the end of such fiscal quarter, (B) an unaudited, consolidated income statement of the Company and its Subsidiaries for such fiscal quarter and (C) an unaudited, consolidated statement of cash flows of the Company and its Subsidiaries for such fiscal quarter and (C) an unaudited, consolidated statement of cash flows of the Company and its Subsidiaries for such fiscal quarter and (C) an unaudited, consolidated statement of cash flows of the Company and its Subsidiaries for such fiscal quarter and (C) and the company and its Subsidiaries for such fiscal quarter and (C) and the company and its Subsidiaries for such fiscal quarter and (C) and the company and its Subsidiaries for such fiscal quarter and (C) and the company and its Subsidiaries for such fiscal quarter and (C) and the company and its Subsidiaries for such fiscal quarter and (C) and the company and its Subsidiaries for such fiscal quarter and (C) and the company and its Subsidiaries for such fiscal quarter and (C) and the company and its Subsidiaries for such fiscal quarter and (C) and the company a

(iv) within 30 days after the last day of each month, commencing with the month ending May 31, 2022, a monthly management report of the Company and its Subsidiaries covering operations for such month, in such form prepared by the Company's management team consistent with its past practice and reasonably acceptable to the Investor (it being understood that the sample management report provided by the Company to the Investor on June 10, 2022 is reasonably acceptable to the Investor).

(b) Notwithstanding the foregoing, financial statements and other reports required to be delivered pursuant to this <u>Section 2.6</u> filed by the Company with the SEC and available on EDGAR (or such other free, publicly-accessible internet database that may be established and maintained by the SEC as a substitute for or successor to EDGAR) shall be deemed to have been delivered to the Investor on the date on which the Company posts such documents to EDGAR (or such other free, publicly-accessible internet database that may be established and maintained by the SEC as a substitute for or successor to EDGAR); *provided, however*, that in each case, the Company shall use its commercially reasonable efforts to notify the Investor in writing (which may be by electronic mail) of the posting of any such documents.

(c) For so long as the Investor Parties hold record and beneficial ownership (within the meaning of Rule 13d-3 under the Exchange Act) of Notes or of Conversion Shares representing more than five percent (5%) of the outstanding Common Stock of the Company (on a fully-diluted basis), the Investor Parties or the employees of the Investor Parties shall have the reasonable right to consult from time to time with the officers of the Company at its principal place of business regarding operating and financial matters of the Company; *provided*, *that*, the exercise of such right does not materially interfere with the operations of the business of the Company.

**2.7 Information Rights**. So long as Investor Parties have record and beneficial ownership (within the meaning of Rule 13d-3 under the Exchange Act) of Notes or of Conversion Shares representing more than five percent (5%) of the outstanding Common Stock of the Company (on a fully-diluted basis), the Investor shall receive a copy of all written board packages delivered to the Board of Directors in connection with meetings thereof and shall have access to all monthly financial statements of the Company; *provided, that*, the Investor shall only distribute any such materials to Investor Parties. Notwithstanding the foregoing, the Company shall not be required to afford any such information if doing so would (a) cause a violation of any agreement to which the Company or any of its Subsidiaries is a party, (b) cause a risk of a loss of privilege to such party or any of its Subsidiaries, (c) constitute a violation of any applicable law, rule or regulation or (d) otherwise disclose competitively sensitive material.

#### 2.8 Section 16 Matters.

(a) The Company shall use commercially reasonable efforts in the form of resolutions adopted by the Board in its discretion as may be required to cause the receipt of Conversion Shares pursuant to the Purchase Agreement by each Purchaser Party (including, with respect to the Purchaser and its Affiliates, as "directors by deputization") that may from time to time be subject to the reporting requirements of Section 16(a) of the Exchange Act with respect to the Company, to be exempt under Rule 16b-3 promulgated under the Exchange Act, to the extent such exemption is available to such Persons.

(b) If the Company becomes a party to a consolidation, merger or other similar transaction that may result in the Purchaser Parties, their Affiliates and/or an Investor Nominee being deemed to have made a disposition of equity securities of the Company or derivatives thereof for purposes of Section 16 of the Exchange Act, and if an Investor Nominee is serving on the Board at such time or has served on the Board during the preceding six months, then (i) the Board will pre-approve such disposition of equity securities or derivatives thereof for the express purpose of exempting the Purchaser Parties', their Affiliates' and such Investor Nominee's interests (with respect to the Purchaser and its Affiliates, as "directors by deputization") in such transaction from Section 16(b) of the Exchange Act pursuant to Rule 16b-3 thereunder and (ii) if the transaction involves (A) a merger or consolidation to which the Company is a party and capital stock is, in whole or in part, converted into or exchanged for equity securities of a different issuer, (B) a potential acquisition by the Purchaser Parties, their Affiliates and/or an Investor Nominee of equity securities of such other issuer or derivatives thereof and (C) an Affiliate or other designee of the Purchaser Parties or their Affiliates will serve on the board of directors (or its equivalent) of such other issuer, then if the Purchaser Parties reasonably request that the other issuer pre-approve any acquisition of equity securities or derivatives thereof for the express purpose of exempting the interests of any director or officer of the Company or any of its Subsidiaries in such transactions from Section 16(b) of the Exchange Act pursuant to Rule 16b-3 thereunder, the Company shall require that such other issuer pre-approve any such acquisitions of equity securities or derivatives thereof for the express purpose of exempting the Purchaser Parties', their Affiliates and such Investor Nominee's interests (with respect to the Purchaser and its Affiliates, to the extent such persons may be deemed to be "directors by deputization" of such other issuer) in such transactions from Section 16(b) of the Exchange Act pursuant to Rule 16b-3 thereunder, to the extent such exemption is available to such Persons.

### 2.9 Voting Agreement.

(a) For so long as the Investor Parties have the right to designate or nominate one or more directors to the Board of Directors pursuant to <u>Section 2.1</u> or any Investor Nominee continues to serve on the Board of Directors, at each meeting of the stockholders of the Company and at every postponement or adjournment thereof, the Investor Parties or Other Holders, as applicable, shall take such action as may be required so that all Conversion Shares owned, directly or indirectly, of record or beneficially by such Investor Parties and Other Holders and entitled to vote at such meeting of stockholders are voted (a) in favor of each director nominated or recommended by the Board of Directors for election at any such meeting, and against the removal of any director who has been elected following nomination or recommendation by the Board of Directors, (b) against any stockholder nomination for director that is not approved and

recommended by the Board of Directors for election at any such meeting, (c) in favor of the Company's "say-on-pay" proposal and any proposal by the Company relating to equity compensation that has been approved by the Board of Directors or the Compensation Committee of the Board of Directors (or any successor committee or committee performing similar functions, however denominated), and (d) in favor of the Company's proposal for ratification of the appointment of the Company's independent registered public accounting firm. In furtherance of the foregoing, for so long as the Investor Parties have the right to designate or nominate a director to the Board of Directors pursuant to Section 2.1, the Investor Parties and Other Holders shall take such action as may be required so that the applicable Investor Parties and Other Holders are present, in person or by proxy, at each meeting of the stockholders of the Company and at every postponement or adjournment thereof so that all Conversion Shares owned, directly or indirectly, of record or beneficially by the Investor Parties and Other Holders may be counted for the purposes of determining the presence of a quorum and voted in accordance with the terms and conditions of this Section 2.9.

(b) The provisions in <u>Section 2.9(a)</u> will terminate and cease to have effect if, following any conversion of Notes into shares of Common Stock of the Company in connection with the Rights Offering contemplated by the Purchase Agreement, the Investor acquires control of a majority of the voting power of the Company.

#### 2.10 Nasdaq Listing.

(a) To the extent it has not already done so, promptly following the execution of this Agreement, the Company shall apply to cause the Common Stock issuable upon conversion of the Notes in connection with the Rights Offering contemplated by the Purchase Agreement to be approved for listing on the Nasdaq Stock market, subject to official notice of issuance.

(b) The Company agrees to use its best efforts to maintain its listing on the Nasdaq Stock Market at all times that the Investor holds any Notes or Conversion Shares.

2.11 **Information Barriers**. The Company (i) acknowledges that the Investor has advised the Company that the Investor and its Affiliates are in the business of holding, investing and trading in securities, and (ii) so long as no information is disclosed in violation of <u>Section 2.5</u>, agrees that this Agreement shall not in any way restrict or limit the activities of the Investor, any of its Affiliates or any of their respective employees (or, for the avoidance of doubt, the Investor Nominee) in the business of holding, investing and trading in securities so long as such activities are in compliance with applicable securities laws, including, but not limited to, where (A) such activities are undertaken by the Investor, employees and/or their Affiliates (or, for the avoidance of doubt, the Investor Nominee) that are not then in possession of Confidential Information constituting material non-public information to the Investor, other employees and Affiliates or (C) an affirmative defense pursuant to paragraph (c) of Rule 10b5-1 under the Exchange Act of 1934 is otherwise available.

#### SECTION 3 MISCELLANEOUS

**3.1 Counterparts**. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement and will become effective when one or more counterparts have been signed by a party and delivered to the other parties. Copies of executed counterparts of signature pages to this Agreement may be transmitted by PDF (portable document format) or facsimile and such PDFs or facsimiles (including any electronic signature covered by the U.S. federal ESIGN Act of 2000, Uniform Electronic Transactions Act, the Electronic Signatures and Records Act or other applicable law, e.g., www.docusign.com) will be deemed as sufficient as if actual signature pages had been delivered.

### 3.2 Governing Law.

(a) This Agreement shall be governed by, and construed in accordance with, the laws of the state of New York, without giving effect to any choice of law or conflict of law rules or provisions (whether of the state of New York or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the state of New York.

(b) Any dispute relating hereto shall be heard first in the New York Court of Appeals, and, if applicable, in any state or federal court located in of New York in which appeal from the New York Court of Appeals may validly be taken under the laws of the State of New York (each a "*Chosen Court*" and collectively, the "*Chosen Courts*"), and the parties agree to the exclusive jurisdiction and venue of the Chosen Courts. Such Persons further agree that any proceeding seeking to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement or the transactions contemplated hereby or by any matters related to the foregoing (the "*Applicable Matters*") shall be brought exclusively in a Chosen Court, and that any proceeding arising out of this Agreement or any other Applicable Matter shall be deemed to have arisen from a transaction of business in the state of New York, and each of the foregoing Persons hereby irrevocably consents to the jurisdiction of such Chosen Courts in any such proceeding and irrevocably and unconditionally waives, to the fullest extent permitted by law, any objection that such Person may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such Chosen Court or that any such proceeding brought in any such Chosen Court forum.

(c) Such Persons further covenant not to bring a proceeding with respect to the Applicable Matters (or that could affect any Applicable Matter) other than in such Chosen Court and not to challenge or enforce in another jurisdiction a judgment of such Chosen Court.

(d) Process in any such proceeding may be served on any Person with respect to such Applicable Matters anywhere in the world, whether within or without the jurisdiction of any such Chosen Court. Without limiting the foregoing, each such Person agrees that service of process on such party as provided in <u>Section 3.5</u> shall be deemed effective service of process on such Person.

(e) *Waiver of Jury Trial.* EACH PARTY HERETO, FOR ITSELF AND ITS AFFILIATES, HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, SUIT OR OTHER PROCEEDING (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THE ACTIONS OF THE PARTIES HERETO OR THEIR RESPECTIVE AFFILIATES PURSUANT TO THIS AGREEMENT OR IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE OR ENFORCEMENT HEREOF.

**3.3 Entire Agreement; No Third Party Beneficiary**. This Agreement and the Purchase Agreement contain the entire agreement by and among the parties with respect to the subject matter hereof and all prior negotiations, writings and understandings relating to the subject matter of this Agreement. This Agreement is not intended to confer upon any Person not a party hereto (or their successors and permitted assigns) any rights or remedies hereunder, provided that (a) the Non-Recourse Parties shall be third-party beneficiaries of <u>Section 3.14</u>, (b) the Observer and the Director Nominees shall each be third-party beneficiaries of <u>Section 2.11</u>.

**3.4 Expenses**. All fees, costs and expenses incurred in connection with this Agreement, the Purchase Agreement and the transactions contemplated hereby and thereby, respectively, including accounting and legal fees shall be paid by the party incurring such expenses, except that, upon consummation of the First Tranche Investment, the Company shall reimburse the Investor for all reasonable and documented out-of-pocket costs and expenses, including legal fees, expenses, other professional fees and expenses, and all reasonable out-of-pocket due diligence expenses, incurred by the Investor in connection with the transactions contemplated by this Agreement and the Purchase Agreement.

**3.5** Notices. All notices, requests, demands and other communications under this Agreement shall be in writing and shall be deemed to have been duly given or made as follows: (a) if sent by registered or certified mail in the United States return receipt requested, upon receipt; (b) if sent by nationally recognized overnight air courier, one (1) Business Day after mailing; (c) if sent by e-mail transmission, with a copy sent on the same day in the manner provided in the foregoing clause (a) or (b), when transmitted and receipt is confirmed; and (d) if otherwise actually personally delivered, when delivered, *provided*, that such notices, requests, demands and other communications are delivered to the address set forth below, or to such other address as any party shall provide by like notice to the other parties to this Agreement:

If to the Company, to:

Eargo, Inc. 2665 North First Street, Suite 300 San Jose, CA 95134 Email: [\*\*\*] Attention: Chief Legal Officer

with a copy (which shall not constitute notice) to:

Latham & Watkins LLP 10250 Constellation Blvd. Suite 1100 Los Angeles, CA 90067 E-mail: [\*\*\*] Attention: David Zaheer; Charles Ruck; Kathleen Wells

If to Investor, to:

c/o Patient Square Equity Advisors, LP 2884 Sand Hill Road, Suite 100 Menlo Park, CA 94025 E-mail: [\*\*\*] Attention: Adam Fliss, Justin Sabet-Peyman

with a copy (which shall not constitute notice) to:

Ropes & Gray LLP Three Embarcadero Center San Francisco, CA 94111 Email: [\*\*\*] Attention: Jason Freedman

and

Ropes & Gray LLP Prudential Tower 800 Boylston Street Boston, MA 02199 E-mail: [\*\*\*] Attention: Tom Fraser

**3.6 Successors and Assigns**. This Agreement will be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns. This Agreement may be assigned in connection with a Transfer permitted by <u>Section 2.2</u>. No other assignment of this Agreement or of any rights or obligations hereunder may be made by any party hereto without the prior written consent of the other parties hereto. Any purported assignment or delegation in violation of this Agreement shall be null and void ab initio.

**3.7 Headings**. The Section, Article and other headings contained in this Agreement are inserted for convenience of reference only and will not affect the meaning or interpretation of this Agreement.

**3.8 Amendments and Waivers**. This Agreement may not be modified or amended except by an instrument or instruments in writing signed by each party hereto. Any party hereto may, only by an instrument in writing, waive compliance by any other party or parties hereto with any term or provision hereof on the part of such other party or parties hereto to be performed or complied with. No failure or delay of any party in exercising any right or remedy hereunder

shall operate as a waiver thereof, nor will any single or partial exercise of any right or power, or any abandonment or discontinuance of steps to enforce such right or power, preclude any other or further exercise thereof or the exercise of any other right or power. The waiver by any party hereto of a breach of any term or provision hereof shall not be construed as a waiver of any subsequent breach. The rights and remedies of the parties hereunder are cumulative and are not exclusive of any rights or remedies that they would otherwise have hereunder.

#### 3.9 Interpretation; Absence of Presumption.

(a) For the purposes hereof: (i) words in the singular shall be held to include the plural and vice versa and words of one gender shall be held to include the other gender as the context requires; (ii) the terms "hereof," "herein," and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement as a whole (including all of the Schedules and Exhibits) and not to any particular provision of this Agreement, and Section, paragraph, Exhibit and Schedule references are to the Sections, paragraphs, Exhibits, and Schedules to this Agreement unless otherwise specified; (iii) the word "including" and words of similar import when used in this Agreement shall mean "including, without limitation," unless the context otherwise requires or unless otherwise specified; and (iv) the word "or" shall not be exclusive.

(b) With regard to each and every term and condition of this Agreement and any and all agreements and instruments subject to the terms hereof, the parties hereto understand and agree that the same have or has been mutually negotiated, prepared and drafted, and if at any time the parties hereto desire or are required to interpret or construe any such term or condition or any agreement or instrument subject hereto, no consideration will be given to the issue of which party hereto actually prepared, drafted or requested any term or condition of this Agreement or any agreement or instrument subject hereto.

**3.10 Severability**. Any provision hereof that is held to be invalid, illegal or unenforceable in any respect by a court of competent jurisdiction, shall be ineffective only to the extent of such invalidity, illegality or unenforceability, without affecting in any way the remaining provisions hereof, *provided*, *however*, that the parties will attempt in good faith to reform this Agreement in a manner consistent with the intent of any such ineffective provision for the purpose of carrying out such intent.

**3.11 Specific Performance**. The parties hereto agree that irreparable damage could occur and that a party may not have any adequate remedy at law in the event that any of the provisions of this Agreement are not performed in accordance with their terms or were otherwise breached. Accordingly, each party shall without the necessity of proving the inadequacy of money damages or posting a bond be entitled to seek an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms, provisions and covenants contained therein, this being in addition to any other remedy to which they are entitled at law or in equity.

3.12 Corporate Opportunities. The Company, on behalf of itself and its Subsidiaries, to the fullest extent permitted by applicable law, notwithstanding anything to the contrary (including Company policies, guidelines and codes in effect from time to time), (a) acknowledges and affirms that the Investor and its Representatives, including any Investor Nominee (the "Investor Group"): (i) have participated (directly or indirectly) and will continue to participate (directly or indirectly) in private equity, venture capital and other direct investments in corporations, joint ventures, limited liability companies and other entities ("Other Investments"), including Other Investments engaged in various aspects of businesses similar to those engaged in by the Company and its Subsidiaries (and related services businesses) that may, are or will be competitive with the Company's or any of its Subsidiaries' businesses or that could be suitable for the Company's or any of its Subsidiaries' interests, (ii) have interests in, participate with, aid and maintain seats on the board of directors or similar governing bodies of, Other Investments, (iii) may develop or become aware of business opportunities for Other Investments; and (iv) may or will, as a result of or arising from the matters referenced in this Section 3.12, the nature of the Investor Group's businesses and other factors, have conflicts of interest or potential conflicts of interest, (b) hereby renounces and disclaims any interest or expectancy in any business opportunity (including any Other Investments or any other opportunities that may arise in connection with the circumstances described in the foregoing clauses (a)(i) through (a)(v) (each, a "Renounced Business Opportunity")), and (c) acknowledges and affirms that no member of Investor Group, including any Investor Nominee, shall have any obligation to communicate or offer any Renounced Business Opportunity to the Company or any of its Subsidiaries, and any member of Investor Group may pursue a Renounced Business Opportunity. Notwithstanding anything to the contrary in the foregoing, the Company does not renounce its interest in any corporate opportunity if such corporate opportunity was expressly offered to any Investor Nominee solely in his or her capacity as a member of the Board of Directors; provided that such opportunity has not been separately presented to the Investor or its Affiliates. Notwithstanding anything to the contrary in the foregoing, the Company shall not be prohibited from pursuing any Renounced Business Opportunity as a result of this Section 3.12.

**3.13 Public Announcement**. Subject to each party's disclosure obligations imposed by applicable law or the rules of any stock exchange upon which its securities are listed, each of the parties hereto will cooperate with each other in the development and distribution of all news releases and other public information disclosures with respect to this Agreement and any of the transactions contemplated by this Agreement, and neither the Company nor any Investor will make any such news release or public disclosure without first consulting with the other, and, in each case, also receiving the other's consent (which shall not be unreasonably withheld or delayed) and each party shall coordinate with the party whose consent is required with respect to any such news release or public disclosure. Notwithstanding the foregoing, this <u>Section 3.13</u> shall not apply to any press release or other public statement made by the Company or an Investor (a) that is consistent with prior disclosure and does not contain any information relating to the transactions that has not been previously announced or made public in accordance with the terms of this Agreement or (b) is made to its auditors, attorneys, accountants, financial advisors, existing and prospective limited partners or other Permitted Transferees.

**3.14 Non-Recourse Parties**. Without limiting any of the provisions contained in the equity commitment letter executed by Patient Square Equity Partners, LP on the date hereof, this Agreement may only be enforced against, and any claim or cause of action based upon, arising out of, or related to this Agreement, or the transactions contemplated hereby may only be brought against the entities that are expressly named as parties hereto and their respective successors and assigns. Except as set forth in the immediately preceding sentence, no past, present or future director, officer, employee, incorporator, member, partners, stockholder, Affiliate, agent, attorney,

advisor or representative of any party hereto (collectively, the "*Non-Recourse Parties*") shall have any liability for any obligations or liabilities of any party hereto under this Agreement or for any claim based on, in respect of, or by reason of, the transactions contemplated hereby (whether in contract, in tort, in law or in equity or provided by statute, whether by or through attempted piercing of the corporate, limited partnership or limited liability company veil or otherwise), except to the extent expressly set forth in this Agreement. For the avoidance of doubt, this <u>Section 3.14</u> is intended to benefit and may be enforced by each Non-Recourse Party (and each such Person shall be a third-party beneficiary of this <u>Section 3.14</u>) and shall be binding on all respective successors and permitted assigns thereof.

(The next page is the signature page)

The parties have caused this Investor Rights Agreement to be executed as of the date first written above.

EARGO, INC.

By: /s/ Christian Gormsen Name: Christian Gormsen Title: Chief Executive Officer and President

PSC ECHO, LP

By: /s/ Adam Fliss

Name: Adam Fliss Title: General Counsel and Chief Compliance Officer

[Signature Page to Investors' Rights Agreement]

## EXHIBIT A

## **BOARD OBSERVER AGREEMENT**

## EXHIBIT B

## INDEMNIFICATION AND ADVANCEMENT AGREEMENT

Certain confidential information contained in this document, marked by [\*\*\*], has been omitted pursuant to Item 601(b)(10)(iv) of Regulation S-K because it is both (i) not material and (ii) the type of information that the registrant treats as private or confidential.

### REGISTRATION RIGHTS AGREEMENT

### BY AND BETWEEN

EARGO, INC.

AND

PSC ECHO, LP

Dated as of June 24, 2022

# TABLE OF CONTENTS

Article I Resale Sł	half Degistration	Page
	-	1
Section 1.1	Resale Shelf Registration Statement	1
Section 1.2 Section 1.3	Effectiveness Period	1
Section 1.4	Subsequent Shelf Registration Supplements and Amendments	1 2
Section 1.4	Subsequent Holder Notice	2
Section 1.6	Underwritten Offering	2
Section 1.7	Take-Down Notice	3
Article II Compan	y Registration	3
Section 2.1	Notice of Registration	3
Section 2.2	Underwriting	3
Section 2.3	Right to Terminate Registration	4
Article III Additio	nal Provisions Regarding Registration Rights	4
Section 3.1	Registration Procedures	4
Section 3.2	Limitation on Subsequent Registration Rights	6
Section 3.3	Expenses of Registration	6
Section 3.4	Information by Holders	6
Section 3.5	"Market Stand-Off" Agreement	7
Article IV Indemn	ification	8
Section 4.1	Indemnification by Company	8
Section 4.2	Indemnification by Holders	8
Section 4.3	Notification	9
Section 4.4	Contribution	10
Section 4.5	Indemnification Priority	10
Article V Transfer	and Termination of Registration Rights	11
Section 5.1	Transfer of Registration Rights	11
Section 5.2	Termination of Registration Rights	11
Article VI Miscell	aneous	11
Section 6.1	Counterparts	11
Section 6.2	Governing Law; Waiver of Jury Trial.	11
Section 6.3	Entire Agreement; No Third Party Beneficiary	12
Section 6.4	Expenses	12
Section 6.5	Notices	12

i

Section 6.6	Successors and Assigns	13
Section 6.7	Headings	14
Section 6.8	Amendments and Waivers	14
Section 6.9	Interpretation; Absence of Presumption	14
Section 6.10	Severability	14
Section 6.11	Non-Recourse Parties	14
	ü	
	11	

#### **REGISTRATION RIGHTS AGREEMENT**

This REGISTRATION RIGHTS AGREEMENT (the "Agreement") is entered into as of June 24, 2022, by and between Eargo, Inc., a Delaware corporation (including its successors and permitted assigns, the "Company") and the investor listed on the signature page hereto and the other investors from time to time party hereto (each, an "Investor" and collectively, the "Investors"). Capitalized terms used but not defined elsewhere herein are defined in <u>Exhibit A</u>.

The Company has entered into a Note Purchase Agreement, dated as of the date hereof (as amended from time to time, the "**Purchase Agreement**"), with the Investors pursuant to which the Company is selling to the Investors, and the Investors are purchasing from the Company, up to \$125 million principal amount of Senior Secured Notes, which may be converted into shares of Common Stock, subject to the terms and conditions set forth in the Purchase Agreement.

As a condition to each of the parties' obligations under the Note Purchase Agreement, the Company and the Investors are entering into this Agreement for the purpose of granting certain registration and other rights to the Investors.

In consideration of the premises and the mutual representations, warranties, covenants and agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties agree as follows:

### ARTICLE I RESALE SHELF REGISTRATION

Section 1.1 <u>Resale Shelf Registration Statement</u>. Subject to the other applicable provisions of this Agreement, the Company shall use its reasonable best efforts to file within ten (10) Business days of the Rights Offering Closing Date a registration statement covering the sale or distribution from time to time by the Holders, on a delayed or continuous basis pursuant to Rule 415 of the Securities Act of all of the Registrable Securities on Form S-1 (the "**Resale Shelf Registration Statement**" and such registration, the "**Resale Shelf Registration**"). The Company shall use its reasonable best efforts to cause such Resale Shelf Registration Statement to be declared effective by the Commission as promptly as practicable after the filing thereof.

Section 1.2 <u>Effectiveness Period</u>. Once declared effective, the Company shall, subject to the other applicable provisions of this Agreement, use its reasonable best efforts to cause the Resale Shelf Registration Statement to be continuously effective and usable until such time as there are no longer any Registrable Securities (the "**Effectiveness Period**").

Section 1.3 <u>Subsequent Shelf Registration</u>. If any Shelf Registration ceases to be effective under the Securities Act for any reason at any time during the Effectiveness Period, the Company shall use its reasonable best efforts to promptly cause such Shelf Registration to again become effective under the Securities Act (including obtaining the prompt withdrawal of any order suspending the effectiveness of such Shelf Registration), and in any event shall within thirty (30) days of such cessation of effectiveness, amend such Shelf Registration in a manner reasonably expected to obtain the withdrawal of any order suspending the effectiveness of such Shelf Registration statement (a "Subsequent Shelf Registration") for an offering to be made on a delayed or continuous basis pursuant to Rule 415 of the Securities Act

registering the resale from time to time by Holders thereof of all securities that are Registrable Securities as of the time of such filing. If a Subsequent Shelf Registration is filed, the Company shall use its reasonable best efforts to (a) cause such Subsequent Shelf Registration to become effective under the Securities Act as promptly as is reasonably practicable after such filing, but in no event later than the date that is ninety (90) days after such Subsequent Shelf Registration is filed and (b) keep such Subsequent Shelf Registration (or another Subsequent Shelf Registration) continuously effective until the end of the Effectiveness Period. Any such Subsequent Shelf Registration shall be a Registration Statement on Form S-1 (or a Form S-3, to the extent that the Company is eligible to use such form). If the Company becomes eligible to use a Form S-3, if the Investors so request, the Company shall promptly amend the existing Shelf Registration on such form or file a new Shelf Registration replacing the existing Shelf Registration on such form.

Section 1.4 <u>Supplements and Amendments</u>. The Company shall supplement and amend any Shelf Registration if required by the rules, regulations or instructions applicable to the registration form used by the Company for such Shelf Registration if required by the Securities Act or as reasonably requested by the Investors.

Section 1.5 <u>Subsequent Holder Notice</u>. If a Person becomes a Holder of Registrable Securities after a Shelf Registration becomes effective under the Securities Act, the Company shall, as promptly as is reasonably practicable following delivery of written notice to the Company of such Person becoming a Holder and requesting for its name to be included as a selling securityholder in the prospectus related to the Shelf Registration (a "**Subsequent Holder Notice**"):

(a) if required and permitted by applicable law, file with the Commission a supplement to the related prospectus or a post-effective amendment to the Shelf Registration so that such Holder is named as a selling securityholder in the Shelf Registration and the related prospectus in such a manner as to permit such Holder to deliver a prospectus to purchasers of the Registrable Securities in accordance with applicable law; <u>provided</u>, <u>however</u>, that the Company shall not be required to file more than one post-effective amendment or a supplement to the related prospectus for such purpose in any forty-five (45) day period;

(b) if, pursuant to Section 1.5(a), the Company shall have filed a post-effective amendment to the Shelf Registration, use its reasonable best efforts to cause such post-effective amendment to become effective under the Securities Act as promptly as is reasonably practicable, but in any event by the date that is ninety (90) days after the date such post-effective amendment is required by Section 1.5(a) to be filed; and

(c) notify such Holder as promptly as is reasonably practicable after the effectiveness under the Securities Act of any post-effective amendment filed pursuant to  $\underline{Section 1.5(a)}$ .

Section 1.6 <u>Underwritten Offering</u>. The Investors (at the request of Holders of a majority of the Registrable Securities) may on up to four (4) occasions per twelve (12)-month period after the Resale Shelf Registration Statement becomes effective deliver a written notice to the Company specifying that the sale of some or all of the Registrable Securities subject to the Shelf Registration, not to be less than ten million dollars (\$10,000,000) (or such lesser amount as is then held by the Investors) offering amount of Registrable Securities, is intended to be conducted through an underwritten offering (the "**Underwritten Offering**"). In the event of an Underwritten Offering:

(a) The Investors shall select the managing underwriter or underwriters to administer the Underwritten Offering, subject to the Company's consent (which shall not be unreasonably withheld, conditioned or delayed).

(b) Notwithstanding any other provision of this <u>Section 1.6</u>, if the managing underwriter or underwriters of a proposed Underwritten Offering advises the Board of Directors of the Company that in its or their opinion the number of Registrable Securities requested to be included in such Underwritten Offering exceeds the number which can be sold in such Underwritten Offering in light of market conditions, the Registrable Securities shall be included on a pro rata basis upon the number of securities that each Holder shall have requested to be included in such offering. If any Holder disapproves of the terms of any such underwriting, such Holder may elect to withdraw therefrom by written notice to the Company and the managing underwriter or underwriters.

Section 1.7 <u>Take-Down Notice</u>. Subject to the other applicable provisions of this Agreement, at any time that any Shelf Registration Statement is effective, if a Holder delivers a notice to the Company (a "**Take-Down Notice**") stating that it intends to effect a sale or distribution of all or part of its Registrable Securities included by it on any Shelf Registration Statement (a "**Shelf Offering**"), including an Underwritten Offering, and stating the number of Registrable Securities to be included in such Shelf Offering, then, subject to the other applicable provisions of this Agreement, the Company shall promptly amend or supplement the Shelf Registration Statement as may be necessary in order to enable such Registrable Securities to be sold and distributed pursuant to the Shelf Offering (and shall use its reasonable best efforts to do so within three (3) Business Days of receipt of the Take-Down Notice).

## ARTICLE II COMPANY REGISTRATION

Section 2.1 <u>Notice of Registration</u>. If at any time or from time to time the Company shall determine to file a registration statement for an underwritten public offering of its Common Stock (for the avoidance of doubt, the following will <u>not</u> apply to any registration statement filed on a Form S-4, Form S-8 or any successor forms), the Company will:

(a) promptly give to the Investors written notice thereof; and

(b) subject to <u>Section 2.2</u>, include in such registration and underwritten offering (and any related qualification under blue sky laws or other compliance) all the Registrable Securities specified in a written request or requests made within ten (10) days after receipt of such written notice from the Company by the Investors (on behalf of the applicable Holders).

Section 2.2 <u>Underwriting</u>. The right of any Holder to registration pursuant to <u>Section 1.6</u> or this <u>Article II</u> shall be conditioned upon such Holder's participation in such underwriting and the inclusion of Registrable Securities in the underwriting to the extent provided herein. Each Holder proposing to distribute its securities through such underwriting shall (together with the Company and any other holders distributing their securities through such underwriting)

enter into and perform such Holder's obligations under an underwriting agreement with the managing underwriter selected for such underwriting. Notwithstanding any other provision of this <u>Article II</u>, if the managing underwriter or underwriters of a proposed underwritten offering with respect to which Holders of Registrable Securities have exercised their piggyback registration rights advise the Board of Directors of the Company that in its or their opinion the number of Registrable Securities requested to be included in the offering thereby and all other securities proposed to be sold in the offering exceeds the number which can be sold in such underwritten offering in light of market conditions, the Registrable Securities and such other securities to be included in such underwritten offering shall be allocated, (a) first, in the event such offering was initiated by the Company, up to the total number of securities that the Company has requested to be included in such registration and (b) second, and only if all the securities referred to in clause (a) have been included, up to the total number of securities referred to in clause dupon the number of securities that each of them shall have requested to be included in such offering (pro rata based upon the number of securities that each of them shall have requested to be included in such offering that, in the opinion of the managing underwriter or underwriters can be sold without having such adverse effect. If any Holder disapproves of the terms of any such underwriting, such Holder may elect to withdraw therefrom by written notice to the Company and the managing underwriter or underwriters shall be withdrawn from such registration.

Section 2.3 <u>Right to Terminate Registration</u>. The Company or the holders of securities who have caused a registration statement to be filed as contemplated by this <u>Article II</u>, as the case may be, shall have the right to have any registration initiated by it or them under this <u>Article II</u> terminated or withdrawn prior to the effectiveness thereof, whether or not any Holder has elected to include securities in such registration.

#### ARTICLE III ADDITIONAL PROVISIONS REGARDING REGISTRATION RIGHTS

Section 3.1 <u>Registration Procedures</u>. In the case of each registration effected by the Company pursuant to <u>Article I</u> or <u>II</u>, the Company will keep the Investors reasonably informed as to the status thereof and, at its expense, the Company will:

(a) prepare and file with the Commission a registration statement with respect to such securities in accordance with the applicable provisions of this Agreement;

(b) prepare and file with the Commission such amendments, including post-effective amendments, and supplements to such registration statement and the prospectus used in connection with such registration statement as may be necessary to comply with the provisions of the Securities Act with respect to the disposition of all securities covered by such registration statement and as may be necessary to keep the registration statement continuously effective for the period set forth in this Agreement;

(c) furnish to the Investors and to the Holders' legal counsel copies of the registration statement proposed to be filed, and provide the Investors and such their legal counsel the reasonable opportunity to review and comment on such registration statement;

(d) furnish to the Investors and to the underwriters of the securities being registered such reasonable number of copies of the registration statement, preliminary prospectus and final prospectus as the Investors or such underwriters may reasonably request in order to facilitate the public offering of such securities;

(e) enter into and perform such customary agreements (including, as applicable, underwriting agreements in customary form) and take all such other actions as the Investors or the underwriters, if any, reasonably request in order to expedite or facilitate the disposition of such Registrable Securities (including, without limitation, making available the executive officers of the Company and participating in "road shows," investor presentations, marketing events and other selling efforts and effecting a stock or unit split or combination, recapitalization or reorganization);

(f) use reasonable best efforts to notify the Investors at any time when a prospectus relating thereto is required to be delivered under the Securities Act of the Company's knowledge of the happening of any event as a result of which the prospectus included in such registration statement, as then in effect, includes an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make the statements therein not misleading or incomplete in the light of the circumstances then existing, and, subject to <u>Section 3.1(j)</u>, at the request of the Investors, prepare promptly and furnish to the Investors a reasonable number of copies of a supplement to or an amendment of such prospectus as may be necessary so that, as thereafter delivered to the purchaser of such shares, such prospectus shall not include an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading or incomplete in the light of the circumstances then existing and supplement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading or incomplete in the light of the circumstances then existing;

(g) use reasonable best efforts to register and qualify the securities covered by such registration statement under such other securities or blue sky laws of such jurisdictions as shall be reasonably requested by the Investors; <u>provided</u>, <u>however</u>, that the Company shall not be required in connection therewith or as a condition thereto to qualify to do business or to file a general consent to service of process in any such states or jurisdictions;

(h) in the event that the Registrable Securities are being offered in an underwritten public offering, enter into and perform its obligations under an underwriting agreement in accordance with the applicable provisions of this Agreement;

(i) use reasonable best efforts to furnish, on the date that such Registrable Securities are delivered to the underwriters for sale, if such securities are being sold through underwriters, (i) an opinion, dated as of such date, of the legal counsel representing the Company for the purposes of such registration, in form and substance as is customarily given to underwriters in an underwritten public offering, addressed to the underwriters, if any, and (ii) a letter dated as of such date, from the independent certified public accountants of the Company, in form and substance as is customarily given by independent certified public offering, addressed to the underwriters; and

(j) notwithstanding any other provision of this Agreement, if the Board of Directors of the Company has determined in good faith that the disclosure necessary for continued use of the prospectus and registration statement by the Holders could be materially detrimental to the Company, the Company shall have the right not to file or not to cause the effectiveness of any registration covering any Registrable Securities and to suspend the use of the prospectus and the registration statement covering any Registrable Security for such period of time as its use would be materially detrimental to the Company by delivering written notice of such suspension to the Investors; <u>provided</u>, <u>however</u>, that in any 12-month period the Company may exercise the right to such suspension not more than twice. From and after the date of a notice of suspension under this <u>Section 3.1(j)</u>, each Holder agrees not to use the prospectus or registration statement until the earlier of (1) notice from the Company that such suspension has been lifted or (2) the day following the sixtieth (60th) day of suspension within any 12-month period.

Section 3.2 <u>Limitation on Subsequent Registration Rights</u>. From and after the date hereof, the Company shall not enter into any agreement granting any holder or prospective holder of any securities of the Company registration rights with respect to such securities that conflict with the rights granted to the Holders herein, without the prior written consent of the Investors. It is agreed that the granting of pro rata registration rights to any other investor in the Company shall not be considered to conflict with the rights granted to the Holders herein.

Section 3.3 <u>Expenses of Registration</u>. All Registration Expenses incurred in connection with any registration pursuant to <u>Article I</u> or <u>II</u> shall be borne by the Company. All Selling Expenses relating to securities registered on behalf of the Holders shall be borne by the Holders of the registered securities included in such registration.

Section 3.4 <u>Information by Holders</u>. The Holder or Holders of Registrable Securities included in any registration shall furnish to the Company such information regarding such Holder or Holders and their Affiliates, the Registrable Securities held by them and the distribution proposed by such Holder or Holders and their Affiliates as the Company may reasonably request in writing and as shall be required in connection with any registration, qualification or compliance referred to in this Agreement. It is understood and agreed that the obligations of the Company under <u>Article I or II</u> are conditioned on the timely provisions of the foregoing information by such Holder or Holders and, without limitation of the foregoing, will be conditioned on compliance by such Holder or Holders with the following:

(a) such Holder or Holders will, and will cause their respective Affiliates to, cooperate with the Company in connection with the preparation of the applicable registration statement, and for so long as the Company is obligated to keep such registration statement effective, such Holder or Holders will and will cause their respective Affiliates to, provide to the Company, in writing and in a timely manner, for use in such registration statement (and expressly identified in writing as such), all information regarding themselves and their respective Affiliates and such other information as may be required by applicable law to enable the Company to prepare such registration statement and the related prospectus covering the applicable Registrable Securities owned by such Holder or Holders and to maintain the currency and effectiveness thereof;

(b) during such time as such Holder or Holders and their respective Affiliates may be engaged in a distribution of the Registrable Securities, such Holder or Holders will, and they will cause their Affiliates to, comply with all laws applicable to such distribution, including Regulation M promulgated under the Exchange Act, and, to the extent required by such laws, will, and will cause their Affiliates to, among other things: (i) not engage in any stabilization activity in connection with the securities of the Company in contravention of such laws; (ii) distribute the Registrable Securities acquired by it solely in the manner described in the applicable registration statement; and (iii) if required by applicable law, cause to be furnished to each agent or broker-dealer to or through whom such Registrable Securities may be offered, or to the offeree if an offer is made directly by such Holder or Holders or their respective Affiliates, such copies of the applicable prospectus (as amended and supplemented to such date) and documents incorporated by reference therein as may be required by such agent, broker-dealer or offeree;

(c) such Holder or Holders shall, and they shall cause their respective Affiliates to, permit the Company and its representatives and agents to examine such documents and records and will supply in a timely manner any information as they may be reasonably request to provide in connection with the offering or other distribution of Registrable Securities by such Holder or Holders; and

(d) on receipt of written notice from the Company of the happening of any of the events specified in <u>Section 3.1(j)</u>, or that requires the suspension by such Holder or Holders and their respective Affiliates of the distribution of any of the Registrable Securities owned by such Holder or Holders, then such Holders shall, and they shall cause their respective Affiliates to, cease offering or distributing the Registrable Securities owned by such Holder or Holders until the offering and distribution of the Registrable Securities owned by such Holder or Holders may recommence in accordance with the terms hereof and applicable law.

Section 3.5 <u>"Market Stand-Off" Agreement</u>. If requested by the representatives of any underwriters in a registered public sale of securities by the Company, in which the Company gave the Holders an opportunity to participate in accordance with <u>Article II</u>, the Holders shall not sell, transfer, make any short sale of, grant any option for the purchase of, or enter into any hedging or similar transaction with the same economic effect as a sale with respect to, any Common Stock (or other securities of the Company) held by the Holders (other than those included in the registration) for a period specified by the representatives of the managing underwriter or underwriters of Common Stock (or other securities of the Company convertible into Common Stock) not to exceed ten (10) days prior and ninety (90) days following the date of the final prospectus relating to such offering. Each of the Holders also shall execute and deliver any "lock-up" agreement reasonably requested by the representatives of any underwriters in connection with the foregoing.

#### ARTICLE IV INDEMNIFICATION

Section 4.1 Indemnification by Company. To the extent permitted by applicable law, the Company will, with respect to any Registrable Securities as to which registration or qualification or compliance under applicable "blue sky" laws has been effected pursuant to this Agreement, indemnify each Holder, each Holder's current and former officers, directors, partners and members, and each Person controlling such Holder within the meaning of Section 15 of the Securities Act, and each underwriter thereof, if any, and each Person who controls any such underwriter within the meaning of Section 15 of the Securities Act (collectively, the "Company Indemnified Parties"), against all expenses, claims, losses, damages and liabilities, joint or several, (or actions in respect thereof) arising out of or based on any untrue statement (or alleged untrue statement) of a material fact contained in any registration statement, prospectus, preliminary prospectus, offering circular or other document, or any amendment or supplement thereto incident to any such registration, qualification or compliance or based on any omission (or alleged omission) to state therein a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances in which they were made, not misleading, or any violation by the Company of any rule or regulation promulgated under the Securities Act, Exchange Act or state securities laws applicable to the Company in connection with any such registration, and the Company will reimburse each of the Company Indemnified Parties for any reasonable legal and any other expenses reasonably incurred in connection with investigating, preparing or defending any such claim, loss, damage, liability or action, as such expenses are incurred. The indemnity agreement contained in this Section 4.1 shall not apply to amounts paid in settlement of any loss, claim, damage, liability or action if such settlement is effected without the prior written consent of the Company (which consent shall not be unreasonably withheld or delayed), nor shall the Company be liable to a Holder in any such case for any such loss, claim, damage, liability or action (a) to the extent that it arises out of or is based upon a violation or alleged violation of any state or federal law (including any claim arising out of or based on any untrue statement or alleged untrue statement or omission or alleged omission in the registration statement or prospectus) which occurs in reliance upon and in conformity with written information furnished expressly for use in connection with such registration by or on behalf of any Holder regarding such Holder (it being understood that such information shall be limited to the name of such Holder, the number of securities held by such Holder, and information regarding beneficial ownership of such Holder included in the footnote to the selling stockholder table (such information, "Selling Stockholder Information")) or (b) in the case of a sale directly by a Holder of Registrable Securities (including a sale of such Registrable Securities through any underwriter retained by such Holder engaging in a distribution solely on behalf of such Holder), such untrue statement or alleged untrue statement or omission or alleged omission was corrected in a final or amended prospectus, and such Holder failed to deliver a copy of the final or amended prospectus at or prior to the confirmation of the sale of the Registrable Securities to the Person asserting any such loss, claim, damage or liability in any case in which such delivery is required by the Securities Act.

Section 4.2 <u>Indemnification by Holders</u>. To the extent permitted by applicable law, each Holder will, if Registrable Securities held by such Holder are included in the securities as to which such registration or qualification or compliance under applicable "blue sky" laws is being effected, indemnify, severally and not jointly, the Company, each of its directors, officers, partners and members, each underwriter, if any, of the Company's securities covered by such a registration, each Person who controls the Company or such underwriter within the meaning of Section 15 of the Securities Act, and each other Holder and each of such Holder's officers, partners and members and each Person controlling such Holder within the meaning of Section 15 of the Securities Act (collectively, the "Holder Indemnified Parties"), against all expenses, claims,

losses, damages and liabilities (or actions in respect thereof) arising out of or based on any untrue statement (or alleged untrue statement) of a material fact contained in any registration statement, prospectus, preliminary prospectus, offering circular or other document, or any amendment or supplement thereto incident to any such registration, qualification or compliance or based on any omission (or alleged omission) to state therein a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances in which they were made, not misleading, or any violation by such Holder of any rule or regulation promulgated under the Securities Act, Exchange Act or state securities law applicable to such Holder, and will reimburse each of the Holder Indemnified Parties for any reasonable legal or any other expenses reasonably incurred in connection with investigating, preparing or defending any such claim, loss, damage, liability or action, as such expenses are incurred, in each case to the extent, but only to the extent, that such untrue statement (or alleged untrue statement) or omission (or alleged omission) is made in such registration statement, prospectus, offering circular or other document in reliance upon and in conformity with written Selling Stockholder Information furnished to the Company by such Holder and stated to be specifically for use therein, provided, however, that in no event shall any indemnity under this Section 4.2 payable by a Holder exceed the amount by which the net proceeds actually received by such Holder from the sale of Registrable Securities included in such registration exceeds the amount of any other losses, expenses, settlements, damages, claims and liabilities that such Holder has been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission or violation. The indemnity agreement contained in this Section 4.2 shall not apply to amounts paid in settlement of any loss, claim, damage, liability or action if such settlement is effected without the prior written consent of the applicable Holder (which consent shall not be unreasonably withheld or delayed), nor shall the Holder be liable for any such loss, claim, damage, liability or action where such untrue statement or alleged untrue statement or omission or alleged omission was corrected in a final or amended prospectus, and the Company or the underwriters failed to deliver a copy of the final or amended prospectus at or prior to the confirmation of the sale of the Registrable Securities to the Person asserting any such loss, claim, damage or liability in any case in which such delivery is required by the Securities Act

Section 4.3 <u>Notification</u>. Each party entitled to indemnification under this <u>Article IV</u> (the "**Indemnified Party**") shall give notice to the party required to provide indemnification (the "**Indemnifying Party**") promptly after such Indemnified Party has actual knowledge of any claim as to which indemnity may be sought, and shall permit the Indemnifying Party to assume the defense of any such claim or any litigation resulting therefrom, <u>provided</u>, <u>however</u>, that counsel for the Indemnifying Party, who shall conduct the defense of such claim or litigation, shall be approved by the Indemnified Party (whose approval shall not unreasonably be withheld or delayed), and the Indemnified Party may participate in such defense at such party's expense; <u>provided</u>, <u>further</u>, <u>however</u>, that an Indemnified Party (together with all other Indemnified Parties) shall have the right to retain one (1) separate counsel, with the reasonable fees and expenses to be paid by the Indemnifying Party, if representation of such Indemnified Party and any other party represented by such counsel in such proceeding. The failure of any Indemnified Party to give notice as provided herein shall relieve the Indemnifying Party of its obligations under this <u>Article IV</u>, only to the extent that, the failure to give such notice is materially prejudicial or harmful to an Indemnifying Party's ability to defend such action. No Indemnifying Party, in the defense of any such claim or litigation, shall, except with the prior written consent of each Indemnified Party (which consent shall not be unreasonably withheld or

delayed), consent to entry of any judgment or enter into any settlement which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such Indemnified Party of a release from all liability in respect to such claim or litigation. The indemnity agreements contained in this <u>Article IV</u> shall not apply to amounts paid in settlement of any loss, claim, damage, liability or action if such settlement is effected without the prior written consent of the Indemnifying Party, which consent shall not be unreasonably withheld or delayed. The indemnification set forth in this <u>Article IV</u> shall be in addition to any other indemnification rights or agreements that an Indemnified Party may have.

Section 4.4 Contribution. If the indemnification provided for in this Article IV is held by a court of competent jurisdiction to be unavailable to an Indemnified Party, other than pursuant to its terms, with respect to any claim, loss, damage, liability or action referred to therein, then, subject to the limitations contained in Article IV, the Indemnifying Party, in lieu of indemnifying such Indemnified Party hereunder, shall contribute to the amount paid or payable by such Indemnified Party as a result of such claim, loss, damage, liability or action in such proportion as is appropriate to reflect the relative fault of the Indemnifying Party on the one hand and the Indemnified Party on the other in connection with the actions that resulted in such claims, loss, damage, liability or action, as well as any other relevant equitable considerations. The relative fault of the Indemnifying Party and of the Indemnified Party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission to state a material fact related to information supplied by the Indemnifying Party or by the Indemnified Party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission. The Company and the Holders agree that it would not be just and equitable if contribution pursuant to this Section 4.4 were based solely upon the number of entities from whom contribution was requested or by any other method of allocation which does not take account of the equitable considerations referred to above in this Section 4.4. In no event shall any Holder's contribution obligation under this Section 4.4 exceed the amount by which the net proceeds actually received by such Holder from the sale of Registrable Securities included in such registration exceeds the amount of any other losses, expenses, settlements, damages, claims and liabilities that such Holder has been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission or violation less any amounts paid pursuant to Section 4.3. No Person guilty of fraudulent misrepresentation (within the meaning of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation.

Section 4.5 Indemnification Priority. The Company hereby acknowledges and agrees that any of the Persons entitled to indemnification by the Company pursuant to Article IV (each, a "Company Indemnitee" and collectively, the "Company Indemnitees") may have certain rights to indemnification, advancement of expenses and/or insurance provided by other sources. The Company hereby acknowledges and agrees (i) that it is the indemnitor of first resort (i.e., its obligations to a Company Indemnitee are primary and any obligation of such other sources to advance expenses or to provide indemnification for the same expenses or liabilities incurred by such Company Indemnitee are secondary) and (ii) that it shall be required to advance the full amount of expenses incurred by a Company Indemnitee and shall be liable for the full amount of all expenses, judgments, penalties, fines and amounts paid in settlement to the extent legally permitted and as required by the terms of this Agreement without regard to any rights a Company Indemnitee may have against such other sources. The Company Indemnitee has sought indemnification, advancement of expenses or insurance from the Company Indemnitee with respect to any claim for which such Company Indemnitee has sought indemnification, advancement of expenses or insurance from the Company shall affect the foregoing, and that such other sources shall have a right of contribution and/or be subrogated to the extent of such advancement or payment to all of the rights of recovery of such Company Indemnitee against the Company.

#### ARTICLE V TRANSFER AND TERMINATION OF REGISTRATION RIGHTS

Section 5.1 <u>Transfer of Registration Rights</u>. The rights to cause the Company to register securities granted to a Holder under this Agreement may be assigned to a Permitted Transfere or Other Holder in connection with a Transfer of Registrable Securities to such Permitted Transfere or Other Holder in accordance with the Investors' Rights Agreement; <u>provided</u>, <u>however</u>, that (i) such transfer may otherwise be effected in accordance with applicable securities laws, (ii) prior written notice of such assignment is given to the Company, and (iii) such Permitted Transferee or Other Holder agrees in writing to be bound by, and subject to, this Agreement as a "Holder" pursuant to a written instrument in form and substance reasonably acceptable to the Company.

Section 5.2 <u>Termination of Registration Rights</u>. The rights of any particular Holder to cause the Company to register securities under <u>Articles I</u> and <u>II</u> shall terminate with respect to such Holder upon the date upon which such Holder no longer holds any Registrable Securities.

### ARTICLE VI MISCELLANEOUS

Section 6.1 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement, and will become effective when one or more counterparts have been signed by a party and delivered to the other parties. Copies of executed counterparts transmitted by telecopy, telefax or other electronic transmission service shall be considered original executed counterparts for purposes of this <u>Section 6.1</u>, provided that receipt of copies of such counterparts is confirmed.

#### Section 6.2 Governing Law; Waiver of Jury Trial.

(a) This Agreement shall be governed by, and construed in accordance with, the laws of the state of New York, without giving effect to any choice of law or conflict of law rules or provisions (whether of the state of New York or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the state of New York.

(b) Any dispute relating hereto shall be heard first in the Delaware Court of Chancery, and, if applicable, in any state or federal court located in of Delaware in which appeal from the Court of Chancery may validly be taken under the laws of the State of Delaware (each a "Chosen Court" and collectively, the "Chosen Courts"), and the parties agree to the exclusive jurisdiction and venue of the Chosen Courts. Such Persons further agree that any proceeding seeking to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement or the transactions contemplated hereby or by any matters related to the foregoing (the "Applicable Matters") shall be brought exclusively in a Chosen Court, and that any proceeding arising out of this Agreement or any other Applicable Matter shall be deemed to have

arisen from a transaction of business in the state of Delaware, and each of the foregoing Persons hereby irrevocably consents to the jurisdiction of such Chosen Courts in any such proceeding and irrevocably and unconditionally waives, to the fullest extent permitted by law, any objection that such Person may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such Chosen Court or that any such proceeding brought in any such Chosen Court has been brought in an inconvenient forum.

(c) Such Persons further covenant not to bring a proceeding with respect to the Applicable Matters (or that could affect any Applicable Matter) other than in such Chosen Court and not to challenge or enforce in another jurisdiction a judgment of such Chosen Court.

(d) Process in any such proceeding may be served on any Person with respect to such Applicable Matters anywhere in the world, whether within or without the jurisdiction of any such Chosen Court. Without limiting the foregoing, each such Person agrees that service of process on such party as provided in <u>Section 6.5</u> shall be deemed effective service of process on such Person.

(e) <u>Waiver of Jury Trial</u>. EACH PARTY HERETO, FOR ITSELF AND ITS AFFILIATES, HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, SUIT OR OTHER PROCEEDING (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THE ACTIONS OF THE PARTIES HERETO OR THEIR RESPECTIVE AFFILIATES PURSUANT TO THIS AGREEMENT OR IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE OR ENFORCEMENT HEREOF.

Section 6.3 Entire Agreement; No Third Party Beneficiary. This Agreement, the Investors' Rights Agreement and the Note Purchase Agreement contain the entire agreement by and among the parties with respect to the subject matter hereof and all prior negotiations, writings and understandings relating to the subject matter of this Agreement. Except as provided in <u>Article IV</u>, this Agreement is not intended to confer upon any Person not a party hereto (or their successors and permitted assigns) any rights or remedies hereunder.

Section 6.4 <u>Expenses</u>. Without limitation of <u>Section 3.3</u>, all fees, costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby, including accounting and legal fees, shall be paid by the Company.

Section 6.5 <u>Notices</u>. All notices, requests, demands and other communications under this Agreement shall be in writing and shall be deemed to have been duly given or made as follows: (a) if sent by registered or certified mail in the United States return receipt requested, upon receipt; (b) if sent by nationally recognized overnight air courier, one (1) business day after mailing; (c) if sent by e-mail transmission, with a copy sent on the same day in the manner provided in <u>Section 6.5(a)</u> or (b), when transmitted and receipt is confirmed; and (d) if otherwise actually personally delivered, when delivered, provided, that such notices, requests, demands and other communications are delivered to the address set forth below, or to such other address as any party shall provide by like notice to the other Parties to this Agreement:

If to the Company, to:

Eargo, Inc. 2665 North First Street, Suite 300 San Jose, CA 95134 Email: [\*\*\*] Attention: Chief Legal Officer

with a copy (which shall not constitute notice) to:

Latham & Watkins LLP 10250 Constellation Blvd. Suite 1100 Los Angeles, CA 90067 E-mail: [\*\*\*] Attention: David Zaheer; Charles Ruck; Kathleen Wells

If to the Investors, to:

c/o Patient Square Equity Partners, LP 2884 Sand Hill Road, Suite 100 Menlo Park, CA 94025 Email: [\*\*\*] Attention: Adam Fliss, Justin Sabet-Peyman

with a copy (which shall not constitute notice) to:

Ropes & Gray LLP Three Embarcadero Center San Francisco, CA 94111 Email: [\*\*\*] Attention: Jason Freedman

and

Ropes & Gray LLP Prudential Tower 800 Boylston Street Boston, MA 02199 Email: [\*\*\*] Attention: Tom Fraser

Section 6.6 <u>Successors and Assigns</u>. This Agreement will be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns. Except as provided in <u>Section 5.1</u>, no assignment of this Agreement or of any rights or obligations hereunder may be made by any party hereto without the prior written consent of the other parties hereto;. Any purported assignment or delegation in violation of this Agreement shall be null and void *ab initio*.

Section 6.7 <u>Headings</u>. The Section, Article and other headings contained in this Agreement are inserted for convenience of reference only and will not affect the meaning or interpretation of this Agreement.

Section 6.8 <u>Amendments and Waivers</u>. This Agreement may not be modified or amended except by an instrument or instruments in writing signed by the Company and the Investors. Any party hereto may, only by an instrument in writing, waive compliance by any other party or parties hereto with any term or provision hereof on the part of such other party or parties hereto to be performed or complied with. No failure or delay of any party in exercising any right or remedy hereunder shall operate as a waiver thereof, nor will any single or partial exercise of any right or power, or any abandonment or discontinuance of steps to enforce such right or power, preclude any other or further exercise thereof or the exercise of any other right or power. The waiver by any party hereto of a breach of any term or provision hereof shall not be construed as a waiver of any subsequent breach. The rights and remedies of the parties hereunder are cumulative and are not exclusive of any rights or remedies that they would otherwise have hereunder.

## Section 6.9 Interpretation; Absence of Presumption.

(a) For the purposes hereof: (i) words in the singular shall be held to include the plural and vice versa and words of one gender shall be held to include the other gender as the context requires; (ii) the terms "hereof," "herein," and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement as a whole and not to any particular provision of this Agreement, and Section and paragraph references are to the Sections and paragraphs in this Agreement unless otherwise specified; (iii) the word "including" and words of similar import when used in this Agreement shall mean "including, without limitation," unless the context otherwise requires or unless otherwise specified; and (iv) the word "or" shall not be exclusive.

(b) With regard to each and every term and condition of this Agreement, the parties hereto understand and agree that the same have or has been mutually negotiated, prepared and drafted, and if at any time the parties hereto desire or are required to interpret or construe any such term or condition, no consideration will be given to the issue of which party hereto actually prepared, drafted or requested any term or condition of this Agreement.

Section 6.10 <u>Severability</u>. Any provision hereof that is held to be invalid, illegal or unenforceable in any respect by a court of competent jurisdiction, shall be ineffective only to the extent of such invalidity, illegality or unenforceability, without affecting in any way the remaining provisions hereof, <u>provided</u>, <u>however</u>, that the parties will attempt in good faith to reform this Agreement in a manner consistent with the intent of any such ineffective provision for the purpose of carrying out such intent.

Section 6.11 <u>Non-Recourse Parties</u>. Without limiting any of the provisions contained in the equity commitment letter executed by Patient Square Equity Partners, LP on the date hereof, this Agreement may only be enforced against, and any claim or cause of action based upon, arising out of, or related to this Agreement, or the transactions contemplated hereby may only be brought against the entities that are expressly named as parties hereto and their respective successors and assigns. Except as set forth in the immediately preceding sentence, no past, present or future

director, officer, employee, incorporator, member, partners, stockholder, Affiliate, agent, attorney, advisor or representative of any party hereto (collectively, the "Non-Recourse Parties") shall have any liability for any obligations or liabilities of any party hereto under this Agreement or for any claim based on, in respect of, or by reason of, the transactions contemplated hereby (whether in contract, in tort, in law or in equity or provided by statute, whether by or through attempted piercing of the corporate, limited partnership or limited liability company veil or otherwise), except to the extent expressly set forth in this Agreement. For the avoidance of doubt, this Section 6.11 is intended to benefit and may be enforced by each Non-Recourse Party (and each such Person shall be a third-party beneficiary of this Section 6.11) and shall be binding on all respective successors and permitted assigns thereof.

(The next page is the signature page)

IN WITNESS WHEREOF, the parties have executed this Registration Rights Agreement as of the date first above written.

# EARGO, INC.

By: <u>/s/ Christian Gormsen</u> Name: Christian Gormsen Title: Chief Executive Officer and President

## PSC ECHO, LP

By: /s/ Adam Fliss

Name: Adam Fliss Title: General Counsel and Chief Compliance Officer

[Signature Page to Registration Rights Agreement]

#### EXHIBIT A DEFINED TERMS

1. The following capitalized terms have the meanings indicated:

"Affiliate" of any Person means any Person, directly or indirectly, controlling, controlled by or under common control with such Person.

"Commission" means the Securities and Exchange Commission.

"Common Stock" means the Company's common stock, par value \$0.0001 per share.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, or any similar successor federal statute, and the rules and regulations of the Commission thereunder, all as the same shall be in effect from time to time.

"Holder" means (a) any Investor holding Registrable Securities and (b) any Permitted Transferee to which the rights under this Agreement have been transferred in accordance with <u>Section 5.1</u>.

"Investor Rights Agreement" means the Investor Rights Agreement, dated the date hereof, by and among the Company and the Investors.

"Other Holder" has the meaning given to such term in the Investor Rights Agreement.

"Permitted Transferee" has the meaning given to such term in the Investor Rights Agreement.

"Person" means an individual, corporation, partnership, limited liability company, joint venture, association, trust, unincorporated organization, other legal entity, or any government or governmental agency or authority.

"register", "registered" and "registration" refer to a registration effected by preparing and filing a registration statement in compliance with the Securities Act, and the declaration or ordering of the effectiveness of such registration statement.

"Registration Expenses" means (a) all expenses incurred by the Company in complying with <u>Articles I</u> and <u>II</u>, including, without limitation, all registration, qualification, listing and filing fees, printing expenses, escrow fees, fees and disbursements of counsel for the Company, blue sky fees and expenses, and the expense of any special audits incident to or required by any such registration and (b) the fees and expenses of any counsel to the Holders, not to exceed twenty-five thousand dollars (\$25,000) with respect to any particular registration pursuant to <u>Article I</u> or <u>II</u>, and any such excess shall be a Selling Expense; <u>provided</u>, <u>however</u>, that Registration Expenses shall not include any Selling Expenses.

A-1

"**Registrable Securities**" means (a) any shares of Common Stock issued by the Company to the Holders, and (b) any Common Stock or other securities actually issued in respect of the securities described in clause (a) above or this clause (b) upon any stock split, stock dividend, recapitalization, reclassification, merger, consolidation or similar event; provided, however, that the securities described in clauses (a) and (b) above shall only be treated as Registrable Securities until the earliest of: (i) the date on which such security has been registered under the Securities Act and disposed of in accordance with an effective Registration Statement relating thereto; (ii) the date on which such security has been sold pursuant to Rule 144 and the security is no longer a Restricted Security; (iii) the date on which all Registrable Securities owned by the Holder thereof may be resold without volume or other restrictions during any and all three-month periods pursuant to Rule 144; or (iv) the date on which such security is transferred in a transaction pursuant to which the registration rights are not also assigned in accordance with <u>Section 5.1</u>.

"Restricted Securities" means any Common Stock required to bear the legend set forth in the Investors' Rights Agreement.

"Rights Offering Closing Date" means the closing date of a rights offering relating to the Common Stock as contemplated by the Note Purchase Agreement.

"Rule 144" means Rule 144 promulgated under the Securities Act and any successor provision.

"Rule 405" means Rule 405 promulgated under the Securities Act and any successor provision.

"SEC" means the Securities and Exchange Commission.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder or any similar federal statute and the rules and regulations of the Commission thereunder, all as the same shall be in effect at the time.

"Selling Expenses" means (a) all underwriting discounts, selling commissions and stock transfer taxes applicable to the securities registered by the Holders, and (b) the fees and expenses of any counsel to the Holders in excess of twenty-five thousand dollars (\$25,000) with respect to any particular registration pursuant to <u>Article I</u> or <u>II</u>.

"Shelf Registration" means the Resale Shelf Registration or a Subsequent Shelf Registration, as applicable.

"Transfer" has the meaning given to such term in the Investor Rights Agreement.

A-2

2. The following terms are defined in the Sections of the Agreement indicated:

## INDEX OF TERMS

Term Agreement Applicable Matters Chosen Court Company Company Indemnified Parties Company Indemnitee Effectiveness Period Holder Holder Indemnified Parties Indemnified Party Indemnifying Party Investor Market Stand-Off Purchase Agreement Resale Shelf Registration Resale Shelf Registration Statement Selling Stockholder Information Shelf Offering Subsequent Holder Notice Subsequent Shelf Registration Take-Down Notice Underwritten Offering

Section Preamble Section 6.2(b) Section 6.2(b) Preamble Section 4.1 Section 4.5 Section 1.2 Section 5.1 Section 4.2 Section 4.3 Section 4.3 Preamble Section 3.6 Preamble Section 1.1 Section 1.1 Section 4.1 Section 1.7 Section 1.5 Section 1.3 Section 1.7 Section 1.6

A-3

### EXHIBIT B RULE 144 REPORTING

The Company and the Investors agree that this Exhibit B is attached to, and forms a part of, the Agreement.

With a view to making available the benefits of Rule 144 to the Holders, the Company agrees that, for so long as a Holder owns Registrable Securities, the Company will use its reasonable best efforts to:

1. File with the Commission in a timely manner all reports and other documents required of the Company under the Exchange Act; and

2. So long as a Holder owns any Restricted Securities, furnish to the Holder forthwith upon written request a written statement by the Company as to its compliance with the reporting requirements of the Exchange Act.



## Eargo Strengthens Financial Position with up to \$125 Million Strategic Investment from Patient Square Capital

Company to Hold Conference Call and Webcast on June 27, 2022 at 8 a.m. Eastern Time

SAN JOSE, Calif., June 27, 2022 (GLOBE NEWSWIRE) – Eargo, Inc. (Nasdaq: EAR) (the "Company"), a medical device company on a mission to improve the quality of life of people with hearing loss, today announced it has signed a definitive agreement with <u>Patient Square Capital</u> ("Patient Square") to raise \$100 million through the sale of senior secured convertible notes, with an additional \$25 million future investment subject to certain conditions. The Company intends to use the net proceeds of Patient Square's investment for working capital purposes, to fund the Company's omnichannel growth strategy and to repay all of the Company's approximately \$15 million in existing third-party debt as well as related pay-off expenses.

Under the agreement, the Company will issue \$100 million aggregate principal amount of senior secured convertible notes to Patient Square upon closing (the "First Tranche Investment").

Within 180 days of the issuance of the First Tranche Investment, the Company expects to undertake a rights offering that will be registered under the Securities Act of 1933, as amended (the "Securities Act"), pursuant to which its stockholders as of a record date to be determined will have the right to purchase their pro rata share of 375 million shares of newly issued common stock at a purchase price of \$0.50 per share.

Patient Square has agreed to purchase up to an additional \$25 million aggregate principal amount of senior secured convertible notes (the "Second Tranche Investment") if the rights offering has not been completed within 150 days after closing of the First Tranche Investment, or if the rights offering is completed within 150 days after the closing of the First Tranche Investment, and existing stockholders of the Company subscribe for less than 75 million shares of newly-issued common stock of the Company in such rights offering.

If the rights offering is completed before 150 days following closing of the First Tranche Investment, and existing stockholders subscribe for at least 75 million of shares, total cash proceeds to Eargo (from both the issuance of the senior secured convertible notes and the rights offering) increases to \$137.5 million.

Any proceeds from the rights offering will be used to redeem the senior secured convertible notes and for general corporate purposes. In the event the rights offering is not fully subscribed, the remaining senior secured convertible notes not redeemed in the rights offering will convert into 375 million shares of common stock less the number of shares subscribed for in the rights offering.

The senior secured convertible notes accrue interest at a rate of 12.00% per annum payable in kind, and mature 1 year after the closing of the First Tranche Investment. The senior secured convertible notes are redeemable at the option of the Company at any time after issuance. Upon the maturity date, redemption or conversion, the senior secured convertible notes must be repaid at an annualized return on their initial face value of no less than 12.00% or, if greater, at 150% of the initial principal amount of such senior secured convertible notes.

In advance of the rights offering, the Company, at its discretion, is able but not obligated to seek alternative forms of capital to pay off the convertible notes or to sell the Company. The Company expects that in the event it enters into a sale transaction, then it would not proceed with the rights offering.

The Company expects to hold a meeting of its stockholders within 120 days of the closing of the First Tranche Investment, seeking stockholder approval of the issuance of the common stock upon conversion of the notes and an increase in the number of shares authorized under the Company's charter to an amount sufficient to consummate the rights offering.

Upon closing of the transaction, Patient Square will have a board observer position on the Company's Board of Directors. Following the completion of the rights offering, Patient Square will have the right to nominate a number of directors to the Board of Directors that is proportionate to its ownership of the Company.

Christian Gormsen, President and Chief Executive Officer, said, "We believe this commitment of up to \$125 million in additional capital from an experienced healthcare investor will enable us to pursue our omni-channel growth strategy powered by telecare. This includes further optimizing our cash-pay business, focusing on potentially regaining insurance coverage of Eargo for government employees under the FEHB program, and refining and expanding our physical retail strategy."

Jim Momtazee, Managing Partner of Patient Square, said, "With over 40 million people in the U.S. with hearing loss but only approximately a quarter of them owning a hearing aid, the hearing care market is significantly underpenetrated. Eargo's innovative virtually invisible product and its unique telecare support infrastructure position the Company well for growth across multiple channels. We are excited to support Eargo in its mission to improve quality of life for people with hearing loss."

Perella Weinberg Partners LP and BofA Securities, Inc. acted as financial advisors to Eargo.

The offer and sale of the senior secured convertible notes and any shares of common stock issued upon conversion are being made in a transaction not involving a public offering and have not been registered under the Securities Act or applicable state securities laws and will be sold in a private placement pursuant to Section 4(a)(2) and/or Regulation D of the Securities Act. The securities may not be offered or sold in the United States absent registration or pursuant to an exemption from the registration requirements of the Securities Act and applicable state securities laws. The Company has agreed to file a registration statement covering the resale of the common stock issuable upon conversion of the senior secured convertible notes.

This press release does not constitute an offer to sell or the solicitation of an offer to buy the securities, nor shall there be any sale of the securities in any state or jurisdiction in which such offer or sale would be unlawful prior to the registration or qualification under the securities laws of such state or jurisdiction.

### **Conference Call and Webcast Information**

Eargo will host a conference call to discuss the transaction before market open on Monday, June 27, 2022, at 5 a.m. Pacific Time / 8 a.m. Eastern Time. The conference call can be accessed live over the phone at (833) 649-1234 for U.S. callers or (914) 987-7293 for international callers, using conference ID: 9639369. The live webcast can be accessed at ir.eargo.com.

## <u>About Eargo</u>

Eargo is a medical device company dedicated to improving the quality of life of people with hearing loss. Our innovative product and go-to-market approach address the major challenges of traditional hearing aid adoption, including social stigma, accessibility and cost. We believe our Eargo hearing aids are the first virtually invisible, rechargeable, completely-in-canal, FDA regulated, exempt Class I or Class II devices indicated to compensate for mild to moderate hearing loss. Our differentiated, consumer-first solution empowers consumers to take control of their hearing. Consumers can purchase online or over the phone and get personalized and convenient consultation and support from hearing professionals via phone, text, email or video chat. The Eargo device is offered to consumers at approximately half the cost of competing hearing aids purchased through traditional channels in the United States.

Eargo's sixth generation device, Eargo 6, is an FDA Class II exempt hearing device featuring Sound Adjust technology that automatically optimizes the soundscape as the user moves between environments. Eargo 6 is available for purchase here.

Related Links <u>http://eargo.com</u>

### About Patient Square Capital

Patient Square Capital (www.patientsquarecapital.com) is a dedicated health care investment firm that partners with best-in-class management teams whose products, services and technologies improve health. Patient Square utilizes deep industry expertise, a broad network of relationships and a true partnership approach to make investments in companies that will grow and thrive. Patient Square invests in businesses that strive to improve patient lives, strengthen communities and create a healthier world. Patient Square's team of industry-leading executives is differentiated by the depth of focus in health care, the breadth of health care investing experience, and the network it can activate to drive differentiated outcomes.

### Forward-Looking Statements

This press release contains forward-looking statements within the meaning of Section 27A of the Securities Act and Section 21E of the Securities Exchange Act of 1934, as amended. All statements other than statements of historical fact contained in this press release are forward-looking statements, including statements regarding the notes transaction; the rights offering; the Company's plans to optimize its cash-pay business, focus on potentially regaining insurance coverage of Eargo for government employees under the Federal Employees Health Benefits (FEHB) program, and refine and expand its physical retail strategy; and Eargo's market position and strategy. Forward-looking statements are not guarantees of future performance and are subject to risks, uncertainties and assumptions that could cause actual results and events to differ materially from those anticipated, including, but not limited to, risks, uncertainties and assumptions related to: the closing of the notes transaction; completion of the rights offering; the extent to which we may be able to validate processes to support the submission of claims for reimbursement from the FEHB program in the future, if at all, and our ability to maintain or increase insurance coverage of our hearing aids; the timing or results of ongoing claims audits and medical records reviews by third-party payors; the extent of losses from hearing aids delivered to customers from September 21, 2021 until December 8, 2021; the impact of third-party payor audits and the regulatory landscape for hearing aid devices on our business and results of operations; our expectations concerning additional orders by existing customers; our expectations regarding the potential market size and size of the potential consumer populations for our products and any future products, including insurance coverage of our hearing aids; our ability to release new hearing aids and the anticipated features of any such hearing aids; developments and projections relating to our competitors and our industry, including competing products; our ability to maintain our competitive technological advantages against new entrants in our industry; the pricing of our hearing aids; our expectations regarding the ability to make certain claims related to the performance of our hearing aids relative to competitive products; our expectations with regard to changes in the regulatory landscape for hearing aid devices, including the anticipated implementation of a pending over-the-counter hearing aid regulatory framework; and our estimates regarding the COVID-19 pandemic, including but not limited to, its duration and its impact on our business and results of operations. These and other risks are described in greater detail under the section titled "Risk Factors" contained in our Annual Report on Form 10-K and Quarterly Reports on Form 10-Q and our other filings with the Securities and Exchange Commission. Any forward-looking statements in this press release are made pursuant to the Private Securities Litigation Reform Act of 1995, as amended, are based on current expectations, forecasts and assumptions, and speak only as of the date of this press release. Except as required by law, we undertake no obligation to publicly update any forward-looking statements, whether as a result of new information, future events or otherwise.

Investor Contact Nick Laudico Senior Vice President of Corporate Strategy and Investor Relations ir@eargo.com